



Special Issue. Innovation in Family Business. Research papers: Guest editor: Alfredo De Massis, Josip Kotlar, Daniel Lorenzo, Pedro Núñez-Cacho Utrilla, Ignacio Osborne, Antonio Gallardo

Innovation in family firms: a holistic bibliometric overview of the research field Gloria Aparicio, Txomin Iturralde, Valeriano Sanchez-Famoso	71-84
An Analysis of Open Innovation Determinants: the Case Study of Singapore based Family owned Enterprises Annie Koh, Esther Kong, Giuseppe Timperio	85-101
Collaborative Innovation in the family SME: conceptualization, goals, and success factors Unai Arzubiaga, Amaia Maseda, Amaia Uribarri, Jesús Manuel Palma Ruiz	102-114
Does too much love hinder innovation? Family involvement and firms' Innovativeness in family-owned Small Medium Enterprises (SMEs) Filippo Ferrari	115-127
Entrepreneurial orientation and product innovation. The moderating role of family involvement in management Taras Fredyna, Daniel Ruiz Palomo, Julio Dieguez Soto	128-145

### **Research paper:**

**Private Equity focused on Family Firms & Small and Medium Sized Companies:** 146-158 **Review and Science Mapping Analysis of the Recent Scientific Field** Laura Arteche Bueno, Camilo Prado Román, Antonio Fernández Portillo





UNIVERSIDAD DE MÁLAGA





INSTITUTO DE LA EMPRESA FAMILIAR





uma 💦

editorial

Editors-in-Chief

EUROPEAN JOURNAL OF FAMILY BUSINESS

RESA FAMILIAR DE LA UNIVERSIDAD DE MÁLAGA

M<sup>a</sup> Concepción López Fernández. Universidad de Cantabria Amaia Maseda García. Universidad del País Vasco

### Associate Editors

María Katiuska Cabrera Suárez. Universidad de Las Palmas de Gran Canaria José Carlos Casillas Bueno. Universidad de Sevilla Julio Diéguez Soto. Universidad de Málaga Txomin Iturralde Jainaga. Universidad del País Vasco Ángel Luis Meroño Cerdán. Universidad de Murcia Antonio Navarro García. Universidad de Sevilla Gregorio Sánchez Marín. Universidad de Murcia

### **Managing Editors**

José Roberto Vila Oblitas. Universidad de Málaga Vanesa F. Guzmán Parra. Universidad de Málaga

#### **Editorial Advisory Board**

Remedios Hernández Linares (Universidad de Extremadura) Felipe Hernández Perlines (Universidad de Castilla-La Mancha María Jesús Hernández Ortiz (Universidad de Jaén) Claudio Müller Valenzuela (Universidad de Chile) José Antonio Novo Peteiro (Universidad de A Coruña) Fernando Olivares (Universidad de Alicante) José Antonio Otero Parra (Universidad Europea Miguel de Cervantes) Antonio Padilla Meléndez (Universidad de Málaga) María José Pérez Rodríguez (Universidad Complutense de Madrid) María José Quero Gervilla (Universidad de Málaga) Juan Hernangómez Barahona (Universidad de Valladolid) Carmina Jambrino Maldonado (Universidad de Málaga) Santiago Lago Peñas (Universidad de Vigo) Virginia Llorente Muñoz (Universidad de Máľaga) Guadalupe Manzano García (Universidad de La Rioja) María del Pilar Marques Gou (Universidad de Girona) Rocío Martínez Jiménez (Universidad de Jaén) María José Martínez Romero (Universidad de Almería) M. Teresa Mata Sierra (Universidad de León) João José de Matos Ferreira (Universidad da Beira Interior)

Madrid) Sebastián Molinillo Jiménez (Universidad de Málaga) María Jesús Morales Caparrós (Universidad de Málaga) María Jesús Moreno Domínguez (Universidad de Huelva) Ana María Moreno Menéndez (Universidad de Sevilla) Manuel Rey Moreno (Universidad de Sevilla) Elena Rivo López (Universidad de Vigo) José Javier Rodríguez Alcaide (Universidad de Córdoba) Lázaro Rodríguez Ariza (Universidad de Granada Alfonso A. Rojo Ramirez (Universidad de Almería) Francisco Javier Rondán Cataluña (Universidad de Sevilla) Juan Carlos Rubio Romero (Universidad de Málaga) José L. Ruiz-Alba Robledo (University of Westminster, London, United Kingdom, José Antonio C. Santos (Universidade do Algarve) Salvador Rus Rufino (Universidad de León) Pilar Saldaña Gonzalo (Universidad de Barcelona) Cristina Quintana García (Universidad de Málaga) Adolfo Vázquez Sánchez (Universidad de Sevilla) Rafael Ventura Fernández (Universidad de Málaga) Georgina Whyatt (Oxford Brookes University)

Pere Mercadé Melé (Univesidad de Málaga)

Víctor Molero Ayala (Universidad Complutense de

Isabel María Abad Guerrero (Universidad de Málaga) Francisco Acedo González (Universidad de Sevilla)

Pilar Alarcón Urbistondo (Universidad de Málaga)

Benjamín del Alcazar Martínez (Universidad de Málaga) Ana Rosa del Águila Obra (Universidad de Málaga)

Cristina Aragón Amonarriz (Universidad de Deusto) Rubén Arrondo García (Universidad de Oviedo)

Juan Carlos Ayala Calvo (Universidad de La Rioja) Tomás Bañegil Palacios (Universidad de

Extremadura)

Mário Lino Barata Raposo (Universidad da Beira Interior)

Rodrigo Basco (American University of Sharajah, Sharjah, Emiratos Árabes)

Helena Maria Batista Alves (Universidad da Beira Interior)

Patricia Benavides Velasco (Universidad de Málaga)

Guillermo Bermúdez González (Universidad de Málaga)

Carlos De las Heras Pedrosa (Universidad de Málaga)

José Díez de Castro (Universidad de Santiago de Compostela)

Yüksel Ekinci (Oxford Brookes University) Alejandro Escribá Esteve (Universidad de Valencia)

Miguel Ángel Gallo (IESE, Barcelona) Ignacio Gallego Domínguez (Universidad de Córdoba)

Germán Gémar Castillo (Universidad de Málaga) Jesús Giménez Sánchez (Universidad Complutense de Madrid)

Eva María González Robles (Universidad de Málaga)

European Journal of Family Business is an open access journal published in Malaga by UMA Editorial. ISSN 2444-8777 ISSN-e 2444-877X Contents published on this journal are licensed under a Creative Commons Atribution-NonCommercial-ShareAlike 4.0 International License (CC BY-NC-SA 4.0).

### CONTENTS

Field

Special issue. Innovation in Family Business. Research papers: Guest Editors: Alfredo De Massis, Josip Kotlar, Daniel Lorenzo, Pedro Núñez-Cacho Utrilla, Ignacio Osborne, Antonio Gallardo

Innovation in family firms: a holistic bibliometric overview of the research field Gloria Aparicio, Txomin Iturralde, Valeriano Sanchez-Famoso	71-84
An Analysis of Open Innovation Determinants: the Case Study of Singapore based Family owned Enterprises Annie Koh, Esther Kong, Giuseppe Timperio	85-101
Collaborative innovation in the family SME: conceptualization, goals, and success factors Unai Arzubiaga,Amaia Maseda, Amaia Uribarri, Jesús Manuel Palma-Ruiz	102-114
Does too much love hinder innovation? Family involvement and firms' innovativeness in family-owned Small Medium Enterprises (SMEs) Filippo Ferrari	115-127
Entrepreneurial orientation and product innovation. The moderating role of family involvement in management. Taras Fredyna, Daniel Ruiz-Palomo, Julio Dieguez-Soto	128-145
Research paper:	
Private Equity focused on Family Firms & Small and Medium Sized Companies: Review and Science Mapping Analysis of the Recent Scientific	146-158

Laura Arteche-Bueno, Camilo Prado-Román, Antonio Fernández-Portillo



EUROPEAN JOURNAL OF FAMILY BUSINESS

http://www.revistas.uma.es/index.php/ejfb



### Editorial letter Innovation in family firms: Bridging the theory-practice gap

Daniel Lorenzo <sup>a\*</sup>, Pedro Núñez-Cacho <sup>b</sup>, Alfredo De Massis <sup>c</sup>, Josip Kotlar <sup>d</sup>

<sup>a</sup> University of Cádiz (Spain)

<sup>b</sup>University of Jaén (Spain)

<sup>C</sup>Free University of Bolzano, Italy and Lancaster University, UK

<sup>C</sup> Politecnico di Milano School of Management, Italy

Innovation in family firms is still a controversial issue within the academic community and poses some unique challenges for family business owners and managers. This special issue on innovation in family firms results from the cooperation of both academic and business guest editors, in a pioneering initiative that is not usual in academic journals. Indeed, a key feature of this Special Issue has been the collaboration with two family business leaders, who have been involved in the editorial process together with the academics.

The two business editors that we involved are Antonio Gallardo, Vicepresident of <u>Almirall</u> and former director of <u>FBN-Family Business Network</u>, and Ignacio Osborne, CEO of the <u>Osborne Group</u> and Chairman of the <u>Spanish Family Firm</u> <u>Institute.</u>

In order to introduce the six papers that make up this special issue on innovation in family firms, we as academic editors are pleased to include some comments from the business editors that emerged during our interactions with the aim to make a step forward toward bridging the gap between research and practice on family business acknowledging the different innovation, adopted perspectives and approaches by academics and practitioners. As the business editor Mr. Osborne points: "Innovation issues in family firms are nowadays more important than ever, due to the rapid developments that are occurring in the business world and its corresponding technologies".

Despite being a topic analyzed by a number of authors over time (Feranita, Kotlar and De Massis, 2017; Aparicio, Iturralde and Sánchez-Famoso. 2020 in this issue; Chrisman, Chua, De Massis, Frattini and Wright, 2015), the study of innovation in family business still requires a greater volume of research to provide answers to the needs of family businesses. The distinctive nature of family firms results in a complex influence on the innovation process (De Massis, Frattini and Lichtenthaler, 2013), which is reflected in mixed research findings. For instance, the conclusions of the published research offer sometimes contradictory results, since family businesses can be considered innovative (Aronoff, 1998; Craig and Moores, 2006) or conservative (Sharma, Chrisman, y Chua, 1997; Zahra, Hayton y Salvato, 2004; Gómez-Mejía et al., 2007), with several studies that can support whatever of the two options. Familv businesses present a number of characteristics that, a priori, seem to favor innovation, such as long-term orientation (Tagiuri and Davis, 1996; Ward and Aronoff, 1994), the desire for continuity through the following generations (Miroshnychenko et al., 2020; Gallo, 1995), patient capital (De Massis, Audretsch, Uhlaner and Kammerlander, 2018; Sirmon and Hitt, 2003), and the long tenure of their main leaders (Lorenzo, 2020). The replacement of the prior generation by the next generation implies the access of younger people to the leadership of the company, who also often present a greater level of qualification (De Massis et al., 2008;

### https://doi.org/10.24310/ejfbejfb.v9i2.9241

Copyright 2019: Daniel Lorenzo, Pedro Núñez-Cacho, Alfredo De Massis, Josip Kotlar. European Journal of Family Business is an open access journal published in Malaga by UMA Editorial. ISSN 2444-8788 ISSN-e 2444-877X This work is licensed under a Creative Commons Atribution-NonCommercial-ShareAlike 4.0 International License (CC BY-NC-SA 4.0).

Cabrera-Suárez. 2011). Young and gualified leaders would provide a new momentum to the firm, by means of the renewal of the firm (Núñez-Cacho and Lorenzo, 2020). Likewise, the successors receive an important legacy by means of the values of the family business (Erdogan et al., 2020), such as effort, perseverance, austerity, excellence, long-term orientation and entrepreneurial spirit, as basic foundations of their way of understanding business activity (Bermejo, 2008). Accordingly, the new generation managers could be in the best conditions to reinvent the company, since they know the business from within and they also provide the new vision of a person with a working life ahead. Another factor that favors the renewal impulse of the next generation is the familial support to carry out a prolonged tenure over time, which will not be as conditioned by short-term results as in other types of companies, by the so-called patient capital (Sirmon and Hitt, 2003) of the family business (Lorenzo, 2020). But, even if these ideal conditions are met in a specific family firm, it is not guaranteed that the company realizes the innovation it needs. Therefore, it is more needed to shed light about the determinants and conditions for innovation.

The editors of this special issue selected a number of papers to reflect the state-of-the-art on this topic, indicating some of the most promising research lines on innovation. According to the business editor Mr. Gallardo, "A very important aspect emerging from this special issue is that the papers published in it reveal that external contributions to the internal know-how of the family and the business are often vital to help produce the changes needed by a family firm for innovation to take place".

Innovation in the family business has been a phenomenon of great interest to researchers, especially in the last decade. This is highlighted in the article that opens this special issue by presenting a complete bibliometric review of the literature on innovation in family businesses. Generally, researchers have noted that the influence of the family is the factor that makes this type of businesses different from the other ones (Habbershon and Williams, 1999; Lorenzo and Núñez-Cacho, 2012). However, in order to conclude that this is really true, it is necessary to identify the nature of these differences and determine how and why they affect the innovative behavior of the family business.

The paper Innovation on family businesses: A holistic bibliometric (Aparicio, et al. 2020) offers an overview of the research field through an analysis of 207 articles that were published between 1994 and 2017. The authors complement other recent reviews such as those by Feranita et al. (2017) and Calabrò, Vecchiarini, Gast, Campopiano, De Massis and Kraus (2019), and reflect about the take-off of research on

D. Lorenzo, P. Núñez-Cacho, A. De Massis, J. Kotlar

innovation that takes place since 2009. In the study two differentiated periods are highlighted: An initial one that covers the years 1994 to 2009, and one of expansion from 2010 to 2017. In addition, they identify the most influential journals, the most referenced articles, the most productive scholars -namely, De Massis, Frattini, Craig, Chrisman, Fang, Kotlar and Nordqvist appear as the most productive and referenced ones- and the main lines of research developed, providing a clear and synthetic map of innovation research in family businesses today. This paper approaches innovation from a more theoretical perspective, and also presents the lines of research that are currently being developed. These lines include the internal factors of the family business and its influence on innovation, as well as external factors, among others advances in research in the subject.

The paper An Analysis of Open Innovation Determinants: The Case Study of Singapore based Family owned Enterprises, by Koh, Kong and Timperio (2020, this issue) analyzes the drivers of open innovation by studying cases of family businesses in Singapore. The authors highlight the external determinants and catalysts of innovation projects, such as family and business culture, access to external funds, government support for initiatives, market dynamics and partnership between companies. In addition to these six external determinants, there are two other factors that have a great influence on open innovation. First, family capital, which is the main source of financing for innovative initiatives. Second, a strong external network, supported by Singapore's legal and regulatory framework that fosters innovation, promotes the development of an enabling business environment so that the spirit of innovation can truly thrive. Most of the surveyed companies' managers mentioned process innovation as the most critical aspect, and also organizational innovation. Process innovation is considered superior by the companies included in the sample due to their capabilities to drive product innovation, marketing and organizational structure (and people). Organizational innovation is also considered of utmost importance, due to the need to adopt technologies such as digitalization, robotics or automation, which require an adequate organizational structure. Some ideas from the surveyed managers highlight these statements, like: "The correct processes create the necessary conditions to shape the products, as well as the marketing and organization structures," as well as "Having cutting-edge processes underway is a key differentiator." This study also reflects the need to establish new financing mechanisms adapted to the peculiarities of innovation processes. External capital injection and stimulus policies are necessary, although not sufficient, since they must be combined with the determinants of the internal functioning of family businesses.

The relevance of the external network is also highlighted in the paper Collaborative innovation in the family SME: conceptualization, goals, and success factors, by Arzubiaga et al.(2020, this issue), which analyzes the strategy of collaborative innovation that seeks the creation of knowledge, new product designs and Improving the efficiency of the production process.

Among the conditions of collaborative innovation. four groups stand out: The composition of the management team (in terms of family members percentage and number of generations involved in abilities (cognitive management), factors, absorption capacity and trajectory in innovation), attitudes, and legacy preservation, (referring to socio-emotional wealth and internal behavior). These factors of small and medium family businesses play a crucial role in the successful design and implementation of collaborative innovation. The main contributions of this paper can be summarized in the need for establish solid bases to deepen in the future the study of collaborative innovation. Moreover, a second contribution refers to the identification of the distinguishing characteristics of family SMEs. Arzubiaga, Maseda, Uribarri and Palma Ruiz (2020, this issue) also propose the analysis of the possible moderating effects of firm size and the sector to refine the impact of the variables in this model. looking to achieve excellence in collaborative innovation. As business reviewer, Mr. Osborne have highlighted collaborative innovation as one of the relevant issues in order to reinforce the role of innovation in their companies.

Absorptive capacity is another aspect of great interest to researchers. There are numerous factors that condition it, some of them are features of the family character that make the behavior of family businesses paradoxical (Kotlar et al. 2020). The paper titled A mediating model of innovative capacity between absorptive capacity and family business performance by Hernández-Perlines, Ariza-Montes and Araya-Castillo (2020, this issue) addresses the issue about absorptive capacity. Absorptive capacity is related to the identification, assimilation and exploitation of new knowledge by the company. businesses that have Those family these capabilities improve their performance. In addition, this effect is enhanced by the innovative capacity of the company, which acts as a mediator between absorption capacity and the performance, reinforcing company's this relationship. Thus, family business managers should focus their efforts on providing their organizations with the necessary skills to absorb and exploit knowledge. This will be easier if the company has developed innovative capabilities. In this sense, the business editor Mr. Gallardo points that: "There is also the possibility of establishing an advisory council with external collaborators that serves as a contrast to the company's board, in which oftentimes the weight of the family is too decisive."

The last two papers in this special issue address the role of family involvement in relation to innovation. Does too much love hinder innovation? Family involvement and firms' innovativeness in family-owned Small Medium Enterprises (SMEs), by Filippo Ferrari (2020, this issue) reflects on the role of family cohesion and its flexibility in the process of innovation, drawing upon the Olson Circumplex model (Olson, 2000) which is applied in a sample of Italian family businesses. The study indicates that unbalanced families show the lowest levels of innovation, although family cohesion and flexibility do not show a significant correlation with the overall level of organizational innovation. Flexibility shows a positive correlation with the process and behavioral innovation, which can be explained by the demand for new forms and organizational routines to deal with process innovation. Here the author suggests some human resources practices that promote flexibility, such as labor rotation (Ortega, 2001), or the development of a horizontal internal career (Ichniowsky et al. 1996, 1997, 1999). Families that lack cohesion show a negative correlation with strategic innovation and process innovation. Ferrari (2020, this issue) considers as disconnected family systems those in which family members are not cohesive and have little family loyalty. On the other hand, innovation in processes is encouraged with new ideas through contributions in terms of new ways of doing things. According to the authors, the Olson Circumplex model (Olson, 2000) offers a framework that can diagnose the extent to which family systems are balanced and how the effects of balanced or unbalanced family dynamics can affect the family business (Daspit et al. 2018). Business reviewers were especially interested on the conclusions of this paper, and also pointed that it would be necessary more research on that kind of negative influences stemmed from lack of cohesion within the business family.

Entrepreneurial orientation and product innovation: The moderating role of family involvement in management, by Fredyma, Ruiz Palomo and Diéguez (2020, this issue) addresses a classic concept closely linked to the study of innovation such as entrepreneurial orientation. The relationships between this variable and product innovation, incremental innovation and radical innovation are examined. The influence of family performance on the company is also analyzed. In their conclusions, Fredyma et al.

(2020.this issue) point out that familv involvement weakens the positive effect of entrepreneurial orientation in product innovation, especially in case of radical innovation. Therefore, the family business must be aware of these weaknesses to correct them, professionalizing with non-family managers and including their participation in innovation decisions. This conclusion is stressed by both business editors, Mr. Osborne and Mr. Gallardo, who point out that: "Having a network of external collaborators, some of them generalists and others specialized in specific problems, is nowadays practically indispensable."

Finally, the academic editors sincerely appreciate the contributions of two prominent Spanish businessmen, who have contributed to enrich this special issue with a business perspective, which helps to overcome the division that is sometimes perceived between the academic world and the business one. Both Antonio Gallardo and Ignacio Osborne represent the entrepreneurial vision that they have been able to maintain in their families and in their companies for generations. We all know how challenging it is for a family business to be entrepreneurial across generations (e.g., De Massis, Eddleston and Rovelli, 2020). Last but now least, we want to express our gratitude to the editor of the European Journal of Family Business, Professor Vanesa Guzmán for her collaboration and contributions.

The Osborne Group, founded in 1772, is one of the oldest family businesses in Europe. The group evolved from the original business of raising and exporting wines from Jerez to a wider food and beverage group which includes guality wines from various Spanish designations of origin, premium spirits, and products derived from Iberian pork, with a growing international acceptance, entering markets as demanding as China. Ignacio Osborne, a member of the sixth family generation, is the current president of the company since 2017, after 21 years as CEO. The company has been especially innovative in marketing, creating the symbol of the bull in the 50s, which has become a symbol that identifies the Spanish, transcending its initial origin as a reference for the winery.

Almirall is a pharmaceutical company founded in 1943. It is currently run by the second generation, which are giving way to the third. Although innovation is an essential requirement to compete in pharmaceutics, Almirall has managed to develop some well-known products in Spain, as Almax and Cleboril, becoming one of most innovative companies in the industry. Antonio Gallardo is honorary vice president of his company, which he chaired for 26 years. In addition, he was also president of the Family Council and the Family Office, as well as a member of the Executive Committee of the Family Business Network and vice president of the Family Business Institute.

### References

- Aronoff, C.E. (1998). Megatrends in family business. Family Business Review, 11 (3), 181-186.
- Bermejo, M. (2008). Hacia la empresa familiar líder. Una visión práctica y optimista. Prentice Hall.
- Cabrera Suárez, M. K. (2011). El proceso de sucesión en la empresa familiar. En Corona, J. (2011): *Empresa Familiar. Aspectos jurídicos y económicos*. Editorial Deusto / Instituto de la Empresa Familiar.
- Calabro A., Vecchiarini M., Gast J., Campopiano G., De Massis A., Kraus s. (2019). Innovation in family firms: A systematic literature review and guidance for future research. *International Journal of Management Reviews*, 21(3), 317-355.
- Chrisman J.J., Chua J.H., De Massis A., Frattini F., Wright M. (2015). The ability and willingness paradox in family firm innovation. *Journal of Product Innovation Management*, 32(3), 310-318.
- Craig, J.B.L. and Moores, K. (2006). A 10-Year Longitudinal Investigation of Strategy, Systems, and Environment on Innovation in Family Firms. Family Business Review, Vol. 19 (1), pp. 1-10.
- Daspit, J.J., Madison K., Barnett T., & Long R.G. (2018). The emergence of bifurcation bias from unbalanced families: Examining HR practices in the family firm using Circumplex Theory. Human Resource Management Review, 28 (1), 18-32. http://dx.doi.org/10.1016/j.hrmr.2017.05.003
- De Massis, A., Audretsch, D., Uhlaner, L., Kammerlander, N. (2018). Innovation with limited resources: Management lessons from the German Mittelstand. Journal of Product Innovation Management, 35(1), 125-146.
- De Massis A., Chua J.H., Chrisman J.J. (2008). Factors Preventing Intra-Family Succession. Family Business Review, 21(2), 183-199.
- De Massis A., Eddleston K., Rovelli P., (2020). Entrepreneurial by design: How organizational design affects family and nonfamily firms' opportunity exploitation. Journal of Management Studies. In press.
- https://doi.org/10.1111/joms.12568 De Massis A., Frattini F., Lichtenthaler U. (2013). Research on Technological Innovation in Family Firms: Present Debates and Future Directions. Family Business Review, 26(1), 10-31.
- Erdogan I., Rondi E., De Massis A. (2020). Managing the tradition and innovation paradox in family firms: A family imprinting perspective. Entrepreneurship Theory & Practice, 44(1), 20-54
- Feranita, F., Kotlar, J., De Massis, A. (2017). Collaborative innovation in family firms: Past research, current debates and agenda for future research. *Journal of Family Business Strategy*, 8(3), 137-156.
- Gallo, M.Á. (1995): Empresa familiar. Texto y casos. Editorial Praxis.
- Gómez-Mejía, L., Takács Hyanes, K., Núñez-Nickel, M., Jacobson, K. J. L. and Moyano-Fuentes, J. (2007): Socioemotional Wealth and Business Risks in Family-controlled Firms: Evidence from Spanish Olive Oil Mills. Administrative Science Quarterly, 52(1), 106-137.
- Habbershon T. G. & Williams M. L. (1999). A resourcebased framework for assessing the strategic

Lorenzo, D.. Núñez-Cacho, P., De Massis, A., Kotlar, J. (2019). Innovation in family firms: bridging the theory-practice gap. *European Journal of Family Business*, 9(2), 66-70.

advantages of family firms. *Family Business Review*, 12, 1, 1-25.

- Ichniowski, C., & Prennushi, G. (1997). The effect of human resource management practices on productivity. *American Economical Revue*, 87(3) 291-313.
- Ichniowski, C., & Shaw. K. (1999). The effects of human resource management systems on economic performance: An international comparison of U.S. and Japanese plants. *Management Science*, 45(5) 704-721.
- Ichniowski, C., Kochan T.A., Levine D., Olson C., & Strauss, G. (1996). What works at work: Overview and assessment. Industrial Relations: A Journal of Economy and Society, 35(3), 299-333.
- Kotlar, J., De Massis, A., Frattini, F., Kammerlander, N. (2020). Motivation gaps and implementation traps: the paradoxical and time-varying effects of family ownership on firm absorptive capacity. Journal of Product Innovation Management, 37(1), 2-25
- Lorenzo, D. (2020). Barriers to change in family businesses. *European Journal of Family Business*, 10 (1) (forthcoming).
- Lorenzo, D. and Núñez-Cacho, P. (2012). Inercia e innovación en la empresa familiar: una primera aproximación. *Revista de Empresa Familiar*, 2 (2), 23-40.
- Miroshnychenko I., De Massis A., Miller D., Barontini R. (2020). Family business growth around the world. Entrepreneurship Theory & Practice. In press. https://doi.org/10.1177/1042258720913028.
- Núñez-Cacho, P. and Lorenzo, D. (2020). Temporary factors that condition innovation: Comparison between family and non-family businesses. Entrepreneurship and sustainability issues, 7, (3) (forthcoming).
- **Olson, D.H. (2000).** Circumplex Model of Marital and Family Systems. Journal of Family Therapy, 22, 144-167.
- Ortega, J. (2001). Job Rotation as a Learning Mechanism. *Management Science*, 47, 10, pp. 1361-1370.
- Sharma, P. Chrisman, J.J. and Chua, J.H. (1997). Strategic management of family business: Past research and future challenges. *Family Business Review*, 10 (1), 1-35
- SIrmon, D. and Hitt, M. A. (2003). Managing resources: Linking unique resources, management, and wealth creation in family firms. *Entrepreneurship Theory and Practice*, 27(4), 339-358.
- Tagiuri, R. and Davis, J. (1996). Bivalent Attributes of Family Business, *Family Business Review*, 9(2), 199-208.
- Ward, J. L., & Aronoff, C. E. (1994). How family affects strategy. *Small Business Forum*, 1 (2), 85-90.
- Zahra, S. A., Hayton, J. C. and Salvato, C. A. (2004). Entrepreneurship in Family vs. Non-Family Firms: A Resource-Based Analysis of the Effect of Organizational Culture, *Entrepreneurship Theory and Practice*, *28*(4), 363-381.

Lorenzo, D.: Núñez-Cacho, P., De Massis, A., Kotlar, J. (2019). Innovation in family firms: bridging the theory-practice gap. *European Journal of Family Business*, 9(2), 66-70.



EUROPEAN JOURNAL OF FAMILY BUSINESS

http://www.revistas.uma.es/index.php/ejfb

# Innovation in family firms: a holistic bibliometric overview of the research field

Gloria Aparicio<sup>a</sup>, Txomin Iturralde<sup>a\*</sup>, Valeriano Sanchez-Famoso<sup>a</sup>

<sup>a</sup>University of the Basque Country, UPV/EHU (Spain)

Received 4 January 2019; accepted 19 December 2019

JEL CLASSIFICATION M1,M2

**KEYWORDS** Family firms; Innovation; Bibliometric review Abstract This paper presents a bibliometric analysis of innovation in family firms, focusing on aspects that prior literature review studies did not fully understand or evaluate. It is based on the bibliometric evaluation of 207 scientific articles published from 1994 to 2017 with innovation in family firms (IFF) as the title of the subject, keywords, and abstract. The authors discuss the results from the perspective of performance indicators and co-authorship visualization, giving a holistic bibliometric overview of the research topic. Research on IFF has emerged as an important study area, with an increasingly established position. The field has attracted researchers and has led to the development of a wide body of literature. This study provides a synthesis and organization of existing knowledge on IFF research.

CÓDIGOS JEL M1,M2

PALABRAS CLAVE Empresa Familiar; Innovación; Revisión bibliométrica Innovación en empresas familiares: una visión bibliométrica holística del campo de investigación

**Resumen** Este artículo presenta un análisis bibliométrico de la innovación en empresas familiares, enfocándose en aspectos que los estudios previos de revisión de literatura no entendieron o evaluaron completamente. Se basa en la evaluación bibliométrica de 207 artículos científicos publicados de 1994 a 2017 con la innovación en las empresas familiares (IEF) como título del tema, las palabras clave y el resumen. Los autores discuten los resultados desde la perspectiva de los indicadores de rendimiento y la visualización de la coautoría, dando una visión general bibliométrica holística del tema de investigación. La investigación sobre IEF ha surgido como un área de estudio importante, con una posición cada vez más establecida. El campo ha atraído a investigadores y ha llevado al desarrollo de un amplio cuerpo de literatura. Este estudio proporciona una síntesis y organización del conocimiento existente sobre la investigación de IEF.

### http://dx.doi.org/10.24310/ejfbejfb.v9i2.5458

Copyright 2019: Gloria Aparicio, Txomin Iturralde, Valeriano Sanchez-Famoso. European Journal of Family Business is an open access journal published in Malaga by UMA Editorial. ISSN 2444-8788 ISSN-e 2444-877X This work is licensed under a Creative Commons Atribution-NonCommercial-ShareAlike 4.0 International License (CC BY-NC-SA 4.0).

### Acknowledgements

Funding for this research was provided by the Spanish Ministry of Science, Innovation & Universities (MCIU/AEI/FEDER-UE) under the Grant Number RTI2018-097579-B-100, by UPV/EHU under the Grant Number GIU16/46, by DFB/BFA and by FESIDE.

### Introduction

Research interest in family firms has increased significantly over the last decade (Debicki, Matherne, Kellermanns, & Chrisman, 2009). This can be attributed partly to the fact that familyowned firms are the oldest form of business organizations (Gersick, Davis, Hampton, & Lansberg, 1997), and the fact that such type of business organizations continue to hold key positions in all economies worldwide (Credit Suisse, 2018; PwC, 2018).

Scholars focusing on family firms have been interested in understanding the distinctive and idiosyncratic characteristics of family firmsexamining such firms based on two strategic theories: resource-based view (RBV) (Wernerfelt, 1995) and agency theory (AT) (Schulze, Lubatkin, Dino, & Buchholtz, 2001; Schulze, Lubatkin, & Dino, 2003). Both of these approaches combined with maintaining family control can influence behavior on innovation in family firms (IFF) (Chrisman, Chua, & Steier, 2005). Nevertheless, research on family business has shifted such that scholars now consider behavioral theories, which highlight that IFF is motivated by non-financial objectives (Miller & Le-Breton-Miller, 2014) such as socio-emotional wealth (SEW) (Gomez-Mejia, Hayes, Nunez-Nickel, Jacobson, & Monavo-Fuentes, 2007) and social capital (SC) (Arregle, Hitt, Sirmon, & Very, 2007; Gedajlovic & Carney, 2010; Pearson, Carr, & Shaw, 2008). Behavioral theories imply that resources are difficult to duplicate (Zahra, Hayton, & Salvato, 2004). In line with this, the concept of "familiness" and family culture may create an environment of trust and shared goals (Dibrell & Moeller, 2011), which can be a possible source of competitive advantage.

On the one hand, it is widely recognized in research on family firms that the first goal of family firms is to pass on the company to the next generation (e.g., Sanchez-Famoso, Maseda, & Iturralde, 2014; Sharma, 2004; Sirmon, Arregle, Hitt, & Webb, 2008). On the other hand, it is a common practice in the majority of family firms for the family to maintain control over the company (Chirico & Salvato, 2016; Cruz & Nordqvist, 2012) through either ownership (Sanchez-Famoso, Akhter, Iturralde, Chirico, & Maseda, 2015) and/or management (D'Amato, 2017; Nordqvist & Melin, 2010; Sanchez-Famoso, Maseda, & Iturralde, 2017). As a result, a family firm will have more than one generation in both management ownership and positions (Drakopoulou Dodd, Anderson, & Jack, 2013; Craig & Dibrell, 2006; Llach & Nordqvist, 2010). These singularities of the vast majority family firms are viewed as positive aspects for accomplishing IFF. However, there are some negative effects that constrain IFF including their conservative posture (Gilinsky, Santini, Lazzeretti, & Eyler, 2008; Habbershon, Williams,

& MacMillan, 2003), risk aversion (De Massis, Sharma, Chua, & Chrisman, 2012), and limited propensity to invest capital in innovation projects (Block, 2010). These relationships are more complex and multidimensional than predicted. Therefore, there is limited current understanding on the topic of IFF and new research seems necessary.

To advance scientific knowledge, researchers emphasize the generally importance of classifying the literature of a research area based on the main trends in the discipline (Bjork, Offer, & Söderberg, 2014). A review of literature engages researchers and practitioners not only by providing a transparent audit trail for legitimizing the order and flow of articles but also by highlighting and updating the landscape. A systematic literature review on IFF already exists. For instance, De Massis, Frattini, and Lichtenthaler (2013) focused on technological innovation. The review by Feranita, Kotlar, and De Massis (2017) contributes to the research on IFF from the perspective of collaborative innovation. More recently, Calabro, Vecchiarini, Gast, Campopiano, De Massi & Kraus (2018) intend to expand the existing understanding of IFF through the construction of a theoretical bridge that includes organizational innovation and a business model. Despite the important contributions made by these literature reviews, the increasing development of computer technology, Internet, and bibliographic electronic databases is what allows for the incorporation of a bibliometric perspective (Baier-Fuentes, Merigó, Amoros, & Gaviria-Marin, 2018; Cobo, Martínez, Gutiérrez-Salcedo, Fujita, & Herrera-Viedma, 2015) in the bibliographic literature review for an in depth understanding of the research area. Bibliometric methods offer systematized and repeatable processes that can help to understand the dissemination of knowledge in a field, while highlighting gaps and opportunities that can help advancements in the field, and also provide objective criteria for assessing research development in a field and represent an important and valuable tool for evaluating scholarship quality and productivity (Cobo et al., 2015)

Thus, the present study aims to add this bibliometric perspective to research on IFF. This provides a synthesis and organization of existing knowledge. First, using performance analysis and, in particular, certain productivity and impact indicators, it is possible to highlight and provide an update on an overview of the research on IFF, revealing patterns in journals, articles, and authors. Second, establishing a network among scholars helps to detect which of them are collaborating in the publications. Third, with a deeper reading of the articles in each group of publications resulting from coauthorship analysis, the authors identify some

Aparicio, G., Iturralde T., Sanchez-Famoso, V. (2019). Innovation in family firms: a holistic bibliometric overview of the research field *European Journal of Family Business*, 9(2), 71-84.

areas of research that have been developed by the most prolific scholars in the area.

### Methodology

Bibliometrics can be defined as a part of scientometrics that utilizes mathematical and statistical methods to study and analyze the scientific activity in a research field (Callon, Courtial, & Laville, 1991). Bibliometric analysis provides objective criteria for evaluating development in a research field, and the technique's importance in assessing academic quality and productivity is increasing (Murgado-Armenteros, Gutiérrez-Salcedo, Torres-Ruiz, & Cobo, 2015). Bibliometric methods have two main uses: performance analysis and science mapping (Cobo, López-Herrera, Herrera-Viedma, & Herrera, 2011). Performance analysis seeks to evaluate the research and publications of individuals and institutions. Science mapping aims to reveal the structure and dynamics of scientific fields (Zudic Æ Carter. 2015). Therefore, bibliometrics contribute to the advancement of science in several ways, such as most relevant scientific recognizing the publications and productive authors, establishing the most cited studies and identifying key themes in the field, and generating indicators of scientific production. This general methodological approach is adapted for the purpose of this study. Thus, to identify some lines of research that have been developed by the most productive researchers in the area, we conduct science mapping based on co-authorship analysis. The methodological design used for this bibliometric review has been developed in a sequential process. First, to begin the study, it is necessary to adopt a systematic search of articles, as this involves the selection of source document. Second, it is important to show the evolution of the number of publications per year in the area because it is the starting point to place the ascent interest of the topic in the scholarship. Third, to show the journals, authors and articles with maximum influence (citations) and contribution (production) in the area, we use performance analysis. This double view to measure the importance of the research allows us to make an important reflection that helps to conduct the final part of the article, that is, coauthorship analysis. This final analysis is focused on the most productive groups of authors that allow for the identification of main research trends.

### Selection of source document

The first step in a bibliometric analysis is to create a database of articles for conducting the analysis. There are several sources for accessing data, including Web of Science (WoS), Scopus, and Google Scholar. The scope of the various databases of scientific documents is different since they do not cover journals in the same wav. The present study only considers bibliographic records obtained from WoS. The authors use this database because it provides a set of metadata that are essential for this type of analysis, including abstracts, references, number of citations, lists of authors, institutions, countries, and the journal impact factor (Carvalho, Fleury, & Lopes, 2013).

To conduct the analysis, we considered articles covering a period of 24 years, that is, from 1994 to 2017 (retrieved on September 3, 2018). We selected 1994 as the starting year because the first paper on IFF with the search combination terms is that of Souder and Thomas (1994). We know that prior to 1994 there were some articles of interest on the topic (Calabro et al., 2018), but these previously published articles do not appear in the systematic search. First, the selection of articles related to IFF was conducted using a combination of key terms such as ("Innovate\*") AND ("family business\*" OR "family firm\*" OR "family enterprise\*" OR "family influence\*" OR "family owned\*" OR "familv controlled\*" OR "family SME\*" OR "family involvement") in the topic tab (including titles, Author Keywords, and abstracts of the bibliographic references) of the WoS

Subsequently, documents from WoS were processed following limitations such as: (1) corpus of the research document was restricted to "articles" and "reviews"; and (2) documents included in the categories "business" or "management" on the WoS. Using such search criteria, we obtained 207 articles that formed the resource for our bibliometric study.

### IFF in the Family Firm Scholarship

The interest of a research topic can be evidenced from two complementary perspectives. First, reflecting the number of published papers and their evolution over time (the most common perspective of a bibliometric performance analysis). Second, considering the relative importance of the topic in a global framework, a family firm is the general reference of scholarship in this case.

Figure 1 shows the evolution of the literature on family firms, on IFF, and the proportion of articles on IFF within the total family firm

research area. We observe that research on IFF is evolving and this development is even more significant on the total number of family firms that are within the scope of this research area. Therefore, the topic of innovation in family business is being consolidated.



Figure 1 Evolution of scientific research on IFF.

Table 1 presents the number of articles per year (TP) and the total number of citations (TC) obtained for them, considering the citation thresholds of 150, 100, 50, 20, 10and 1. Based on the number of articles per year on IFF, the evolution of the field of study shows that the subject area's overall tendency has been ascendant, especially since 2009. Therefore, it is possible to distinguish two periods: the initial period before 2009 (T1) and the expansion period after 2009 (T2). The dates indicate that in the initial period (1994 -2009) only 24 articles had been published, while in the expansion period (2010-2017) there were nearly 183 publications, with 124 belonging to the last three years (70% of the expansion period). All these indicators reinforce that research on IFF is attracting increasing interest among scholars, considering the number of articles. Further, Table 1 will be analyzed in the next section for the most influential articles, because three articles with more than 150 citations and three others with more than 100 citations could be reference articles on the topic. Nearly 25% of the studies did not receive any citations with most of them falling in the latest period.

## Most Influential Journals, Authors and Articles IFF.

This section presents and ranks some indicators on the influence of journals, authors, and articles, based on the information found in WoS. There are two methods to analyze the influence, counting the number of papers published and the number of citations received (Merigo & Yang, 2017). The present article implemented a combined method to measure the importance of research on IFF. It allows making an important reflection that conducts the final part of the article where the conceptual structure of the subject area IFF is analyzed through the main research group of authors in the field.

### Most influential journals

Articles on IFF are published in a wide range of journals. This field has progressed remarkably and has a wide structure of academic resources. Table 2 shows the ranking of the 29 most productive and influential journals in the field of IFF.

 Table 1 Citation structure of research on innovation in family firms.

YEAR	TP	TC	>=150	>=100	>=50	>=20	>=10	>=1
Initial period (T1)								
1994	1	5	0	0	0	0	0	1
1995	0	0	0	0	0	0	0	0
1996	1	108	0	1	1	1	1	1
1997	0	0	0	0	0	0	0	0
1998	0	0	0	0	0	0	0	0
1999	0	0	0	0	0	0	0	0
2000	1	0	0	0	0	0	0	0
2001	1	73	0	0	1	1	1	1
2002	0	0	0	0	0	0	0	0
2003	2	10	0	0	0	0	0	2
2004	1	84	0	0	1	1	1	1
2005	3	442	1	2	2	3	3	3
2006	3	191	0	0	2	3	3	3
2007	3	403	1	1	2	3	3	3
2008	5	305	1	2	2	2	5	5
2009	3	115	0	0	1	2	2	3
			Expa	nsion Pe	riod (T2)			
2010	6	220	0	0	1	4	5	6
2011	9	252	0	1	1	4	5	8
2012	13	515	0	1	4	8	10	13
2013	17	366	0	0	1	7	13	16
2014	14	238	0	0	1	4	10	14
2015	40	377	0	0	1	5	14	36
2016	50	110	0	0	0	1	3	31
2017	34	19	0	0	0	0	0	10
TOTAL	207	3833	3	8	21	49	79	157
% over TC			1,45%	3,86%	10,14%	23,67%	38,16%	75,85%
R: rank; TF	P and T	C: tota	l papers	and cite	s; >= 150,	>=100, >	=50, >= 20	),
>=10, >=1	numbe	er of pa	pers wit	h 150, 10	00, 50, 0,	10 and 1	o more cit	tations.

Specifically, Table 2 shows journals with three or more articles on IFF, as well as those with only one or two articles but with more than 50 or 20 citations, respectively. In addition, some bibliometric indicators, such as the total number of IFF papers (TP), total number of citations (TC), and the ratio of TC/TP are presented. Further, we have included articles with more than 150, 100, 50, 10 citations, and even a single citation. Finally, Table 2 shows the number of articles in each period, that is, T1 and T2, and the 2017 impact factors of the journals.

According to Table 2, it is observed that the most influential journals researching IFF, considering the number of articles published, are two journals with specific topics on family firms, namely *Family Business Review* and *Journal of Family Business Strategy*. It is necessary to note that if we consider the *expansion period* (T2), *Journal of Family business Strategy* publishes a greater number of articles. Table 2 also highlights the importance *Journal of Product Innovation Management* gives to research on IFF, since this journal published eight articles on the topic during the expansion period.

Aparicio, G., Iturralde T., Sanchez-Famoso, V. (2019). Innovation in family firms: a holistic bibliometric overview of the research field *European Journal of Family Business*, 9(2), 71-84.

Table 2 Most influential journals in IFF.

R	JOURNAL	TP	тс	TC/TP	>=150	>=100	>=50	>=20	>=10	>=1	T1	T2	IF
1	FBR	18	1119	62,17	2	2	6	12	14	18	6	12	3,82
2	JFBS	15	81	5,40	0	0	0	1	3	12	0	15	2,61
3	SBE	12	298	24,83	0	0	3	4	6	9	2	10	2,86
4	JPIM	9	195	21,67	0	0	0	4	8	9	1	8	4,31
5	JBR	9	93	10,33	0	0	0	1	3	8	2	7	2,51
6	ETP	7	298	42,57	0	2	2	4	5	7	2	5	5,32
7	APJM	7	65	9,29	0	0	0	2	3	6	0	7	2,47
8	JSBM	6	169	28,17	0	0	2	3	3	5	1	5	3,25
9	ALA	6	1	0,17	0	0	0	0	0	1	0	6	0,62
10	CMR	5	30	6,00	0	0	0	0	2	5	0	5	3,30
11	EJIM	5	8	1,60	0	0	0	0	0	4	0	5	0,67
12	JFBM	5	6	1,20	0	0	0	0	0	3	0	5	
13	JBE	4	85	21,25	0	0	0	2	2	4	0	4	2,92
14	IJEBR	4	40	10,00	0	0	0	1	1	4	0	4	1,19
15	RIBS	4	3	0,75	0	0	0	0	0	1	0	4	
16	JBV	3	320	106,67	0	2	3	3	3	3	1	2	6,00
17	CGIR	3	42	14,00	0	0	0	0	3	3	1	2	2,75
18	BH	3	21	7,00	0	0	0	0	1	3	1	2	1,08
19	MD	3	9	3,00	0	0	0	0	0	2	0	3	1,53
20	IEAMJ	3	6	2,00	0	0	0	0	0	2	0	3	2,41
21	MILI	3	1	0,33	0	0	0	0	0	1	0	3	
22	ERD	2	140	70,00	0	0	1	2	2	2	0	2	2,79
23	MIR	2	36	18,00	0	0	0	0	2	2	1	1	2,28
24	OWC	2	35	17,50	0	0	0	1	1	2	1	1	1,19
25	RMS	2	23	11,50	0	0	0	1	1	2	0	2	1,48
36	JMS	1	163	163,00	1	1	1	1	1	1	1	0	5,33
37	IMM	1	131	131,00	0	1	1	1	1	1	1	0	3,68
38	JCR	1	84	84,00	0	0	1	1	1	1	1	0	3,54
39	SMJ	1	55	55,00	0	0	1	1	1	1	0	1	5,48

Note: R: rank; TP and TC: total papers and cites, TC/TP cites per article. IF: Impact Factor. ; >= 150, >=100, >=50, >= 20, >=10, >idelsol@iefamiliar.comidelsol@iefamiliar.com=1,= number of papers with 150, 100, 50, 0, 10 and 1 o more citations. T1,The first initial period before 2009, T2 the expansion period after that year. FBR, Family Business Review; JFBS, Journal of Family Business Strategy; SBE, Small Business Economics; JPIM, Journal of Product Innovation Management; JBR, Journal of Business Research; ETP, Entrepreneurship Theory and Practice; APJM, Asia Pacific Journal of Management; JSBM, Journal of Small Business Management; ALA, Academia-Revista Latinoamericana de Administracion; CMR, California Management Review; EJIM, European Journal of International Management; JFBM, Journal of Family Business Management; JBE, Journal of Business Ethics; IJEBR, International Journal of Entrepreneurial Behaviour & Research; RIBS, Review of International Business and Strategy; JBV, Journal of Business Venturing; CGIR, Corporate Governance-An International Review; BH, Business History; MD, Management Decision; IEAMJ, International Journal of International Journal of International Journal of International Journal of International Business History; MD, Management Decision; IEAMJ, International Lournal for International Journal of International Review; BH, Business History; MD, Management Decision; IEAMJ, International Journal of Innovation Management; ERD, Entrepreneurship and Regional Development; MIR, Management International Review; JMO, Journal of Management & Organization; JMS, Journal of Management Rev

Another important aspect to consider in this section is the analysis on the total number of citations on IFF. Family Business Review stands out for having the largest number of TCs with 1119 citations, followed by Journal of Business Small Business Economics and Venturing, Entrepreneurship, Theory and Practice, with 320, 298, and 298 citations, respectively. Subsequently there is a third group of journals such as Journal of Product Innovation Management, Journal Small of Business Management, Entrepreneurship and Regional

Development, Journal of Management Studies, and Industrial Marketing Management, which have more than 100 citations. If we analyze the average number of citations per article, those published in Journal of Management Studies are highlighted for their 163 citations; articles published in Industrial Marketing Management with 131 citations, and three published in Journal of Business Venturing with more than 100 citations per article on average. Several factors explain this marked difference between the groups of journals (Baier-Fuentes et al.,

Aparicio, G., Iturralde T., Sanchez-Famoso, V. (2019). Innovation in family firms: a holistic bibliometric overview of the research field. *European Journal of Family Business*, 9(2), 71-84.

2018). The first factor is the overall quality of the journals. Note that most of the journals with a high number of IFF citations have a high impact factor (IF) and are well recognized in their respective fields. A second factor could be the orientation of the journals. For example, all the journals presented in Table 2 have a TC indicator that is greater than 100 and a strong orientation for publishing articles that are related to innovation in companies, research on family firms, or those that have featured issues on management, entrepreneurship, and/or small businesses.

From Table 2, it is interesting to analyze the progress of IFF research in journals. The numbers of papers published are grouped into two periods: T1 and T2. In T1, six articles were published in Family Business Review, two articles each in Small Business Economics, Journal of Business Research and Entrepreneurship Theory and Practice, and one article in ten other journals. Later, we will examine (see Table 3) the articles published in T1 with greater impact. Finally, from a general perspective, the results show that research on IFF has been progressively published in higher number of journals. Specifically, in T1 the 24 articles were published in 16 journals, while the 183 articles published in T2 did so in 75 journals. In T2 a wide variety of journals from covering business and management areas published IFF-based articles to explain their phenomena of study.

### Most cited articles

The number of citations of an article reflects the influence, popularity, and attention received from the scientific community. In this section, we analyze the most-cited articles in journals covering business and management areas in WoS. Table 3 shows the articles with more than 50 citations in WoS -total cites (TC), considering only the citations of the 207 articles (TCIFF), and the citations per article (TC/TP)-.

The three most influential articles of the *initial* period have a significant number of citations (less than 284 and above 163). These citations have accumulated over more than ten years (2005-2017), so the average number of citations is more than 10 or 20 per year. It is most likely that this is not a high volume of citations per article and year, compared to other more established areas; but considering that prior to 2005 there were only seven articles on this topic, it is certain that most of the 200 papers between 2005-2017 frequently cited these three articles. Furthermore, Zhara (2005) is clearly the reference article based on the highest number of citations in WoS (284) and among papers published in the IFF dataset. Analyzing the research areas of this most cited, it is worth mentioning that all these articles have a common

interest in innovation, analyzing it from an organizational point of view, instead of technical or product innovation. Zahra (2005) focuses on the influence of family ownership and CEO founders in promoting entrepreneurship and selective venturing into new market domains. Naldi, Nordqvist, Sjöberg, & Wiklund (2007), the second most referenced article from the dataset's articles (45 citations), follows the same focus of research and addresses a comparison between family and nonfamily firms for their risk-taking tendency. and the relationship between proactiveness, innovativeness, and impact on firm performance. Eddleston, Kellermanns, & Sarathy (2008) show that strategic planning is more important for family firms that lack (technical) innovative capacities. Finally, in T2 it is important to highlight Chua, Chrisman, Steier, & Rau (2012) article, as it contributes to a better understanding of the heterogeneity of family firms by examining how vision and goals, as well as the discretion engendered by family control, influence various competitive strategies, and specifically innovation.

### Most productive and cited authors

The field of IFF is characterized by continuous growth and the participation of a large number of authors. The results indicate that 455 authors contributed toward publishing 207 documents on IFF research, which is an average of 2.20 authors per article. The fact that only 12 authors have published more than three articles indicates that concentration in the field is not high. It is also noteworthy that 85.5% (389/455 authors) published only one article. It is a common result because, as mentioned above, it is a relatively new area of research with growing number of publications that has not yet reached maturity.

To obtain a broader view of IFF research, authors with greater presence and influence, as well as temporal evolution of publications in the field, are determined.

Table 4 illustrates 25 authors who have published three or more papers; 9 authors with two papers but with more than 50 cited; and 9 authors with one paper but with more than 100 cited.

Note that the authors are ordered decreasingly according to their productivity in the field (TP). In the case of a tie, we considered the total number of citations in the field (TC). Furthermore, Table 4 shows TC and TP of each author per period, that is, initial period (T1) and *expansion period* (T2).

Aparicio, G., Iturralde T., Sanchez-Famoso, V. (2019). Innovation in family firms: a holistic bibliometric overview of the research field *European Journal of Family Business*, 9(2), 71-84.

Table 3 List of most cited papers.

R	Title	Authors	Journal	Year	тс	TCIFF	TCIFF/	T1	T2
1	Entrepreneurial risk taking in family firms	Zahra, SA	FBR	2005	284	53	18,7%	Х	
2	Entrepreneurial orientation, risk taking, and	Naldi, Lucia; Nordqvist, Mattias;	FBR	2007	262	45	17,2%	Х	
3	performance in family firms Resource configuration in family firms: Linking resources, strategic planning and technological	Eddleston, Kimberly A.; Kellermanns, Franz Willi; Sarathy, Ravi	JMS	2008	163	23	14,1%	х	
4	Innovativeness among small businesses: Theory	Hausman, A	IWW	2005	131	10	7,6%	Х	
5	Should I stay or should I go? Career choice intentions of students with family business	Zellweger, Thomas; Sieger, Philipp; Halter, Frank	JBV	2011	116		0,0%		Х
6	Sources of Heterogeneity in Family Firms: An Introduction	Chua, Jess H.; Chrisman, James J.; Steier, Lloyd P.; Rau, Sabine B.	ETP	2012	112	17	15,2%		Х
7	Industry characteristics and	Boter, H; Holmquist, C	JBV	1996	108		0,0%	х	
8	The Role of Family Influence in Firms' Strategic Responses to Threat of Imitation	Sirmon, David G.; Arregle, Jean-Luc; Hitt, Michael A.; Webb, Justin W.	ETP	2008	100	23	23,0%	х	
9	R&D investments in family and founder firms:	Block, Joern H.	JBV	2012	96	42	43,8%		х
10	Corporate governance and strategic change in SMEs: The effects of ownership, board composition and top management teams	Brunninge, Olof; Nordqvist, Mattias; Wiklund, Johan	SBE	2007	94		0,0%	x	
11	Long-term orientation: Implications for the entrepreneurial orientation and performance of family businesses	Lumpkin, G. T.; Brigham, Keith H.; Moss, Todd W.	ERD	2010	91	14	15,4%		х
12	Research on Technological Innovation in Family Firms: Present Debates and Future Directions	De Massis, Alfredo; Frattini, Federico; Lichtenthaler, Ulrich	FBR	2013	85	52	61,2%		х
13	Families and innovative consumer behavior: A triadic analysis of sibling and parental influence	Cotte, J; Wood, SL	JCR	2004	84		0,0%	Х	
14	The natural environment, innovation, and firm performance: A comparative study	Craig, Justin; Dibrell, Clay	FBR	2006	84	30	35,7%	х	
15	A 10-year longitudinal investigation of strategy, systems, and environment on innovation in family firms	Craig, JBL; Moores, K; Cassar, G	FBR	2006	75		0,0%	х	
16	Family Firms and Entrepreneurial Orientation in Publicly Traded Firms A Comparative Analysis of the S&P 500	Short, Jeremy C.; Payne, G. Tyge; Brigham, Keith H.; Lumpkin, G. T.; Broberg, J. Christian	FBR	2009	75		0,0%	х	
17	Strategic goals and practices of innovative family businesses	McCann, JE; Leon-Guerrero, AY; Haley, JD	JSBM	2001	73	15	20,5%	Х	
18	Innovativeness in family firms: a family influence perspective	Kellermanns, Franz W.; Eddleston, Kimberly A.; Sarathy, Ravi; Murphy, Fran	SBE	2012	73	26	35,6%		х
19	Entrepreneurial orientation in long-lived family firms	Zellweger, Thomas; Sieger, Philipp	SBE	2012	62	11	17,7%		х
20	Product Innovation in Family versus Nonfamily Firms: An Exploratory Analysis	De Massis, Alfredo; Frattini, Federico; Pizzurno, Emanuele; Cassia, Lucio	JSBM	2015	58	35	60,3%		Х
21	Risk abatement as a strategy for R&D investments in family firms	Patel, Pankaj C.; Chrisman, James J.	SMJ	2014	55	18	32,7%		х
22	The relationship between entrepreneurial orientation and growth: The moderating role of family involvement	Casillas, Jose C.; Moreno, Ana M.	ERD	2010	49	6	12,2%		х
23	The Impact of Family Involvement on the R&D	Munoz-Bullon, Fernando; Sanchez-	FBR	2011	49	23	46,9%		х
24	Family business performance: The effects of gender and management	Danes, Sharon M.; Stafford, Kathryn;	JBR	2007	47		0,0%	х	
25	Corporate Social Performance and Innovation with High Social Benefits: A Quantitative	Wagner, Marcus	JBE	2010	47	6	12,8%		х
26	Analysis The family innovator's dilemma: how family influence affects the adoption of discontinuous	Koenig, Andreas; Kammerlander, Nadine: Enders, Albrecht	AMR	2013	47		0,0%		х
27	technologies by incumbent firms Charting the Future of Family Business	Litz, Reginald A.; Pearson, Allison W.;	FBR	2012	45	19	42,2%		х
28	Technology Acquisition in Family and Nonfamily Firms: A Longitudinal Analysis of	Kotlar, Josip; De Massis, Alfredo; Frattini Federico: Bianchi Mattia:	IPIM	2013	44	24	54 5%		X
29	Spanish Manufacturing Firms	Fang, Hanqing Chrisman, James L: Chua Jess H: De	51 1140	2013		27	J7,J/0		Λ
27	The Ability and Willingness Paradox in Family Firm Innovation	Massis, Alfredo; Frattini, Federico; Wright, Mike	JPIM	2015	43	21	48,8%		Х
R: ra	R: rank. TC: total cites, TCIFF: total cites among articles of the IFF dataset. TC/TP: cites per article								

Aparicio, G., Iturralde T., Sanchez-Famoso, V. (2019). Innovation in family firms: a holistic bibliometric overview of the research field. *European Journal of Family Business*, 9(2), 71-84.

In the *initial period*, Zahra highlights the most cited articles on this topic. It is worth mentioning that Zahra is an important author who has worked in different disciplines of social sciences, particularly in the field of entrepreneurship. The research by Zahra has been diverse, ranging mainly in the three areas of management and strategy- entrepreneurship, international entrepreneurship, and social entrepreneurship (Audretsch, 2015).

Nordqvist is another important author in T1, contributing two articles, but since the author has also published in T2, the total number of articles are four, reaching a total number of citations of TP=4 and TC= 373.

The two articles published in the *initial period*, with Wiklund between others authors, are in category of most-cited articles (see Table 3).

In 2007, Nordqvist and Wiklund published "Entrepreneurial orientation, risk taking, and performance in family firms" in the journal *Family Business Review*, co-authored by Naldi and Sjoberg. This is a reference article focusing on risk taking as an important dimension of entrepreneurial orientation and its impact on family firms with 262 citations (Naldi et al., 2007).

It is also necessary to highlight the influence of Craig, with three articles during each period. He coauthored with Dibrell the article "The natural environment, innovation, and firm performance: A comparative study" (Craig & Dibrell, 2006), and appears in Table 3 as one of the 21 most cited articles on IFF (13<sup>th</sup> position).

In *the expansion period*, the top rankings are for De Massis with 13 articles and Frattini with 8 articles.

De Massis is the author with best combination of productivity and influence in the context of IFF research. He is an author with high productivity and 339 citations. These authors concentrate all their publications in this period.

The article "Research on Technological Innovation in Family Firms: Present Debates and Future Directions" (De Massis, Frattini & Lichtenhaler, 2013) with more than 17 citations per year is reflected in Table 3 as one of the 12 most-cited articles on IFF.

In this article, the authors analyze the state of research in technological innovation in the area of FF.

# Most Productive Groups of Authors with Significant Research Lines in IFF

Science is collaboration and it can be understood as a form of scientific social network where scholars share knowledge.

This network can be represented as a graph in which the nodes are scholars and links are

specific forms of scientific collaboration between them (Sonnenwald, 2007).

Thus, co-authorship is an indicator of scientific collaboration in which authors publish their research outcomes through writing papers. Whenever scholars publish a co-authored article, they have contributed to an individual coauthorship network, which reveals only those authors that have made direct contributions to the article content.

If individual co-authorship networks are analyzed in aggregate form, this network exhibits interconnected relationships among scholars (Li, Liao, & Yen, 2013).

Assuming that co-authorship indicates a level of scientific collaboration, we used network analysis to identify authors in a co-authorship network (Noyons, Moed, & Van Raan, 1999).

We reduced the network focusing on the connections between scholars who had published two or more articles on IFF, and with a minimum of one co-authoring relationships.

Overall, 24 authors meet these restrictions.

The reduced network is represented in a diagram (Figure 2), created with VOSviewer software (version 1.6.9).

This analysis of co-authorships among the most productive authors allows a systematized vision of the main research groups in the IFF field. Focusing on the thematic similarities between the articles by related authors in each cluster, it is possible to categorize the information into themes. In several instances, the themes were identified by analyzing the title, abstract, and keywords of each article. In other cases, an in depth analysis of the article was required.

After identification and content revision, the articles spread among the co-authorship groups are organized into three main research streams.

Aparicio, G., Iturralde T., Sanchez-Famoso, V. (2019). Innovation in family firms: a holistic bibliometric overview of the research field *European Journal of Family Business*, 9(2), 71-84.

 Table 4 The most productive authors and cited authors.

R	AUTHORS	тс	ТР	TC/TP	T1P	T1C	T1C/T1P	T2P	T2C	T2C/T2P
1	De Massis, A.	339	13	26,08	0	0	0,00	13	339	26,08
2	Frattini, F.	269	8	33,63	0	0	0,00	8	269	33,63
3	Craig, Justin B.	229	6	38,17	3	186	62,00	3	43	14,33
4	Chrisman, James J.	233	5	46,60	0	0	0,00	5	233	46,60
5	Fang, H.	88	5	17,60	0	0	0,00	5	88	17,60
6	Kotlar, J.	84	5	16,80	0	0	0,00	5	84	16,80
7	Nordqvist, M.	373	4	93,25	2	356	178,00	2	17	8,50
8	Kellermanns, F. W.	273	4	68,25	1	163	163,00	3	110	36,67
9	Kammerlander, N.	91	4	22,75	0	0	0,00	4	91	22,75
10	Wright, M.	63	4	15,75	0	0	0,00	4	63	15,75
11	Kraus, S.	39	4	9,75	0	0	0,00	4	39	9,75
12	Memili, E.	31	4	7,75	0	0	0,00	4	31	7,75
13	Zellweger, T.	201	3	67,00	0	0	0,00	3	201	67,00
14	Chua, Jess H.	158	3	52,67	0	0	0,00	3	158	52,67
15	Dibrell, C.	113	3	37,67	1	84	84,00	2	29	14,50
16	Uhlaner, L. M.	52	3	17,33	1	4	4,00	2	48	24,00
17	Matzler, K.	25	3	8,33	0	0	0,00	3	25	8,33
18	Welsh, Dianne H. B.	20	3	6,67	0	0	0,00	3	20	6,67
19	Iturralde, T.	18	3	6,00	0	0	0,00	3	18	6,00
20	Maseda, A.	18	3	6,00	0	0	0,00	3	18	6,00
21	Sanchez-Famoso, V.	18	3	6,00	0	0	0,00	3	18	6,00
23	Dieguez-Soto, J.	6	3	2,00	0	0	0,00	3	6	2,00
24	Calabro, A.	3	3	1,00	0	0	0,00	3	3	1,00
25	Lopez-Fernandez, M.	2	3	0,67	0	0	0,00	3	2	0,67
22	Serrano-Bedia, A.	2	3	0,67	0	0	0,00	3	2	0,67
26	Wiklund, J.	356	2	178,00	2	356	178,00	0	0	0,00
27	Eddleston, K. A.	236	2	118,00	1	163	163,00	1	73	73,00
28	Sarathy, R.	236	2	118,00	1	163	163,00	1	73	73,00
29	Sieger, P.	178	2	89,00	0	0	0,00	2	178	89,00
30	Brigham, K. H.	166	2	83,00	1	75	75,00	1	91	91,00
31	Lumpkin, G. T.	166	2	83,00	1	75	75,00	1	91	91,00
32	Steier, L.	114	2	57,00	0	0	0,00	2	114	57,00
33	Block, J. H.	106	2	53,00	0	0	0,00	2	106	53,00
34	Moores, K	102	2	51,00	2	102	51,00	0	0	0,00
35	Zahra, SA	284	1	284,00	1	284	284,00	0	0	0,00
36	Naldi, L.	262	1	262,00	1	262	262,00	0	0	0,00
37	Sjoberg, K.	262	1	262,00	1	262	262,00	0	0	0,00
38	Hausman, A	131	1	131,00	1	131	131,00	0	0	0,00
39	Halter, F.	116	1	116,00	0	0	0,00	1	116	116,00
40	Rau, S. B.	112	1	112,00	0	0	0,00	1	112	112,00
41	Boter, H	108	1	108,00	1	108	108,00	0	0	0,00

Aparicio, G., Iturralde T., Sanchez-Famoso, V. (2019). Innovation in family firms: a holistic bibliometric overview of the research field. *European Journal of Family Business*, 9(2), 71-84.



Figure 2	. Co-authored	research	on IFF
----------	---------------	----------	--------

 Table 5
 Co-authorship groups and associated research streams.

Cluster 1	AUTHORS Craig, J.; Dibrell, C.; Kraus, S.; Calabro, A.	Research Topics Internal factors of IFF
2	Nordqvist, M.; De Massis, A.; Frattini, F.; Kotlar, J.	Internal Factors of IFF, External Factors of IFF
		Advances in IFF
3	Chua, J. H.; Chrisman, J.; Wright, M.	Internal Factors of IFF
4	Sanchez-famoso, V.; Iturralde T.; Maseda, A.	Internal Factors of IFF
		Advances in IFF
5	Memili, Esra; Fang, Hanqing; Welsh, Dianne	Internal Factors of IFF
		Advances in IFF
6	Serrano-Bedia, A., Lopez- Fernandez, M.;	Internal Factors of IFF
7	Kammerlander,; Zellweger,	Internal Factors of IFF

### Main Research Streams in IFF Coauthorship Groups

In this section, we develop a content overview of each of the three research streams (Table 5), identified from the publications of the coauthorship groups. The three research streams considered are internal factors of IFF, external factors of FF, and advances in IFF.

It is important to emphasize that the same coauthorship group may have developed more than one research stream, and different co-authorship groups may have worked on the same research stream.

### **Cluster A: Internal Factors of IFF**

The studies in this cluster focus on identifying the internal characteristics of family firms that affect innovation behavior. These investigations draw on resource-based view (RBV) (Craig, Dibrell, & Garret, 2014; Basco & Calabro, 2016), upper echelons theory (Craig, Pohjola, Kraus, & Jensen 2014), stewardship theory (Neubaum, Thomas, Dibrell, & Craig, 2017), socio-emotional wealth (SEW) (Calabro, Minola, Campopiano, & Pukall, 2016), and social capital theory (Sanchez-Famoso, Maseda & Iturralde 2014, 2017; Sanchez-Famoso, Iturralde, & Maseda, 2015). Using these theories, this article highlights the heterogeneity of family firms and the role of internal attributes in business strategy (Craig & Dibrell, 2006; Craig, Moores, & Cassar, 2006). In this sense, each family firm possesses a unique set of internal characteristics such as family culture, noneconomic goals, internal relationships, and board composition, which are acquired and developed over time. These family firms' singularities further determine the degree of efficiency and, therefore, strategic decisions such as involvement in innovation.

Family culture is a possible measure of the firm's organizational resources (Craig et al., 2014). It represents the knowledge accumulated throughout the family history and is related to

Aparicio, G., Iturralde T., Sanchez-Famoso, V. (2019). Innovation in family firms: a holistic bibliometric overview of the research field *European Journal of Family Business*, 9(2), 71-84.

better management of fluent communication, which directly relates to innovation and creativity (Neubaum et al., 2017).

Non-economic goals are indeed related to less investments in R&D because family members want to protect their SEW, thus avoiding taking innovation risks, which affect the firm's innovations (Calabro et al., 2016).

Internal relationships refer to relationships inside the family firms (Sanchez-Famoso, Maseda & Iturralde 2014, 2017; Sanchez-Famoso, Iturralde, & Maseda, 2015). These relationships include not only those between family members but also between non-family members; two different groups that coexist in the majority of family firms (Mitchell, Morse, & Sharma, 2003; Sanchez-Famoso et al., 2015). The shared vision of the business, same language, and intensity of the relationships are antecedents of IFF. Ownership and family management (Sanchez-Famoso et al., 2015, Sanchez-Famoso, Maseda & Iturralde, 2017) play a moderate role between internal relationships and innovation. This is could be attributed to the fact that if more family members are part of the top management team (TMT), the firm's innovation propensity increases; however, according to Lohe and Calabro (2017) conflicts can emerge due to the different perspectives expressed by the TMT members.

Overall, the board has a relationship with the firm's innovation behavior (Lopez-Fernandez, Serrano-Bedia, & Perez-Perez, 2016). The board's composition (number of family members, non-family members, and interlocks) is the main characteristic that could affect strategic decisions (Serrano-Bedia, Lope-Fernandez, & Garcia-Piqueres, 2016). The chief executive officer's (CEO) tenure (Lopez-Fernandez, Serrano-Bedia, & Palma-Ruiz, 2016) may have less incentive to invest in R&D in family firms due to SEW protection.

Another internal factor that matters is how the creation of family firms affects innovation (Brunninge, Nordqvist, & Wiklund, 2007; Zellweger & Sieger, 2012). In this respect, entrepreneurial behaviors influence both innovation and long-term success (Kammerlander & Ganter, 2015; Kammerlander, Dessi, Bird, Floris, & Murru, 2015). If the family firm is created with the main aim of preserving SEW, the firm can continue across generations (Welsh, Memili, Kaciak, & Al Sadoon, 2014); however, firms can fail in starting or operating new ideas and innovations (Memili, Welsh, & Kaciak, 2014). In this sense, family support in difficult times is fundamental for continuing with the company (Memili, Fang, & Welsh, 2015). The authors who focused on this internal aspect of family firms are based in specific theories such as planned behavior theory (Zellweger, Sieger, & Halter, 2011; Zellweger & Sieger, 2012) and organizational psychological capital (Memili, Welsh, & Kaciak, 2014, Memili, Fang & Welsh, 2015; Welsh, et al., 2014)

### **Cluster B: External Factors of IFF**

A central finding in the literature is that innovation in family firms depends on external resources. In this sense, IFF is also a process that derives from the strengthening of the family firm's core competences. These resources include not only financial or human capital but also connections with other firms and institutions (Kotlar, De Massis, Fratiini, Bianchi, & Fang, 2013; Kotlar, Fang, De Massis, & Frattini, 2014). In a competitive era, success depends on the ability to produce new or improved products and tacit knowledge constitutes the most important basis for innovation-based value creation (Memili, Fang, Chrisman, & De Massis, 2015). However, it is difficult to exchange a firm's innovation activities over long distances (Chrisman, Fang, Kotlar, & De Massis, 2015). Therefore, innovation activities are collective achievements that require key roles from numerous entrepreneurs (Memili et al., 2015), rather than an isolated decision within a single firm. External financial availability liberates family firms from the need to generate funds internally by helping them raise capital from external sources at a reasonable cost (Welsh et al., 2014).

### Cluster C: Advances in IFF

In this cluster, advanced research studies on IFF appear. They develop a theory at the important intersection of family firms and innovation (De Massis, Frattini & Lichtenhaler, 2013). As it is shown that innovation helps family firms to respond effectively to shifts in market dynamism (Chrisman, Chua, De Massis, Frattini, & Wright, 2015; De Massis, Frattini, Kotlar, Petruzzelli, & Wright, 2016), articles in this cluster call for some research streams to study why innovation makes it possible to recognize market dynamism (De Massis, Di Minin, & Frattini, 2015). The theories used in this research stream are agency, RBV, grounded, and SEW. The singularities of family firms may be important precursors of the innovation. They call for undertaking research considering the essence of focus, which explains why there is no common result on how to make innovation happen in family firms. They agree that there is a need to address how family firms manage radical innovations and explore disruptive innovation (Brunninge et al., 2007; Casprini, De Massis, Di Minin, Frattini, & Piccaluga, 2017; De Massis, Frattini & lichtenhaler, 2013). Thus,

Aparicio, G., Iturralde T., Sanchez-Famoso, V. (2019). Innovation in family firms: a holistic bibliometric overview of the research field. *European Journal of Family Business*, 9(2), 71-84.

building on this, it would be interesting to explore the relations between ambidexterity and new product development management associated with loose-coupled partners and complementors, and the limits and risks of a pivot strategy. On the other hand, there is a need to focus on the critical capabilities that are important for innovation, such as dynamic and integration capability (De Massis, Frattini, Pizzurno, & Cassia, 2015; Fitz-Koch & Nordqvist, 2017). These streams of research are expected to not only advance theoretical understanding but also improve how family firms manage and organize innovation.

### Conclusion

This study examined the scientific research in IFF between 1994 and 2017, based on publications available in the WoS database. The evolution in the field of study shows that the topic's overall tendency has been ascendant, especially since 2009. Thus, it is possible to distinguish two different periods: the *initial period*, prior to 2009 (T1), and the *expansion period*, after the year 2009 (T2).

Although this study is not the first attempt to conduct a comprehensive and systematic review of academic IFF research, this article adds a bibliometric perspective to the IFF research topic, providing a synthesis and organization of existing knowledge in this research stream. Thus, this article provides a broad view of the research in this field and attempts to contribute to the increased generation of literature on IFF and, in doing so, facilitates the work of academics, consultants, family business students, entrepreneurs and sociologists. According to the study's limitations, the information presented in this study is expected to be complementary and informative to other bibliographic literature reviews. Second, the dataset was gathered exclusively from the WoS database. Thus, data from other sources (e.g., Scopus and Google Scholar) were not used. Finally, co-authorship analysis has been used to identify the main research streams instead of other mapping techniques, such as co-citation and citation, to complete the bibliometric overview.

### References

- Arregle, J. L., Hitt, M. A., Sirmon, D. G., & Very, P. (2007). The development of organizational social capital: Attributes of family firms. *Journal of Management Studies*, 44(1), 73-95.
- Audretsch, D. (2015). Shaker a. Zahra: Pioneering entrepreneurship scholar. Small Business Economics, 44(4), 721-725.
- Baier-Fuentes, H., Merigó, J. M., Amoros, J. E., & Gaviria-Marin, M. (2018). International entrepreneurship: A bibliometric overview. International Entrepreneurship and Management Journal, <u>https://doi.org/10.1007/s11365-017-0487</u>

- Basco, R., & Calabro, A. (2016). Open innovation search strategies in family and non-family SMEs: Evidence from a natural resource-based cluster in chile. Academia Revista Latinoamericana De Administración, 29(3), 279-302.
- Bjork, S., Offer, A., & Söderberg, G. (2014). Time series citation data: The nobel prize in economics. *Scientometrics*, 98(1), 185-196.
- Block, J. (2010). Family management, family ownership, and downsizing: Evidence from S&P 500 firms. *Family Business Review*, 23(2), 109-130.
- Brunninge, O., Nordqvist, M., & Wiklund, J. (2007). Corporate governance and strategic change in SMEs: The effects of ownership, board composition and top management teams. *Small Business Economics*, 29(3), 295-308.
- Calabro, A., Minola, T., Campopiano, G., & Pukall, T. J. (2016). Turnin innovativeness into domestic and international corporate venturing: The moderating effect of high family ownership and influence. *European Journal of International Management*, 10(5), 505-533.
- Calabro, A., Vecchiarini, M., Gast, J., Campopiano, G., De Massis, A., & Kraus, S. (2018). Innovation in family firms: A systematic literature review and guidance for future research. International Journal of Management Reviews, DOI: 10.1111/ijmr.12192
- Callon, M., Courtial, J. P., & Laville, F. (1991). Coword analysis as a tool for describing the network of interactions between basic and technological research: The case of polymer chemsitry. *Scientometrics*, 22(1), 155-205.
- Carvalho, M., Fleury, A. C., & Lopes, A. P. (2013). An overview of the literature on technology roadmapping (TRM): Contributions and trends. *Technological Forecasting and Social Change*, 80(7), 1418-1437.
- Casprini, E., De Massis, A., Di Minin, A., Frattini, F., & Piccaluga, A. (2017). How family firms execute open innovation strategies: The Loccioni case. Journal of Knowledge Management, 21(6), 1459-1485.
- Chirico, F., & Salvato, C. (2016). Knowledge internalization and product development in family firms: When relational and affective factors matters. Entrepreneurship Theory and Practice, 40(1), 201-229.
- Chrisman, J. J., Chua, J. H., De Massis, A., Frattini, F., & Wright, M. (2015). The ability and willingness paradox in family firm innovation. *Journal of Product Innovation Management*, 32(3), 310-318.
- Chrisman, J. J., Chua, J. H., & Steier, L. P. (2005). Resources and consequences of distinctive familiness: An introduction. *Entrepreneurship Theory and Practice*, 29(3), 237-247.
- Chrisman, J. J., Fang, H., Kotlar, J., & De Massis, A. (2015). A note on family influence and the adoption of discontinuous technologies in family firms. *Journal of Product Innovation Management*, 32(3), 384-388.
- Chua, J. H., Chrisman, J. J., Steier, L. P., & Rau, S.
  B. (2012). Sources of heterogeneity in family firms: An introduction. *Entrepreneurship Theory and Practice*, 36(6), 1103-1113.
- Cobo, M. J., Lopez-Herrera, A. G., Herrera-Viedma, E., & Herrera, F. (2011). Sciencience mapping software tools: Review, analysis, and cooperative study among tools. Journal of the American Society for Information Science and Technology, 2011(7), 1382-1402.

Aparicio, G., Iturralde T., Sanchez-Famoso, V. (2019). Innovation in family firms: a holistic bibliometric overview of the research field *European Journal of Family Business*, 9(2), 71-84.

- Cobo, M. J., Martinez, M. A., Gutierrez-Salcedo, M., Fujita, H., & Herrera-Viedma, E. (2015). 25 years at knowledge-based systems: A bibliometric analysis. *Knowledge-Based Systems*, 80(1), 3-13.
- Craig, J. B., & Dibrell, C. (2006). The natural environment, innovation, and firm performance: A comparative study. *Family Business Review*, 19(4), 275-288.
- Craig, J. B., Dibrell, C., & Garrett, R. (2014). Examining relationships among family influence, family culture, flexible planning systems, innovativeness and firm performance. *Journal of Family Business Strategy*, 5(3), 229-238.
- Craig, J. B., Moores, K., & Cassar, G. (2006). A 10year longitudinal investigation of strategy, systems, and environment on innovation output in family and non-family firms. *Family Business Review*, 19(2), 169-169.
- Craig, J. B., Pohjola, M., Kraus, S., & Jensen, S. H. (2014). Exploring relationships among proactiveness, risk-taking and innovation output in family and non-family firms. *Creativity and Innovation Management*, 23(2), 199-210.
- Credit Suisse Research Institute. (2018). The CS family 1000 in 2018. Zurich: Credit Suisse Group A6
- Cruz, C., & Nordqvist, M. (2012). Entrepreneurial orientation in family firms: A generational perspective. Small Business Economics, 38(1), 33-49.
- D'Amato, A. (2017). Do women perform better in family firms? Exploring the moderating role of family firm's status. *International Journal of Wine Business Research*, 29(3), 299-315.
- De Massis, A., Di Minin, A., & Frattini, F. (2015). Family-driven innovation: Resolving the paradox in family firms. *Califronia Management Review*, 58(1), 5-19.
- De Massis, A., Frattini, F., Kotlar, J., Petruzzelli, A. M., & Wright, M. (2016). Innovation through tradiion: Lessons from innovative family businesses and directions for future research. Academy Management Perspectives, 30(1), 93-116.
- De Massis, A., Frattini, F., & Lichtenthaler, U. (2013). Research on technological innovation in family firms: Present debates and future directions. *Family Business Review*, 26(1), 10-31.
- De Massis, A., Frattini, F., Pizzurno, E., & Cassia, L. (2015). Product innovation in family versus nonfamily firms: An exploratory analysis. *Journal of Small Business Management*, 53(1), 1-36.
- De Massis, A., Sharma, P., Chua, J. H., & Chrisman, J. J. (2012). Family business studies: An annotated bibliography. Cheltenham: Edward Elgar.
- Debicki, B. J., Matherne III, C. F., Kellermanns, F. W., & Chrisman, J. J. (2009). Family business research in the new millennium. An overview of the who, the where, the what, and the why. *Family Business Review*, 22(2), 151-166.
- Dibrell, C., & Moeller, M. (2011). The impact of a service-dominant focus strategy and stewardship culture on organizational innovativeness in family-owned businesses. *Journal of Family Business Strategy*, 2(1), 43-51.

- Drakopoulou Dodd, S., & Anderson, A. (2013). Being in time and the family owned firm. Scandinavian Journal of Management, 29(1), 35-47.
- Eddleston, K. A., Kellermanns, F. W., & Sarathy, R. (2008). Resource configuration in family firms: Linking resources, strategic planning and technological opportunities to performance. *Journal* of Management Studies, 45(1), 26-50.
- Feranita, F., Kotlar, J., & De Massis, A. (2017). Collaborative innovation in family firms: Past research, current debates and agenda for future research. Journal of Family Business Strategy, 8(3), 137-156.
- Fitz-Koch, S., & Nordqvist, M. (2017). The reciprocal relationship of innovation capabilities and socioemotional wealth in a family firm. *Journal of Small Business Management*, 55(4), 547-570.
- Gedajlovic, E. R., & Carney, M. (2010). Markets, hierarchies, and families: Toward a transaction cost theory of the family firm. *Entrepreneurship Theory* and Practice, 34(6), 1145-1172.
- Gersick, K. F., Davis, J. A., Hampton, M. A., & Lansberg, I. (1997). *Generation to generation: Life cycles of the family business*. Cambridge: Harvard Business School Press.
- Gilinsky, A., Santini, C., Lazzeretti, L., & Eyler, R. (2008). Desperately seeking serendipity. Exploring the impact of country location on innovation in the wine industry. *International Journal of Wine Business Research*, 20(4), 302-320.
- Gomez-Mejia, L. R., Haynes, K. T., Nunez-Nickel, M., Jacobson, K. J., & Monayo-Fuentes, J. (2007). Socioemotional wealth and business risks in familycontrolled firms: Evidence from Spanish olive oil mills. Administrative Science Quartely, 52(1), 106-137.
- Habbershon, T. G., Williams, M. L., & MacMillan, I. C. (2003). A unified systems perspective of family firm performance. *Journal of Business Venturing*, 18(4), 451-465.
- Kammerlander, N., Dessi, C., Bird, M., Floris, M., & Murru, A. (2015). The impact of shared stories on family firm innovation: A multicase study. *Family Business Review*, 28(4), 332-354.
- Kammerlander, N., & Ganter, M. (2015). An attention-based view of family firm adaptation to discontinuous technological change: Exploring the role of family CEOs' noneconomic goals. Journal of Product Innovation Management, 32(3), 361-383.
- Kotlar, J., De Massis, A., Frattini, F., Bianchi, M., & Fang, H. (2013). Technology acquisition in family and nonfamily firms: A longitudinal analysis of Spanish manufacturing firms. Journal of Product Innovation Management, 30(6), 1073-1088.
- Kotlar, J., Fang, H., De Massis, A., & Frattini, F. (2014). Profitability goals, control goals, and the R&D investment decisions of family and nonfamily firms. Journal of Product Innovation Management, 31(6), 1128-1145.
- Li, E., Liao, C. H., & Yen, H. R. (2013). Co-authorship networks and research impact: A social capital perspective. *Research Policy*, 42(9), 1515-1530.
- Llach, J., & Nordqvist, M. (2010). Innovation in family and non-family businesses: A resource perspective.

Aparicio, G., Iturralde T., Sanchez-Famoso, V. (2019). Innovation in family firms: a holistic bibliometric overview of the research field. *European Journal of Family Business*, 9(2), 71-84.

International Journal of Entrepreneurial Venturing, 2(3), 381-399.

- Lohe, F. W., & Calabro, A. (2017). Please do not disturb! Differentiating board tasks in family and non-family firms during financial distress. *Scandinavian Journal of Management*, 33(1), 36-49.
- Lopez-Fernandez, M. C., Serrano-Bedia, A. M., & Palma-Ruiz, M. (2016). What hampers innovation in mexican family fims? Academia Revista Latinoamericana De Administración, 29(3), 255-278.
- Lopez-Fernandez, M. C., Serrano-Bedia, A. M., & Perez-Perez, M. (2016). Entrepreneurship and family firm research: A bibliometric analysis of an emerging field. Journal of Small Business Management, 54(2), 622-639.
- Memili, E., Fang, H., Chrisman, J. J., & De Massis, A. (2015). The impact of small-and medium-sized family firms on economic growth. *Small Business Economics*, 45(4), 771-785.
- Memili, E., Fang, H., & Welsh, D. H. B. (2015). Value creation and value appropriation in innovation process in publicly-traded family firms. *Management Decision*, 53(9), 1921-1952.
- Memili, E., Welsh, D. H. B., & Kaciak, E. (2014). Organizational psychological capital of family franchise firms through the lens of the leadermember exchange theory. *Journal of Leathership & Organizational Studies*, 21(2), 200-209.
- Merigó, J. M., & Yang, J. (2017). A bibliometric analysis of operations research and management science. *Omega*, 73(1), 37-48.
- Miller, D., & Le Breton-Miller, I. (2014). Deconstructing socioemotional wealth. Entrepreneurship Theory and Practice, 38(4), 713-720.
- Mitchell, R. K., Morse, E. A., & Sharma, P. (2003). The transacting cognitions of nonfamily employees in the family businesses setting. *Journal of Business Venturing*, 18(4), 533-551.
- Murgado-Armenteros, E. M., Gutierrez-Salcedo, M., Torres-Ruiz, F. J., & Cobo, M. J. (2015). Analysing the conceptual evolution of qualitative marketing research through science mapping analysis. *Scientometrics*, 102(1), 519-557.
- Naldi, L., Nordqvist, M., Sjöberg, K., & Wiklund, J. (2007). Entrepreneurial orientation, risk taking, and performance in family firms. *Family Business Review*, 20(1), 33-47.
- Neubaum, D. O., Thomas, C. H., Dibrell, C., & Craig, J. B. (2017). Stewardship climate scale: An assessment of reliability and validity. *Family Business Review*, 30(1), 37-60.
- Nordqvist, M., & Melin, L. (2010). The promise of the strategy as practice perspective for family business strategy research. *Journal of Family Business Strategy*, 1(1), 15-25.
- Noyons, E. C., Moed, H. F., & Van Raan, A. F. (1999). Integrating research performance analysis and science mapping. *Scientometrics*, 46(3), 591-604.
- Pearson, A. W., Carr, J. C., & Shaw, J. C. (2008). Toward a theory of familiness: A social capital perspective. Entrepreneurship Theory and Practice, 32(6), 949-969.
- PwC Global Family Business Survey. (2018). The values effect. How to build a lasting competitive advantage through your values and purpose in a digital age. Retrieved from www.pwc.com/fambizsurvey2018
- Sanchez-Famoso, V., Akhter, N., Iturralde, T., Chirico, F., & Maseda, A. (2015). Is non-family

social capital also (or *especially*) important for family firm performance? *Human Relations*, 68(11), 1713-1743.

- Sanchez-Famoso, V., Iturralde, T., & Maseda, A. (2015). The influence of family and non-family social capital on firm innovation: Exploring the role of family ownership. European Journal of International Management, 9(2), 240-262.
- Sanchez-Famoso, V., Maseda, A., & Iturralde, T. (2014). The role of internal social capital in organizational innovation. An empirical study of family firms. European Management Journal, 32(6), 950-962.
- Sanchez-Famoso, V., Maseda, A., & Iturralde, T. (2017). Family involvement in top management team: Impact on relationships between internal social capital and innovation. *Journal of Management & Organization*, 23(1), 136-162.
- Schulze, W. S., Lubatkin, M., Dino, R. D., & Buchholtz, A. K. (2001). Agency relationships in family firms. Organization Science, 12(2), 99-116.
- Schulze, W. S., Lubatkin, M., & Dino, R. N. (2003). Towads a theory of agency and altruism in family firms. *Journal of Business Venturing*, 18(4), 473-490.
- Serrano-Bedia, A. M., Lopez-Fernandez, M. C., & Garcia-Piqueres, G. (2016). Analysis of the relationship between sources of knowledge and innovation performance in family firms. *Innovation*, 18(4), 489-512.
- Sharma, P. (2004). An overview of the field of family business studies: Current status and directions for the future. *Family Business Review*, *17*(1), 1-36.
- Sirmon, D. G., Arregle, J. L., Hitt, M. A., & Webb, J. W. (2008). The role of family influence in firms' strategic response to competitive threat. *Entrepreneurship Theory and Practice*, 32(6), 979-998.
- Sonnenwald, D. H. (2007). Scientific collaboration. Annual Review of Information Science and Technology, 41(1), 643-681.
- Souder, W. E., & Thomas, R. J. (1994). Significant issues for the future of product innovation. *Journal* of Product Innovation Management, 11(4), 344-353.
- Welsh, D. H. B., Memili, E., Kaciak, E., & Al Sadoon, A. (2014). Saudi women entrepreneurs: A growing economic segment. *Journal of Business Research*, 67(5), 758-762.
- Wernerfelt, B. (1995). The resource-based view of the firm: Ten years after. *Strategic Management Journal*, 16(3), 171-174.
- Zahra, S. A. (2005). Entrepreneurial risk taking in family firms. *Family Business Review*, 18(1), 23-40.
- Zahra, S. A., Hayton, J. C., & Salvato, C. (2004). Entrepreneurship in family vs. non-family firms: A resource-based analysis of the effect of organizational culture. Entrepreneurship Theory and Practice, 28(4), 363-381.
- Zellweger, T., & Sieger, P. (2012). Entrepreneurial orientation in long-lived family firms. *Small Business Economics*, 38(1), 67-84.
- Zellweger, T., Sieger, P., & Halter, F. (2011). Should I stay or should I go? Career choice intentions of students with family business background. *Journal* of Business Venturing, 26(5), 521-536.
- Zupic, I., & Carter, T. (2015). Bibliometric methods in management organization. Organizational Research Methods, 18(3), 429-472.

Aparicio, G., Iturralde T., Sanchez-Famoso, V. (2019). Innovation in family firms: a holistic bibliometric overview of the research field *European Journal of Family Business*, 9(2), 71-84.



EUROPEAN JOURNAL OF FAMILY BUSINESS

0

. **.** .

http://www.revistas.uma.es/index.php/ejfb

# An Analysis of Open Innovation Determinants: the Case Study of Singapore based Family owned Enterprises

Annie Koh<sup>ab\*</sup>, Esther Kong<sup>ab</sup>, Giuseppe Timperio<sup>ab</sup>

<sup>a</sup>Singapore Management University (Singapore)

<sup>b</sup>Business families institute (Singapore)

Received 26 February 2018; accepted 19 December 2019

JEL CLASSIFICATION M19, O31, O32

KEYWORDS Open Innovation; Innovation Determinants; Family Business; Asia-Pacific; Tradition in Innovation Abstract Family businesses play an important role in the growth of global economy, and while they are arguably perceived as a conservative form of organization with high risk aversion and reluctance to change, counterintuitive empirical evidence show that they are most effective in ideation and commercialization of innovation projects. In the current business environment of rapid change in work patterns, fast adoption of enabling technologies for seamless collaborations across industry and geography, along with intense competition and high uncertainty, enterprises have no choice but to maximize returns on innovation investments. Therefore, they are increasingly dependent on an ecosystem-based approach to innovation management, which has shown greater likelihood to create radical innovations and enable profit generation.

The objective of this paper is to analyse determinants of open innovation practices in family-owned enterprises in consideration of the joint effect of in-company enablers and external factors. Drawing on a sample of 33 Singapore based family-owned firms, our findings confirmed the key drivers such as family and business culture, access to external funds, government supported initiatives, market dynamics, partnership, network, family capital, and external network. Managerial implications about the necessity to leverage both environmental determinants and internal innovation capabilities to foster novel business ideas are also highlighted in the conclusion of the paper.

### CÓDIGOS JEL M19, O31, O32

PALABRAS CLAVE Innovación Colaborativa; Factores de Innovación; Empresa Familiar; Asia Pacífica; Innovación Tradicional

## Un análisis de los factores determinantes de la innovación: el estudio de caso de las empresas familiares con sede en Singapur

Resumen Las empresas familiares desempeñan un papel importante en el crecimiento de la economía global, y aunque posiblemente se las perciba como una forma conservadora de organización con alta aversión al riesgo la evidencia empírica muestra que son más efectivas creando y comercializando proyectos de innovación. En el entorno empresarial actual de cambio rápido en los patrones de trabajo, la adopción rápida de tecnologías habilitadoras para colaboraciones fluidas en toda la industria y la geografía, junto con la intensa competencia y la alta incertidumbre, las empresas no tienen más remedio que maximizar los rendimientos de las inversiones en innovación. Por lo tanto, dependen cada vez más de un enfoque basado en el ecosistema para la gestión de la innovación, que ha demostrado una mayor probabilidad de crear innovaciones radicales y permitir la generación de ganancias. El objetivo de esta investigación es analizar los determinantes de las prácticas de innovación en las empresas familiares en consideración del efecto conjunto de los facilitadores internos y los factores externos. Basándose en una muestra de 33 empresas familiares con sede en Singapur, nuestros hallazgos confirmaron los impulsores clave como la cultura familiar y empresarial, el acceso a fondos externos, las iniciativas respaldadas por el gobierno, la dinámica del mercado, la asociación, la red, el capital familiar y la red externa. Las implicaciones gerenciales sobre la necesidad de aprovechar los determinantes ambientales y las capacidades de innovación interna para fomentar nuevas ideas de negocio también se destacan en la conclusión del artículo.

http://dx.doi.org/10.24310/ejfbejfb.v9i2.5678

Copyright 2019: Annie Koh, Esther Kong, Giuseppe Timperio.

European Journal of Family Business is an open access journal published in Malaga by UMA Editorial. ISSN 2444-8788 ISSN-e 2444-877X This work is licensed under a Creative Commons Atribution-NonCommercial-ShareAlike 4.0 International License (CC BY-NC-SA 4.0).

### Introduction

Traditionally, family businesses around the world have often been perceived as less innovative compared to their non-family owned counterparts due to their apparent risk aversion and reluctance to change [ (Gaskell, 2018), (Marín, Hernández, del Valle, & Castillo. M., 2016)]. This is however open to debate and, in fact, counterintuitive empirical evidence in recent years show that family enterprises are the most innovative organizations among all. They are not only the ones securing the greatest number of patents and new products, but are also most effective in ideation and commercialization of these innovation projects (Duran, Kammerlander, van Essen, & Zellweger, 2015).

Innovation is not a new phenomenon. Arguably, as written by Fagerberg & Mowery (2006), innovation "is as old as mankind itself". Inherently, human beings have the tendency to think and develop new and better ways to do things and to experiment in practice and, as such, research in business innovation have proliferated in recent years, with a steep spike in publications in the early 2000s [ (De Massis, Sharma, Chua, & Chrisman, 2012), (Gunday, Ulusoy, Kilic, & Alpkan, 2011), (Fagerberg & Mowery, 2006)]. The acute competition among firms registered since the turn of the century, the rapid growth of internet and its contribution to globalization and technological advancement, have in fact brought about greater attention on innovation practices, and the ability to create newness is nowadays regarded as a fundamental area in corporate strategy and the main provider of competitive edge (Lopez-Fernandez, Serrano-Bedia, & Gómez-López, 2015).

It is in fact a broadly recognized paradigm that innovation plays the pivotal role of enabling firms to pursue the threefold objective of topline growth, bottom-line growth, and business sustainability in the long run [ (BFI@SMU; UOB, 2015), (Forsman & Temel, 2011), (Fuetsch & Suess-Reves, 2017)]. Therefore its contribution to a firm's survival and growth is without doubt.

As a general definition, innovation essentially refers to the inherent ability of firms to withstand competition through the identification of novel and better ways to acquire and execute tasks related with products, processes, systems, structures, services, and marketing methods (Calantone, Cavusgil, & Zhao, 2002).

However, while the creation of novel ideas, products, and services was oftentimes viewed in the past as an exclusive result of internal activities such as in-house R&D, there is an increasing volume of studies that recognizes the critical contribution of environmental/ external elements as well (Wynarczyk, Piperopoulos, & Mcadam, 2013). Chesbrough, who is considered

the pioneer of this ecosystem-based slant, names this model "open innovation" (Chesbrough, 2003). The open approach, unlike its closed counterpart, leverages on the continuous "inflows and outflows of knowledge" across firms and stakeholders to accelerate creation of newness (Chesbrough & Crowther, 2006). Collaborative idea generation, sharing of intellectual property, and continuous crosssector interaction are the basic principles of the open approach. Their combination in turn leads to an innovation that is open to various individuals and enterprises in the system [ (Wynarczyk, Piperopoulos, & Mcadam, 2013), (Pervan, Al-Ansaari, & Xu, 2015)]. The open approach has undoubtedly been well received by the vast majority of the 21st century companies due to its track record for a consistent raising of the bar of innovation practices across industries and firms, and the academic community has spent significant efforts to document its mechanisms across corporates and industries. However, in spite of its obvious relevance, while open innovation practices have been widely investigated in large corporates, its familyowned counterpart has not always received the scholarly attention it deserves. To date, the literature has been unable to provide conclusive findings to fundamental research questions related to drivers of open innovation in family owned enterprises in consideration of both incompany drivers and environmental determinants (Hossain & Kauranen, 2016). Scholars have in fact focused mostly on the business dynamics piece of innovation, and often overlooked the family system (De Massis, Di Minin, & Frattini, 2015). Our current study is an attempt to fill this gap, and by focusing on cases of Singapore based firms, it aims to make family three contributions. Firstly, to help family-owned firms navigate the complex landscape of innovation initiatives. Secondly, considering that past research had long sought to understand determinants of innovation, with limited work done in consideration of the joint effect of innerworkings of family-owned enterprises and external drivers, this research will bring a holistic approach and provide family firms and policy makers with key strategic insights to push the innovation agenda forward. Thirdly, this work will debate on the impact of tradition and past knowledge in innovation initiatives (de Massis, Frattini, Kotlar, Petruzzelli, & Wright, 2016).

The reasons to focus on family businesses, in Singapore, is twofold. Firstly, family business is the most common ownership business model in the world. They are the backbone of the global economy, with an estimated contribution of over 70% of global GDP, a provision of 50-80% of worldwide private sector jobs, and a financial capital currently supporting 85% of start-ups [

(Osunde, 2017), (European Family Businesses, 2012)]. Southeast Asia, oftentimes defined the engine of global development (United Nations, 2018), has even higher dependency on family firms. Regionally, 80-90% of large companies are family-owned, and there are "good reasons to assume that their role in their respective national economies will remain crucial" (Björnberg, Elstrodt, & Pandit, 2014). Singapore is strategically positioned at the heart of Asia, and besides its well-cemented position as a Southeast-Asian business hub (EDB Singapore, 2017), the city-state is regarded as one of the pioneers of innovation practices in the region. Tracing the history of the nation, it is evident that while in the past, technology and innovation had served in a functional role, today they are "the central engines powering an ambitious economy" (Ng, Lim, & Wong, 2018). Given that Singapore's most valuable resource is its people, the national government has recognized early that a leading role in the field of innovation, science and technology would set Singapore apart on its journey to become the "Global Asia node" [ (Ng, Lim, & Wong, 2018), (Lung, 2018)]. Therefore, developing a deep understanding of Singapore family-owned enterprises' innovation behaviour is fast becoming a crucial long-term priority, not only for those firms who are currently active in the Lion City, but also for investors and policy makers that must ultimately decide how to better foster the development of the region as a whole.

This paper therefore hopes to provide strategic insights to practitioners, family businesses, and policy makers by addressing the following questions:

- How do Singapore family-owned businesses interpret innovation, and what is their underlying innovation culture?

- What are the main environmental and innerworking drivers of innovation in Singapore based family-owned enterprises ?

The rest of the paper is structured as follows. Section 2 reviews literature on innovation, innovation models, and drivers of innovation in family-owned entreprises. This is followed by a detailed analysis of research gaps that the present work seeks to fill. Section 3 discusses the Singapore family business context along with a comprehensive analysis of the innovation landscape in the city-state. Section 4 presents the research framework, along with a detailed analysis of research contribution. Section 5 discusses the methodological approach, input data, and survey structure. Section 6 presents analysis of the results. Lastly, section 7 discusses managerial implications, research limitations and directions for future research.

### **Research Background**

Business innovation management, especially in corporates, has been widely investigated and many research papers are available in this domain. Some research looked at drivers (or determinants) of innovation. some other attempted to define innovation models, whereas another piece conducted in-country analysis to map the status of innovation practices around the globe. This section reviews previous scholarly in these areas while introducing work fundamental concepts such as definitions of innovation, it also provides insights from previous research on drivers of innovative projects, and innovation models.

Defining Innovation. Despite the broad use of the term innovation, there is often no real agreement on what this term means. Some authors look at innovation from a process perspective, others may define the term from the standpoint of methods, ideas or products. However, the common understanding across industry, academic and practice is that innovation encompasses a series of initiatives geared toward the provision of added value to customers, the scaling up of value delivery to a larger set of customers or some combinations of the two (Galper, 2016). Literature typically categorizes innovation definitions into two main groups namely customer experience innovation and company transformation innovation. Customer experience innovation focuses on how customers directly experience companies' products and services or their perception of the brand. whereas company transformation innovation impacts the inner-workings of firms and typically refers either to processes or people (Galper, 2016).

According to Oslo Manual, innovation can materialize in a variety of forms namely product, process, marketing, and organisational (OECD & Eurostat, 2005). Product innovation refers to introducing a new product or service to the it represents market or а significant improvement of an existing one. Improvements might include technical specifications, software, raw materials or components and materials, user friendliness or other functional characteristics (OECD & Eurostat, 2005). Typically, product innovation has a direct impact on top-line growth. Process innovation refers instead to the adoption of an improved production or delivery method, which might include changes in production techniques, equipment and/or software (OECD & Eurostat, 2005). Process innovation is generally cost-cutting in nature (e.g. new manufacturing process to reduce unit cost) and is typically reflected in bottom-line Marketing innovation is about the growth.

implementation of new marketing methods and usually consists of changes of any of the four P's of marketing: product (e.g. design or packaging), promotion, price and/or placement (OECD & Eurostat, 2005). Lastly, organizational innovation is about new organizational methods in business practices, workplace organization or external relations (OECD & Eurostat, 2005).

Innovation Determinants. Family-owned enterprises are intrinsically different from their non-family owned counterparts (Gallo, Tàpies, & Cappuyns, 2004), and these dissimilarities are also reflected in the way innovation practices are driven at firms' levels. In fact, unlike nonfamily owned enterprises, family businesses have the additional family component impacting the way the business is structured and organized (Kellermanns, Eddleston, Barnett, & Pearson, 2008). Literature in innovation management in family businesses divides innovation determinants into two main categories namely external and internal drivers.

External determinants are defined as ecosystemdriven elements that can enable, or limit, the ability of firms to innovate (Avlonitis & Gounaris, 1999). Major external catalysts include government support [ (Hadjimanolis, 1999)] partnerships with academia, public agencies, and other private firms (Perkmann & Walsh, 2007), access to external funds (Zhu, Wittmann, & Peng, 2011), and market dynamics (Huizingh, 2011).

Internal determinants are inner-working enablers of innovation. Internal drivers of innovation comprised family and business culture (Hernández-Perlinesa & Mancebo-Lozano, 2016), which amongst others include factors like family involvement (Lopez-Fernandez, Serrano-Bedia, & Gómez-López, 2015), inter-generational involvement (Sharma, Chrisman, & & Chua, 1997), CEO/ C-suite thinking (Zahra, 2005) and business network (Öberg, 2018).

Innovation Models. The concepts of open and closed innovation have attracted interest from academic and practitioners' communities alike. Of the two approaches, the so-called open model has received greater attention, especially in most recent times (Huizingh, 2011). In fact, although the two paradigms at first glance seem to point in opposite directions, oftentimes open innovation is considered the natural development of its closed counterpart. Scholars debate that this evolution from a closed into an open model was made necessary by the combined effect of environmental factors such as change of work patterns (e.g. higher workforce mobility), rise of technologies enabling collaborations across industries and geographies and increased interest of all stakeholders (e.g. suppliers) to contribute in a new way to the creation of innovation [ (Chesbrough, 2003), (Dahnlander & Gann, 2010)].

Additionally, in the current business context where profit margins are increasingly affected by stiff competition and uncertainty, it is critical for companies to maximize their returns on innovation investments. Firms emphasizing inside-out open innovation have shown great likelihood to create radical innovations and sell a higher number of new products (Inauen & Shenker-Wicki, 2012). Hence a growing pool of firms are shifting towards an ecosystem-based approach to boost internal capabilities to improve processes, systems, and products collaborative through novel solutions Γ, (Wynarczyk, Piperopoulos, & Mcadam, 2013), (Furr & Shipilov, 2018), (EDB Singapore, 2017)]. **Research Gaps** 

Despite the large amount of research papers available in the domain of business innovation management, it appears that the body of literature is currently unable to provide conclusive findings to fundamental questions about the role of key innovation drivers and their underlying structure, especially within the niche of family-owned enterprises. In addition, limited knowledge has been developed in regard to innovation-enabling determinants in a familybusiness dominated economy such as Singapore. In fact, from the analysis of previous scholarly work, it appears that the vast majority of the research have focused on Western societies. where findings are not directly applicable to the Asian context. There is a significant void in literature about open innovation in Asia, whereby cultural values are heavily embedded within the way families conduct business. Unlike in Europe and the United States, Asian familyowned businesses are often young organizations where the family has been in business for not more than 50 years [ (The Business Times, 2018) (Koh & Kong, 2016a)]. Additionally, while there is a fair number of studies which focused on either internal or external determinants of innovation, there is relatively little attention on their joint effects on firms' performance. Thus, this paper aims to fill these gaps, and by analyzing the perspectives of a diverse group of Singapore family enterprise, this work will map the status of innovation practices in the citystate of Southeast Asia.

### The Case Study of Singapore

### Why Singapore Family Enterprises

Located in South East Asia with a GDP of \$297.0 billion, Singapore is considered, together with Hong Kong, South Korea and Taiwan, as one of Asia's four economic tigers. Since its independence in 1965, the city-state has gone through an impressive journey of growth and the nation is considered one of the wealthiest countries in the world per capita (US News, 2018). According to data from the national department of statistics, in 2017 Singapore counted over 220.000 enterprises, of which 99% were SMEs and 85% were locally owned. Similarly, employment reached 3.4 Million people (excluding domestic foreign workers), of which 65% were absorbed by SMEs, and 68% by local firms. Real GDP growth in 2017 was measured at 3,4% (Singapore Department of Statistics, 2017).

Despite the strong appeal and highly favourable conditions for multinational corporations (MNCs) to establish themselves in the city-state, local family-owned enterprises still dominate the domestic economy. Looking at the Singapore Exchange for instance, family-owned firms made up over 60% of listed firms (The Straits Times, 2015), and their role as SMEs, which is by far the key engine of Singapore's economy, is even more central (Gov.sg, 2017). Family businesses contribute to nearly 70% of the national GDP and employ 50% of the workforce (Family Firm Institute, 2018). Additionally, although their presence spans across all sectors of Singapore's economy, their main role is registered in pivotal industries of the local economy such as hospitality, construction. real-estate. manufacturing, services and trading (The Straits Times, 2015).

The country has been consistently ranked as one of the most cosmopolitan cities globally, and its millennials, who often represent the next generation of business leaders, are global citizens while anchored to their own ethnic and cultural roots [ (Yeoh, 2004), (Timperio, Tan, Fratocchi, & Pace, 2016)].

### A Track Record of Open Innovation Excellence

Over its 50 years of history, Singapore has earned a reputation of being a global innovation hub "at the cutting-edge of modern business" (Basulto, 2015). With a proven track record of excellence, the city-state has always been riding the wave of innovation to set itself apart (Basulto, 2015). In the mid-1960s and 1970s the role that Singapore played was as a low-cost manufacturing hub. By late 1970s and early 1980s, the national economy quickly moved up the value chain of electronics and semiconductors industries to become (after the economic stagnation of 1997 Asia financial crisis) a leading knowledge-based economy today [ (Basulto, 2015), (Wong, 2008), (Tan & Phang, 2005)].

At the present time, Singapore business ecosystem collaboratively embraces and promotes the open innovation model [ (Shiao, 2018), (Straitstimes.com, 2018)]. According to a recent research by EDB Singapore published in Harward Business Review (2017), Asia, and Singapore in particular, offers not onlv favourable conditions to do business, but also a wide array of soft-factors which in turn enable the region to be on the "leading edge of the innovation curve" (EDB Singapore, 2017). As a result, countless examples of home-grown family owned enterprises with strong innovation capabilities are available, and organizations like Goldbell Group, PBA (Precision Bearings and Automation) Singapore, Cycle & Electric company (Cyclect), Chef-in-Box by JR Group, Hai Sia Seafood, Q Industies, are few of the most successful innovation cases [ (Koh & Kong, 2018a), (Koh & Kong, 2016b)]. Bloomberg 2017 global Index on the state of innovation ranks Singapore in third position, ahead of Germany, Switzerland and Finland, and first among other Southeast Asian nations (Jamrisko & Lu, 2018).World Intellectual Property Organization positions the Lion City as the most innovative economy for South East Asia, East Asia, and Oceania, and fifth on a global scale (WIPO, 2018).

### Research Framework

In light of previous investigations, both descriptive and empirical in nature, and research gaps described in section two, this study argues that both external (ecosystem) and internal (inner-workings) factors jointly contribute to building family firms' innovation capabilities. This is still a debatable issue within the academic community, and to date the body of literature has been unable to provide conclusive and widely applicable findings. Thus, this study aims to capture the Southeast Asian perspective by mapping the state of innovation practices in the use case of Singapore.

### **External Determinants**

With regards to external determinants of innovation, literature mostly focuses on four distinct factors namely government support, access to external capital, partnerships, and market dynamics.

Government support. Government plays a critical role in building business ecosystems conducive to innovation. They establish policies (Wonglimpiyarat, 2011), design incentive schemes for innovation to thrive, build the legal regulatory framework and (Spithoven, Vanhaverbeke, & Roijakkers, 2013), and shape the workforce's skillsets through the implementation of educational and training programmes (Mani, 2011). According to a recent study by Pervan, et al. (2015) on environmental determinants in Dubai SMEs, government policies have a significant impact on innovation capabities. GreenDot Group for example, which is just one of the many success stories of based family-owned enterprises, Singapore enlisted the support of SPRING Singapore (today named "Enterprise Singapore") to incapsulate

design thinking concepts withing their branding strategy with the aim to gain a deeper understanding of their target group (Koh & Kong, 2018b).

Access to external resources. Availability and access to capital is also a key leverage to generate and implement new business ideas, ventures, or even business models. Singapore offers a wide range of financing possibilities including banks, financial institutions, peer-topeer crowdfunding, angel investors, and more (linkflow Capital, 2018). In this research, we are interested to understand whether the current schemes enable innovation in family businesses, and whether banks and financial institutions in Singapore are supportive in funding innovation ideas and projects.

Partnerships. Academic-industry collaborations as well as partnerships with other private firms or public agencies, are increasingly becoming a key element of efficient national innovation systems. For example, under the academicindustry collaboration scheme, universities become a major contributor of businesses' innovation strategies, especially from the angle of generating and disseminating new and relevant knowledge to private businesses [ (Pertuze, Calder, Greitzer, & Lucas, 2010), (Fischer, Vonortas, Schaeffer, & Queiroz, 2016)]. In this regard, Singapore is growing a vibrant national innovation system where academic research excellence plays an increasingly important role. In 2016, on its Research Innovation Enterprise (RIE) 2020 Plan, the national government has unveiled a S\$19 billion (nearly USD \$14 billion, equal to +18% from previous 2015 RIE) plan to support Singapore's R&D efforts with the aim to address national challenges but also to create value to spin off drivers of economic growth. [ (National Research Foundation, 2018), (Fai & Kek, 2016)].

Market dynamics. The nature of the market, the competition level, and product types, can create the conditions for firms' to foster their innovation capabilities (Reeves & Deimler, 2011). Rapid changes of market conditions create in fact new business opportunities, and tend to speed up innovation processes so as "to meet unmet needs" (Nemet, 2009). Singapore is a highly dynamic market and a catalyst of dynamism across the entire Southeast Asian region. Looking at 2017 alone, the city-state received US\$62b of foreign direct investment, standing in 5th position in the global top 20 foreign investment hubs (Singapore Business Review, 2018). In addition to the dynamism related to foreign investments, favourable regulatory framework, and favourable taxation schemes for foreign firms contribute to raise the bar of competitiveness in the domestic market.

**Internal Determinants** 

According to previous investigations, elements like family and business culture (family involvement, top management or C-Suite philosophy towards innovation and company culture) and extented business network are the internal factors that contribute to family firms' innovation capabilities.

Family and Business Culture. Family and business culture factors are related to innovation management in family enterprises such as family involvement, multi-generational involvement, innovative characteristics/CEO strategic thinking which are more than general organizational culture. Family involvement typically refers to the degree of involvement of family members in the management organization and structure of the family firm (Lopez-Fernandez, Serrano-Bedia, & Gómez-López, 2015). Most research published to date concur that a greater family involvement corresponds to a higher interest of the firm to pursue innovation initiatives. In fact, direct involvement of family members takes away the need for short-term/ quick wins to move the focus to longer-term goals, which are the typical timeframes of innovation initiatives (Mustakallio, Autio, & Zahra, 2002). In regards to multi-generational involvement, according to survey findings of Deloitte (2016), the majority of next-generation business leaders believe that innovation is key for business survival and growth. The 61% of respondents argued that the current generation are well aware of challenges in innovation but, according to 40% of respondents, current generation leaders were not willing to take the risks associated with it. Additionally, 51% of next-generation leaders shared their intention to be more risky in their decision-making, although in a more controlled way. In this regard, PBA Singapore is a good example of differences in cross-generational perceptions of innovation and inner strengths of this "sandbox approach". PBA Singapore's CEO when he first joined the Derrick Yap, organization, had to convince its father of his capabilities to transform the business by taking on the great challenge of transforming the company's Malaysian market business with the support of only a six-member team. Today, Malaysia contributes to 30% of PBA's group revenue streams (Koh & Kong, 2018a). The level of involvement of top management (e.g. CEO) or C-Suite also appears to have a certain degree of impact on the firm's innovation capabilities (Yadav, Prabhu, & Chandy, 2007). In this regard, Dyer and Gregersen (2012) wrote that the "behaviour of leaders matter-big time", in the innovation processes. Additionally, risk-taking behaviour of management, encouragement of creativity, and participation of all staff in the innovation process, to name but a few, are also

critical elements of innovation orientation [ (Maher, 2014), (Szczepańska-Woszczyna, 2014)]. Network. A business network consists of a series of companies that are directly and indirectly connected through social and/or economic ties. Innovation is connected to business networks in several ways. This connection might exist due to the fact that creation of newness may be the result of interaction between business partners (Öberg, 2018), but also due to the change that innovation will bring in the interaction patterns among business partners, [ (Ostendorf, Mouzas, & Chakrabarti, 2014)].

### Contribution of the Current Research

This work will shed light on key catalysts of open innovation in family-owned firms. Particularly, by leveraging on a sample of family firms in Singapore, a highly dynamic, yet culturally Asian society in Southeast-Asia, this research will bring the following contributions:

- Identify the main drivers of innovation across family firms examining both internal and external determinants. This will enable academics and practitioners to identify the most critical factors fostering innovation in Singapore based family firms,

- Illustrate the unique elements of Singapore family-owned business culture with regard to innovation management practices.

- Understand the impact of tradition and past knowledge in the innovation activities to specifically understand whether firms with strong ties to traditional values are able to roll successful innovation initiatives, and if so to what extent.

- Bridge theory with practice, and provide family businesses with a set of key determinants, as well as practical insights for policy makers to build a more inclusive and thriving innovation ecosystem.

### Methodology and Sample Characteristics

### Survey Structure & Measurement of Constructs In order to test the meet the research

objectives, a guestionnaire was developed and a survey administered over the period October -December 2018. The items used to measure the constructs were finalized based on an extensive literature review on business innovation combination with management in group discussions with senior executives of Singapore family-owned businesses. The combination of literature with focus group discussions was needed to ensure that wording and sequencing of questions were appropriate before administering the actual questionnaire. As a result, a questionnaire of 71 items was conceived. Out of these 71 survey items, 33 were about demographics of respondents and characteristics

of industries they operate in, whereas the remaining 38 were in the form of a seven-point Likert scale anchored on environmental and inner-workings determinants of open innovation. Environmental Determinants. Questions designed assess government-supported initiatives to require opinions on the policies designed by the national government to support the development of firms' innovation capacities, and particularly about availability and accessibility to government financial assistance schemes, but also about orientation of innovation initiatives towards creation of intellectual property (IP). Questions on accessibility to financial resources required responses on non-government related financial schemes including both external funding (e.g. angel investors, crowd funding, and bank loans) and internal funding (family capital). Partnerships were measured through questions regarding collaborations with education institutions (universities, polytechnic) and public sector agencies, as well as eagerness to repeat such experiences in future. Finally, market dynamics were assessed using questions on industries' competitiveness level, as well as industries' eagerness to embrace mega trends such as automation, 3D printing, artificial intelligence and machine learning.

Internal Determinants Questions related to family and business culture required responses on family's approach and philosophy towards innovation such as risk-taking behaviour, family involvement, knowledge of the industry, top involvement management in innovation initiatives, passion for newness, and level of next-generation innovation from business leaders. Business network was measured using ad-hoc questions on type of network fostering innovation, and access to friends and acquaintances' finances to fund innovation.

After the data collection phase, a series of multivariate statistical analysis using SPSS v25 software package were undertaken in order to explore the structure of innovation enabling determinants. Details of analysis and results are included in Section 6.

### Sample

Firms to be contacted are family-owned organizations sited in Singapore. Considering that literature has been quite ambiguous in defining enterprises that fall under the umbrella of family businesses (Voordeckers, Van Gils, & Van den Heuvel, 2007), we defined our sample to include organizations that are controlled by one family group through "a clear majority of ordinary voting shares", and have family representation at management level (Lopez-Fernandez, Serrano-Bedia, & Gómez-López, 2015). In order to identify the group of firms to include in the study, the database of the "Business Families

92

Institute. Singapore Management University" (BFI@SMU) was used. In total, 100 family enterprises operating in key industries of Singapore economy such as construction, hospitality, real-estate, manufacturing, services and trading were approached to take part in the research. These organizations constitute the population targeted by this research. The questionnaire was sent via e-mail, with follow up clarifications via phone calls. To motivate a timely and complete response, the respondents were informed that they will receive a summary of research findings when the report is launched. Overall, a total of 61 unique subjects participated in the survey, of which 28 questionnaires were discarded due to incomplete responses. The remaining 33 responses which were eventually considered valid were provided by family enterprises with overseas presence in 15 countries across ASEAN (mostly Singapore, Malaysia, China, Indonesia, and Vietnam) and beyond (Hong Kong, India, Brazil etc.), and business profiles of respondents are summarized in Table 1.

 Table 1 Business Profile of Respondents.

Category	%
Logistics & Supply Chain (Procurement, Transportation, Storage)	30%
Manufacturing Wholesale and retail trade	17% 13%
Food & Beverage related services	<b>7</b> %
Financial Services and/or insurance activities	7% 6%
Professional, scientific and technical activities	<b>6</b> %
and other)	15%
Size of firm	
Below 50 employees	20%
Above 200 employees	30% 50%
Annual Turnover	30/0
Above S\$ 1 and below 5 Mil	18%
Above SS 5 and below 20 Mil	12%
Above S\$ 50 and below 100 Mil	14%
S\$100 Mil and above	35%
Gender of Individual	000/
Female	80% 20%
Group Age of Individual	
18-29	21%
30-39 40-49	52% 15%
50-59	6%
60-69	<b>6</b> %
Generation of Individual	<b>7</b> 0/
Second	58%
Third	27%
Fourth and above	12%

### Analysis & Results

This study serves to examine the correlation between environmental and internal innovation drivers on family-owned firms' innovation capabilities. That is, this research aims to deepen the understanding on innovation determinants taking into consideration the joint effect of both external and inner-working factors. In this section of the paper, we provide insights on the techniques that were selected to meet the objectives of this study as well as research findings. Results are structured into two main areas namely demographic and multivariate statistical analysis.

Demographics. Following the survey, the primary data collected was consolidated. Prior to the testing of the research hypotheses, a descriptive analysis of questionnaire responses matched to demographic profiles of respondents and their industries was undertaken to summarize key characteristics of data collected. Key information is summarized in Tables 2, 3, and 4.

Table 2Respondents' Sentiment about Innovationwithin their Industries.

Category	%
Competition Level	
Intense	<b>90</b> %
Oligopoly	10%
Pace of Innovation	
Fast	33%
Moderate	<b>9</b> %
Slow	58%
Key Driver of Competitiveness	
Heavy physical infrastructure	68%
Intellectual property	18%
Capital intensity	4%
Combination of above options	10%

The large majority of respondents (90%) is operating in industries characterized by intense competition levels. However, surprisingly, respondents characterize the pace of innovation as "slow", which is likely related to the type of businesses which our sampled firms were involved in. The interviewed firms operated mostly in traditional businesses, where physical infrastructure were the main assets rather than intangible assets and resources such as intellectual property.

 Table 3 Respondents' sentiment about the importance of various Innovation forms.

Innovation types Product Process Marketing Organizational	Utmost Importance 25% 43% 7% 25%	Of secondary Importance 14% 39% 18% 20%	Of tertiary Importance 32% 18% 32%	Of Least Importance 29% 0% 43% 20%
Organizational	25%	29%	18%	29%

Most responders (43%) believed that process innovation was the most critical form among all types of innovation, followed by organizational

innovation that had been ranked as top priority by 25% of our respondents. Interestingly, product innovation does not represent the primary form for our sampled firms. In fact, only 14% of respondents ranked this as the main priority.

Table 4Sales due to innovation, market share,and growth of innovation espenses.

Percentage	Revenue percentage growth due to innovation (period 2013- 2018)	Innovation expenses percentage increase (period 2013- 2018)
Less than 5%	40%	43%
Between 5% and 10%	20%	17%
Between 10% and 30%	17%	27%
Between 30% and 50%	17%	7%
Between 50% and 70%	7%	3%
Between 70% and 100%		-
More than 100%		3%

The large majority of the sample did not leverage on innovation to drive their revenue streams in the period of consideration. 40% of the sample had in fact experienced less than 5% revenue growth due to innovation, which partially explained also the limited investment on innovation projects.

Multivariate Statistical Analysis. In order to meet the objectives of this study an exploratory principal component analysis (PCA) along with a measurement of internal consistency (reliability) of constructs via Cronbach's alpha and convergence via composite reliability and Average Variance Extracted (AVE) was conducted.

### Factors Structure

PCA is a variable-reduction mathematical procedure that allows us to reduce a larger set of correlated variables into a smaller set of uncorrelated "artificial" constructs namely "principal components". Principal components are the underlying factors accounting for most of the variance of original variables [ (LAERD , 2018), (Gunday, Ulusoy, Kilic, & Alpkan, 2011)]. As mentioned, in this study PCA was used with the twofold objective of deepening the understanding of intrinsic connections across various constructs (innovation and drivers of innovation. In particular, a PCA with varimax rotation was performed on innovation drivers and, as suggested by Kim & Mueller (1978) only factors with eigenvalues greater than 1 were retained for further analysis (results in Table 5 and 6). As a result, nine factors are extracted of which eight were related with drivers of represented firms' innovation, and one

innovation capabilities.. These factors were labelled based on the items included in each. The total variance explained on innovation drivers was 84%. Cronbach's alpha test and reliability composite were above the recommended 0.600/0.700 principle (except for composite reliability for factor 8, which was slightly below the threshold of 0.6), hence it can be stated that the survey items performed well in capturing their respective latent variables (Hair Jr., F., Black, Babin, & Anderson, 2014). Average Variance Extracted (AVE) was also calculated for convergence validity so as to assess the extent to which the variance of latent variables were explained by the survey items. Considering that all AVEs values are greater than 0.50 threshold, it can be concluded that more than the 50% of the variance of constructs can be explained by their respective items (Vinzi, Chin, Henseler, & Wang, 2010).

Findings from PCA highlighted that all innovation drivers documented in existing literature are relevant for the Singapore context, in addition, two further catalysts namely family capital and (personal) network also play a role in the fostering of innovation initiatives. Thus, results of PCA bring a few considerations. First, private family wealth plays a significant role in fostering innovation initiatives in Singapore. Second, the distinctive elements of family and business culture of Singapore family-owned enterprises include family involvement, know-how of internal processes and products, passion for innovation, innovation spirit, and risk-taking attitude (which are correlated via Cronbach alpha of 0.902). Third, personal network and professional network are independent elements and both are drivers of innovation initiatives. Results of PCA are valid considering that Reliability, Convergence, AVE, and discriminant validity are within acceptable ranges.

After reducing the initial set of variables into a more manageable subset of factors, a correlation analysis was performed to understand correlation among components and results are in Table 6.

While not all correlations were statistically significant, some interesting associations also emerged, suggesting the possible existence of indirect effect (mediations) of innovation drivers on innovation capabilities. Table 5 Cronbach's alpha, Factor Loading, Eigenvalues, Variance explained, AVE, loadings, and composite reliability (Drivers of Innovation).

No	Items	Factor Load	Eigenvalue	Cum. % Variance Explained	Cronbach α	AVE	Composite Reliability
1	Family and business Culture		6.996	19%	0.902	0.608	0.902
	Family involvement drives innovation.	0.736					
	An extensive know-how of internal processes, product, and organization foster innovation.	0.576					
	Passion for newness is a fundamental driver of innovation Innovation spirit foster innovation and progress	0.831 0.793					
	Top management involvement foster innovation	0.809					
	The willingness to take risks impacts outcomes of innovation	0.896					
2	Access to external Funds		3.286	32%	0.883	0.566	0.900
	It is easy relatively easy to access to angel investors to fund innovation initiatives	0.949					
	It is relatively easy to access to crowd funding to fund	0.951					
	It is relatively easy to obtain bank loan to fund innovation initiatives	0.677					
3	Government Supported Initiatives		2,165	43%	0.74	0.501	0.830
	The national government provides with a series of financial assistance schemes such as grants, loan & insurances, tax	0.580					
	innovation in family firms. It is easy to identify the most appropriate government						
	financial assistance scheme(s) that fits with my needs, and I believe it is easy to apply for such schemes.	0.900					
	Advanced technology is a key element to withstand	0.650					
	People are the main source of innovation	0.670					
	Innovation initiatives shall be oriented towards generation of	0.700					
	Intellectual properties	0.700		500/	0 ( 05		0 705
4	Market Dynamics	0.942	1./61	52%	0.685	0.64/	0.785
	Mega trends such as automation, 3D printing, Artificial	0.043					
	businesses and there is the need to leverage on those to withstand the acute competition	0.764					
5	Partnerships		1.574	60%	0.709	0.713	0.832
	Partnerships with educational institutes (universities, high schools) foster innovation	0.781					
	Partnerships with public sector organizations foster innovation	0.903					
6	Network		1.31	68%	0.618	0.585	0.733
	Do you use friends' capitals to fund innovation?	0.881					
	Partnerships with private sector firms foster innovation	0.627					
7	Family capital		1.108	76%	-	0.775	0.775
	Family capital are the main source of funds for innovation initiatives	0.88					
8	External Network		1.053	84%	0.688	0.5	0.662
	Singapore's legal and regulatory framework nurtures						
	innovation and promotes the development of a conducive business environment whereby innovation spirit can truly thrive	0.801					
	A solid network fosters innovation	0.600					

### Table 6 Correlation Analysis.

	μ	SD	Family and business Culture	Access to external Funds	Government Supported Initiatives	Market Dynamics	Partnership	Network	Family capital	External Network
Family and	5.2	1.7	1	-0.026	.329**	0.157 <sup>1</sup>	-0.092	.215	-0.132	.328"
business Culture										
Access to	2.1	1.8		1	0.145 <sup>1</sup>	-0.026	0.096	275	0.000	267
external Funds				•	01110	0.020	0.070		01000	
Government										
Supported	4.4	1.7			1	-0.021	-0.009	.240	0.056	.351"
Initiatives										
Market Dynamics	4.5	1.8				1	0.135 <sup>1</sup>	-0.059	-0.189 <sup>1</sup>	.334"
Partnership	2.2	1.9					1	.216	-0.054	-0.031
Network	2.5	2.0						1	0.289 <sup>1</sup>	.231
Family capital	4.8	1.8							1	308
External Network	4.4	1.6								1

\*\* Correlation is significant at the 0.01 level
 \* Correlation is significant at the 0.05 level
 <sup>1</sup> Correlation is significant at the 0.1 level

### Discussion

This study analysed catalysts of open innovation in Singapore based family-owned enterprises in consideration of both environmental and innerworking determinants. The first finding of this research is a confirmation of the general strong passion of family-owned enterprises to undertake innovation initiatives. Srinivasa Raghavan Nadathur, founder of Nadathur Estates, in a recent conversation about his firms' approach on innovation mentioned that "The passion for innovation, to experiment, to create something that nobody has attempted before is intrinsic in Nadathur DNA" (Business Families Institute, 2018). The same passion has been detected groups throughout focus discussions and comments in the survey.

Regarding the empirical findings, drawing on a sample of 33 relevant family firms' responses, various insights can be offered to both scholars and practitioners.

First, while previous literature highlight six major factors among determinants of open innovation, our current research detected eight determinants. Particularly, two additional innovation determinants namely family capital and external network were detected via PCA. The remaining six catalysts of innovation projects (family and business culture, access to external funds, government supported dynamics, partnership, initiatives, market network, family capital, and external network) were found to play an important role in the Singapore family business ecosystem, confirming existing literature in this regard.

Second, the vast majority of our sampled firms mentioned process innovation as the most type, followed by organizational critical innovation. Process innovation is considered superior by our sampled firms due to its capabilities to drive product innovation, marketing and organisation structure (and people). Similarly, organizational innovation is ranked of utmost importance due to the increasing need to adopt technologies such as digitalization, robotics and automation, which demand that a proper organizational structure be in place. One of the respondents, a 3rd generation family business leader of a Singapore SME in his responses to this survey questionnaire highlighted that "The right processes create the necessary conditions to shape products, as well as marketing and organisation structures". On the same lines, a 2nd generation family business leader of another Singapore based family-owned firm reasoned out that "It is important that we have an efficient process in our business to minimize costs and increase customer satisfaction, especially in the current business context of stiff competition and uncertainty.

Having in place cutting-edge processes is a key differentiator".

Third, access to external funds is guite important to family businesses of Singapore, and findings of this study highlight that for Singapore family enterprises, external funds can be of diverse nature such as from angel investors, or crowd funding, but banks loans and they appeared to be still the preferred funding mechanism for innovation initiatives. Policy makers should take these factors into consideration when strengthening policy intervention for innovation. For instance the government could focus on catalysing innovation-thinking and culture through a process of continuous policy support and strategy development e.g. via facilitation of funds acquisition, training, incubator and acceleration activities.

### Managerial Implications

This study provides strategic insights for family business firms to navigate the complex landscape of innovation initiatives, and also key takeaways for policy makers to strengthen their policy intervention for innovation. Particularly, insights of this study show that while environmental determinants have an important role in the development of innovation capabilities internal factors play a fundamental role too. Therefore, while injection of external capital and policies are still necessary, it appears that those elements alone are but not sufficient. Therefore, while the external ecosystem plays a crucial role for the development of innovation capabilities, it must be coupled with inner-workings determinants of the family firms too. Therefore, for family business to successfully innovate, both dimensions must be concurrently taken into account and nurtured.

### Limitations

This work has a number of limitations. The first is the limited generalizability of the results to other Southeast Asian countries. Despite the fact that Singapore is one of the most cosmopolitan cities in the world characterized by familyowned enterprises with a wide exposure to other neighbouring countries' culture, Singapore's business ecosystem is unique and thus the findings from this study cannot be generalized to the context of other Southeast Asian countries. The second limitation is the small sample size. A wider pool of respondents would reinforce current findings and provide deeper insights on the status of innovation practices in the citystate.

### Future work

Findings of this study opened several avenues for future research. First, one can apply structural equation modelling (SEM) to uncover mediations among variables, weights, loadings, paths, as well as provide an estimation of latent variables scores. As mentioned earlier in this paper, outcomes of PCA and correlation analysis suggest possible existence of indirect effect the (mediations) of innovation drivers on innovation capabilities. Thus a future study might want to fill this gap through a larger sample of data. Second, for future studies one can expand the geographic scope from one single nation to the entire ASEAN region. This would allow crosscomparison of country's business ecosystem, and status of innovation practices across the Southeast Asian region. This is particularly relevant in light of developments in the ASEAN Economic Community (AEC) arising from crossborder trade and thus open innovation practices. future research could comparatively Third, assess the perception of innovation practices between current and next generations. This would allow for the design of appropriate strategies to foster innovation initiatives across multi generations to build lasting innovation capabilities within family firms.

### References

- Ahmed, S., Halim, H. A., & Ahmad, N. H. (2017). Open and Closed Innovation and Enhanced Performance of SME Hospitals—A Conceptual Model. Business Perspectives and Research, 1-12.
- Almirall, E., & Casadesus-Masanell, R. (2010). Open versus Close Innovation: a Model of Discovery and Divergence. Academy of Management Review, 35(1), 27-47.
- An ecosystem to boost innovation and revolutionise business. (21 May, 2018). Retrieved from The Business Times: https://www.businesstimes.com.sg/companiesmarkets/an-ecosystem-to-boost-innovation-andrevolutionise-business
- Artz, K. W., Norman, P. M., Hatfield, D. E., & Cardinal, L. B. (2010). A Longitudinal Study of the Impact of R&D, Patents, and Product Innovation on Firm Performance. The Journal of Product Innovation Management, 27(5), 725-740.
- Atalay, M., Anafarta, N., & Sarva, F. (2013). The Relationship between Innovation and Firm Performance: An Empirical Evidence from Turkish Automotive Supplier Industry. Procedia - Social and Behavioral Sciences, 75, 226-235.
- Atuahene-Gima, K. (1996). Market orientation and innovation. Journal of Business Research, 35(2), 93-103.
- Avlonitis, G. J., & Gounaris, S. P. (1999). Marketing Orientation and Its Determinants: An Empirical Analysis. European Journal of Marketing, 33, 1003-1037.
- Bae, Y., & Chang, H. (2012). Efficiency and effectiveness between open and closed innovation: empirical evidence in South Korean manufacturers.

Technology Analysis & Strategic Management, 24(10), 967-980.

- Baer, M., & Frese, M. (2003). Innovation is not enough: climates for initiative and psychological safety, process innovations, and firm performance. Journal of Organizational Behavior, 24(1), 45-68.
- Baron, R. M., & Kenny, D. (1986). The Moderator-Mediator Variable Distinction in Social Psychological Research: Conceptual, Strategic, and Statistical Considerations. Journal of Personality and Social Psychology, 51(6), 1173-1182.
- Basulto, D. (26 May, 2015). The secrets to Singapore's track record of innovation excellence. Retrieved 31 December, 2018, from The Washington Post: https://www.washingtonpost.com/news/innovatio ns/wp/2015/05/26/the-secrets-to-singaporestrack-record-of-innovationexcellence/?noredirect=on&utm\_term=.009d1a052 c66
- BFI@SMU; UOB. (2015). Riding on Asia's Economic Tranformation - Growth Strategies of Asian Business Families. Retrieved 9 October, 2018, from uobgroup.com:

https://www.uobgroup.com/assets/pdfs/newrelease/UOB-SMU-report.pdf

- Björnberg, Å., Elstrodt, H. P., & Pandit, V. (2014). McKinsey Quarterly. Retrieved 27 December, 2018, from The family-business factor in emerging markets: https://www.mckinsey.com/featuredinsights/winning-in-emerging-markets/the-familybusiness-factor-in-emerging-markets
- Business Families Institute. (2018). Beyond Dollars and Cents. In B. F. Institute, The 4Gs of Business Families (pp. 134-137). Singapore: Business Families Institute.
- Calantone, R., Cavusgil, S., & Zhao, Y. (2002). Learning orientation, firm innovation capability, and firm performance. Industrial Marketing Management, 31(6), 515-524.
- Carney, M., & Gedajlovic, E. (2003). Strategic Innovation and the Administrative Heritage of East Asian Family Business Groups. Asia Pacific Journal of Management, 20, 5-26.
- Chesbrough, H. (2003). Open Innovation: The New Imperative for Creating and Profiting from Technology. Boston: Harward Business School Press.
- Chesbrough, H., & Crowther, A. K. (2006). Beyond high tech: early adopters of open innovation in other industries. R&D Management, 36(3), 229-236.
- Chin, W. (1998). The partial least squares approach to structural equation modeling. In G. Marcoulides, Modern methods for business research (pp. 295-336). Mahwah, NJ: Lawrence Erlbaum Associates.
- Choi, T., & Eboch, K. (1998). The TQM Paradox: Relations among TQM practices, plant performance, and customer satisfaction. Journal of Operations Management, 17(1), 59-75.
- Chua, J. H., Chrisman, J., & Sharma, P. (1999). Defining the Family Business by Behavior. Entrepreneurship: Theory and Practice, 23(4), 19-39.
- Colarelli, G. O. (2005). Open, Radical Innovation: Toward an Integrated Model in Large Established Firms Introduction: The Problem of Radical Innovation in Large Established Firms. Oxford: Henry Chesbrough, Wim Vanhaverbeke and Joel West, eds.. Open Innovation: Researching a New

Koh, A. Kong, E. Timperio, G. (2019). An Analysis of Open Innovation Determinants: the Case Study of Singapore based Family owned Enterprises. *European Journal of Family Business*, 9(2), 85-101.

Paradigm2005. Retrieved 10 October, 2018, from https://www.researchgate.net/publication/267421 423\_Open\_Radical\_Innovation\_Toward\_an\_Integrat ed\_Model\_in\_Large\_Established\_Firms\_Introduction \_The\_Problem\_of\_Radical\_Innovation\_in\_Large\_Est ablished\_Firms

- Colombo, M., Piva, E., & Rossi-Lamastra, C. (2013). Open Innovation and Within-Industry Diversification in Small and Medium Enterprises: The Case of Open Source Software Firms. Research Policy, 43(5), 891-902.
- Dahnlander, L., & Gann, D. (2010). How open is innovation. Research Policy, 39, 699-709.
- de Jong, M., Marston, N., & Roth, E. (April, 2015). The eight essentials of innovation. McKinsey Quarterly. Retrieved from https://www.mckinsey.com/businessfunctions/strategy-and-corporate-finance/ourinsights/the-eight-essentials-of-innovation
- *De Massis, A., Di Minin, A., & Frattini, F. (2015).* Family-Driven Innovation: Resolving the Paradox in Family Firms. California Management Review, 58(1), 5-19.
- de Massis, A., Frattini, F., Kotlar, J., Petruzzelli, A., & Wright, M. (2016). Innovation through Tradition: Lessons from Innovative Family Businesses and. Academy of Management Perspectives, 30(1), 93-116.
- De Massis, A., Sharma, P., Chua, J., & Chrisman, J. J. (2012). Family Business Studies: An Annotated Bibliography. Cheltenham (UK) - Northampton (MA, USA): Edward Elgar.
- Della Torre, E., & Solari, L. (2008). Organizational Innovations and Firm Performance.Evidences From the Case of Medium-Sized Milanese Firms. XXIII Italian Conference of Labour Economics (AIEL). Brescia. Retrieved from https://www.aiel.it/page/old\_paper/dellatorre\_so lari.pdf
- Deloitte. (2016). Next-generation family businesses -Evolution keeping family values alive. Diegem: Deloitte University EMEA CVBA. Retrieved from https://www2.deloitte.com/content/dam/Deloitte /global/Documents/Strategy/gx-family-businessnextgen-survey.pdf
- deOliveira, L., Echeveste, M. S., Cortimiglia, M. N., & Gonçalves, C. (2017). Analysis of determinants for Open Innovation implementation in Regional Innovation Systems. RAI Revista de Administração e Inovação, 14(2), 119-129.
- Duran, P., Kammerlander, N., van Essen, M., & Zellweger, T. (2015). Doing More with Less: Innovation Input and Output in Family Firms. Academy of Management Journal, 59(4), 1224-1264.
- Dyer, J., & Gregersen, H. (5 September, 2012). How Innovative Leaders Maintain Their Edge. Retrieved 7 December, 2018, from Forbes: https://www.forbes.com/sites/innovatorsdna/201 2/09/05/how-innovative-leaders-maintain-theiredge/#52c755495fc1
- Economist. (24 September, 2009). Economist. Retrieved 21 December, 2018, from Dynasty and durability: https://www.economist.com/business/2009/09/24 /dynasty-and-durability

- EDB Singapore. (19 October, 2017). Asia: the Face of Innovation. Harward Business Review Analytical Services. Retrieved from https://www.edb.gov.sg/content/dam/frs/2017articles/asia-the-face-of-innovation/hbr-reportasia-the-face-of-innovation-white-paper.pdf
- edb.gov.sg. (14 June, 2016). Industry academia collaboration: a pipeline for talent. Retrieved 17 October, 2918, from https://www.edb.gov.sg/en/news-andresources/insights/talent/industry-academiacollaboration--a-pipeline-for-talent.html
- edb.gov.sg. (23 August , 2017). Singapore's long game in innovation. Retrieved 17 October, 2018, from https://www.edb.gov.sg/en/news-andresources/insights/innovation/singapores-longgame-in-innovation.html
- European Family Businesses. (June, 2012). Family Business Statistics. Retrieved 8 October, 2018, from European Family Businesses: http://www.europeanfamilybusinesses.eu/uploads /Modules/Publications/family-businessstatistics.pdf
- Explorable.com . (2018). Convenience Sampling. Retrieved 11 October , 2018, from https://explorable.com/convenience-sampling
- Fagerberg, J., & Mowery, D. (2006). Innovation: A Guide to the Literature . Oxford : The Oxford Handbook of Innovation.
- Fai, L., & Kek, X. (8 January, 2016). Govt commits S\$19b to new 5-year plan for R&D initiatives RIE2020. Retrieved 7 December, 2018, from Channel New Asia: https://www.channelnewsasia.com/news/business /govt-commits-s-19b-to-new-5-year-plan-for-ramp-d-initiatives-ri-8214052
- Family Firm Institute. (2018). Global Data Points. Retrieved 16 October , 2018, from https://www.ffi.org/page/globaldatapoints
- Fischer, B. B., Vonortas, N., Schaeffer, P., & Queiroz, R. (2016). Does University-Industry Collaboration Foster Academic Entrepreneurship? Evidence from a Developing Country. Anais do 1° Encontro da Nacional de Economia Industrial e Inovação [=Blucher Engineering Proceedings, v.3 n.4 (pp. 408-424). São Paulo: : Blucher. Retrieved from

http://pdf.blucher.com.br.s3.amazonaws.com/eng ineeringproceedings/1enei/024.pdf

- Flynn, B., Sakakibara, S., Schroeder, R., Bates, K., & Flynn, E. J. (1990). Empirical Research Methods in Operations Management. Journal of Operations Management, 9(2), 150-284.
- Forsman, H., & Temel, S. (2011). Innovation And Business Performance In Small Enterprises: An Enterprise-Level Analysis. International Journal of Innovation Management, 15(3), 641-665.
- Fuetsch, E., & Suess-Reyes, J. (2017). Research of innovation in family businesses: are we building an ivory tower? Journal of Family Business Management, 7(1), 44-92.
- Furr, N., & Shipilov, A. (2018). Building the Right Ecosystem for Innovation. MIT Sloan, 59(4).
- Gallo, M. A., Tàpies, J., & Cappuyns, K. (2004). Comparison of Family and Nonfamily Business: Financial Logic and Personal Preferences. Family Business Review, 17(4), 303-318.

Koh, A. Kong, E. Timperio, G. (2019). An Analysis of Open Innovation Determinants: the Case Study of Singapore based Family owned Enterprises. *European Journal of Family Business*, 9(2), 85-101.

A. Koh, E. Kong, G. Timperio

Galper, Y. (12 October, 2016). Innovation': Giving Structure & Meaning to an Ambiguous Term. Retrieved 2018 December, 2018, from EPAM: https://www.epam.com/insights/blogs/innovation -giving-structure-and-meaning-to-ambiguous-term

Gaskell, A. (1 March, 2018). The Innovative Power of Family Firms. Retrieved 21 December, 21, from forbes.com: https://www.forbes.com/sites/adigaskell/2018/03 /01/the-innovative-power-of-family-

firms/#31632cb37052 Gaskell, A. (1 March, 2018). The Innovative Power Of Family Firms. Retrieved 8 October, 2018, from Forbes:

https://www.forbes.com/sites/adigaskell/2018/03 /01/the-innovative-power-of-familyfirms/#3cabf2bd7052

- Gersick, K. F., Davis, J. A., Hampton, M. M., & Lansberg, I. (1997). Generation to Generation: Life Cycles of the Family Business . Owner Management Business Institute.
- Ghemawat, P.; Bastian, P. (29 March , 2017). Southeast Asia's globalization outperformers. Retrieved 8 October, 2018, from Nikkei Asian Review:

https://asia.nikkei.com/Economy/Southeast-Asias-globalization-outperformers

- Gov.sg. (20 February, 2017). SMEs are at the heart of our economy. Retrieved 16 October, 2018, from https://www.gov.sg/microsites/budget2017/pressroom/news/content/smes-are-at-the-heart-of-oureconomy
- Gu, W., & Surendra, G. (2004). The Effect of Organizational Innovation and Information Technology on Firm Performance. International Productivity Monitor, Centre for the Study of Living Standards, 9, 37-51.
- Guimon, J. (2013). Promoting university-industry collaboration in developing countries (Innovation Policy Platform,OECD and World Bank). Retrieved 7 December, 2018, from The Innovation Policy Platform:

http://innovationpolicyplatform.org/sites/default/ files/rdf\_imported\_documents/PromotingUniversit yIndustryCollaborationInDevelopingCountries.pdf

- Gunday, G., Ulusoy, U., Kilic, K., & Alpkan, L. (2011). Effects of Innovation Types on Firm Performance. International Journal of Production Economics, 133(2), 662-676.
- Hadjimanolis, A. (1999). Barriers to Innovation for SMEs in a Small Less Developed Country (Cyprus). Technovation, 19, 561-570.
- Hair Jr., J., F., Black, W. C., Babin, B. J., & Anderson, R. E. (2014). Multivariate Data Analysis (7th ed.). Edinburgh: Pearson. Retrieved from https://is.muni.cz/el/1423/podzim2017/PSY028/u m/\_Hair\_-

\_Multivariate\_data\_analysis\_7th\_revised.pdf

- Harper, D. (2000). Online Etymology Dictionary. Retrieved 1 October, 2018, from https://www.etymonline.com/word/innovation#et ymonline\_v\_30060
- Harrington, H., & Voehl, F. (2013). Innovation Management: A Breakthrough Approach to Organizational Excellence - Part 1. International Journal of Innovation Science, 5(4), 213-224.
- Hernández-Perlinesa, F., & Mancebo-Lozano, E. (2016). Conditional mediation of competitive strategy and environment in international

entrepreneurial orientation of family businesses. European Journal of Family Business, 6(2), 63-134.

- Hervas-Oliver, J., Sempere-Ripoll, F., & Boronat-Moll, C. (2014). Process innovation strategy in SMEs, organizational innovation and performance: a misleading debate? Small Business Economics, 43(4), 873-886.
- Hossain, M., & Kauranen, I. (2016). Open Innovation in SMEs: a systematic literature review. Journal of Strategy and management, 9(1), 58-73.
- Huizingh, E. K. (2011). Open Innovation: State of the art and future perspectives. Technovation, 31, 2-9.
- Inauen, M., & Shenker-Wicki, A. (2012). Fostering radical innovations with open innovation. European Juornal of Innovation Management, 15(2), 212-231. doi:https://www.emeraldinsight.com/doi/abs/10. 1108/14601061211220986
- InnovationTools.com. (2013). Government's promising role in fostering open innovation. Retrieved 2 November, 2018, from http://www.innovationmanagement.se/imtoolarticles/governments-promising-role-in-fosteringopen-innovation/
- Institute of Policy Studies. (28 January, 2014). Singaporeans still largely conservative, IPS survey finds. Retrieved 9 October, 2018, from straitstimes.com: https://www.straitstimes.com/singapore/singapor

eans-still-largely-conservative-ips-survey-finds

- Ismail, H. S., Yussuf, R. M., & Zulkifli, N. B. (2017). Factors Influencing Implementation of Open Innovation Concept in Malaysian Industries. Journal of Advanced Management Science, 5(1), 35-39.
- Jamrisko, M., & Lu, W. (23 January, 2018). The U.S. Drops Out of the Top 10 in Innovation Ranking. Retrieved 11 November, 2018, from Bloomberg: https://www.bloomberg.com/news/articles/2018-01-22/south-korea-tops-global-innovation-rankingagain-as-u-s-falls
- JP Morgan. (2014). Asean's Bright Future: Growth Opportunities For Corporates In The Asean Region. Retrieved 8 October, 2018, from JP Morgan: https://www.jpmorgan.com/country/US/EN/cib/i nvestment-banking/trade-asean-future
- Kammerlander, N., & Van Essen, M. (25 January, 2017). Research: Family Firms Are More Innovative Than Other Companies. Retrieved 8 October, 2018, from Harvard Business Review: https://hbr.org/2017/01/research-family-firmsare-more-innovative-than-other-companies
- Kellermanns, F. W., Eddleston, K. A., Barnett, T., & Pearson, A. (2008). An Exploratory Study of Family Member Characteristics and Involvement: Effects on Entrepreneurial Behavior in the Family Firm. Family Business Review, 21(1), 1-14.
- Kim, J. S., & Mueller, C. W. (1978). Introduction to Factor Analysis: What It Is and How To Do It. Newbury Park: SAGE.
- Koh, A., & Kong, E. (19 August, 2016a). Changing Times For Business Families In Asia: The Role Of Family Business Advisors. Retrieved & October, 2018, from Business Families Institute: https://bfi.smu.edu.sg/news/2016/08/19/changin g-times-business-families-asia-role-family-businessadvisors
- Koh, A., & Kong, E. (16 November, 2016b). Starting small and then scaling up. Retrieved 18 February, 2019, from The Business Timperio:
https://www.businesstimes.com.sg/hub/enterpris e-50-2016/starting-small-and-then-scaling-up

Koh, A., & Kong, E. (2018a). Advisers For The New Generation: The Connector, Collaborator, And The Co-Investor. Singapore: FFI Practitioner. Retrieved from

https://bfi.smu.edu.sg/news/2018/apr/04/adviser s-new-generation-connector-collaborator-and-coinvestor

- Koh, A., & Kong, E. (2018b). Building Growth Enterprises in Singapore: Public Private Partnership. In Succession and Innovation in Smallto-Medium-Sized Enterprises (pp. 211-244). Taipei: Future Career Professional.
- LAERD . (2018). LAERD Statistics. Retrieved 6 January, 2019, from Principal Components Analysis (PCA) using SPSS Statistics: https://statistics.laerd.com/spsstutorials/principal-components-analysis-pca-usingspss-statistics.php
- LAERD Statistics. (2018). Multiple Regression Analysis using SPSS Statistics. Retrieved 8 January, 2019, from statistics.laerd.com: https://statistics.laerd.com/spsstutorials/multiple-regression-using-spssstatistics.php
- Lee, S., Park, G., Yoon, B., & Park, J. (2010). Open innovation in SMEs--An intermediated network model. Research Policy, 39(2), 290-300.
- linkflow Capital. (2018). SME Financing Ultimate Guide - Insider's Tips To Secure Funding. Retrieved 7 December, 2018, from https://smeloan.sg/smefinancing-guide
- Lopez-Fernandez, M., Serrano-Bedia, A. M., & Gómez-López, R. (2015). Determinants of innovation decision in small and medium-sized family enterprises. Journal of Small Business and Enterprise Development, 23(2), 408-427.
- Lu, Y., Tsang, E., & Peng, M. (2008). Knowledge management and innovation strategy in the Asia Pacific: Toward an institution-based view. Asia Pacific Journal of Management, 25(3), 361-374.
- Lung, N. (16 May, 2018). Singapore Minister for Trade and Industry highlights the role of SME in innovation. Retrieved 17 December, 2018, from OpenGov: https://www.opengovasia.com/singapore-

minister-for-trade-and-industry-highlights-the-roleof-sme-in-innovation/

- Ma, Y., Yin, Q., Pan, Y., Cui, W., Xin, B., & Rao, Z. (2018). Green Product Innovation and Firm Performance: Assessing the Moderating Effect of Novelty-Centered and Efficiency-Centered Business Model Design. Sustainability, 10(6).
- Maher, L. (2014). Building a culture for innovation: a leadership challenge. World Hosp Health Services, 50(1), 4-6.
- Maitra, S., & Yan, J. (2008). Principle Component Analysis and Partial Least. Casuality Actuarial Society. Retrieved 13 December, 2018, from https://www.casact.org/pubs/dpp/dpp08/08dpp7 6.pdf
- Mani, S. (2011). Promoting Knowledge-intensive Entrepreneurship in India. In A. Szirmai, W. Naudé, & M. Goedhuys, Entrepreneurship, Innovation, and Economic Development. Oxford : Oxford Scholarship.

- Marín, G. S., Hernández, A. J., del Valle, I. D., & Castillo. M., Á. S. (2016). Organizational culture and family business: A configurational approach. European Journal of Family Business, 6(2), 63-134.
- Mustakallio, M., Autio, E., & Zahra, S. (2002). Relational and Contractual Governance in Family Firms: Effects on Strategic Decision Making. Family Business Review, 15(3), 205-222.
- National Research Foundation. (6 November, 2018). Spurring Academic Research Excellence. Retrieved 7 December, 2018, from https://www.nrf.gov.sg/rie2020/spurringacademic-research-excellence
- Nemet, G. (2009). Demand-pull, technology-push, and government-led incentives for non-incremental technical change. Research Policy, 38(5), 700-709.
- Ng, R., Lim, S., & Wong, P. (19 July, 2018). Singapore: 50 Years of Science and Technology. Retrieved 28 December, 2018, from National University of Singapore: https://lkyspp.nus.edu.sg/gia/article/singapore-50-years-of-science-and-technology
- Ngamsutti, S., & Ussahawanitchakit, P. (2016). Marketing innovation capability and marketing performance: an empirical study of electrical and electronic appliances in Thailand. The Business and Management Review, 7(5), 339-346.
- Öberg, C. (2018). The role of business networks for innovation. Journal of Innovation & Knowledge, 1-5.
- OECD & Eurostat. (2005). Oslo Manual: Proposed Guidelines for Collecting and Interpreting Technological Innovation Data. Retrieved 3 October, 2018, from https://unstats.un.org/unsd/EconStatKB/Knowledg ebaseArticle10270.aspx
- OECD. (2018). OECD: Better Policies for Better Lives. Retrieved 30 October, 2018, from Oslo Manual: Proposed Guidelines for Collecting and Interpreting Technological Innovation Data, 2nd Edition: http://www.oecd.org/sti/inno/oslomanualpropose dguidelinesforcollectingandinterpretingtechnologic alinnovationdata2ndedition.htm
- Ostendorf, J., Mouzas, S., & Chakrabarti, R. (2014). Innovation in business networks: The role of leveraging resources. Industrial Marketing Management, 43(3), 504-511.
- Osunde, C. (4 March, 2017). Family Businesses and Its Impact on the Economy. Retrieved 2018 October, 8, from Journal of Business & Financial Affairs: https://www.omicsonline.org/open-access/familybusinesses-and-its-impact-on-the-economy-2167-0234-1000251.pdf
- Perkmann, M., & Walsh, K. (2007). Universityindustry relationships and open innovation: Towards a research agenda. International Journal of Management Reviews, 9(4), 259-280.
- Pertuze, J., Calder, E. S., Greitzer, E. M., & Lucas, W. A. (2010). Best Practice for Industry -University Collaboration. Retrieved 7 December, 2018, from MIT Sloan Management Review: https://osp.mit.edu/sites/osp/files/uploads/bestp ractices.pdf
- Pervan, S., Al-Ansaari, Y., & Xu, J. (2015). Environmental Determinants of Open Innovation in Dubai SMEs. Industrial Marketing Management, 50, 60-68.

Koh, A. Kong, E. Timperio, G. (2019). An Analysis of Open Innovation Determinants: the Case Study of Singapore based Family owned Enterprises. *European Journal of Family Business*, 9(2), 85-101.

- Reeves, M., & Deimler, M. (2011). Adaptability: The New Competitive Advantage. Retrieved 7 December, 2018, from Harward Business Review: https://hbr.org/2011/07/adaptability-the-newcompetitive-advantage
- Salim, I., & Sulaiman, M. (2011). Impact of Organizational Innovation on Firm Performance: Evidence from Malaysian-Based ICT Companies. Business and Management Review, 1(5).
- Scott, G., & Chaston, I. (2013). Open innovation in an emerging economy. Management Research Review, 36(10), 1024-1036.
- Sharma, P., Chrisman, J. J., & & Chua, J. H. (1997). Strategic Management of the Family Business: Past Research and Future Challenges. Family Business Review,, 10(1), 1-35.
- Shi, X., & Wu, Y. (2017). The Effect of Internal and External Factors on Innovative Behaviour of Chinese Manufacturing Firms. China Economic Review, 46(Supplement), 50-64.
- Shiao, V. (21 May, 2018). An ecosystem to boost innovation and revolutionise business. Retrieved 9 October, 2018, from businesstimes.com.sg: https://www.businesstimes.com.sg/companiesmarkets/an-ecosystem-to-boost-innovation-andrevolutionise-business
- Sigurdson, J. (2000). Knowledge Creation and Innovation in Geographically Dispersed Organizations. Asia Pacific Journal of Management, 17(2), 297-330.
- Singapore Business Federation, HBSC. (2018). SBF ASEAN Outlook Survey. Retrieved 9 October, 2018, from Singapore Business Federation: https://www.sbf.org.sg/images/2018/SG1475\_Sum maryReport.pdf
- Singapore Business Review. (13 February, 2018). Companies knock on Singapore to break into Southeast Asia. Retrieved 8 October, 2018, from https://sbr.com.sg/retail/news/companies-knocksingapore-break-southeast-asia
- Singapore Business Review. (11 June, 2018). Singapore rallied to fifth spot as world's largest foreign investment hub. Retrieved 7 December, 2018, from Singapore Business Review: https://sbr.com.sg/marketsinvesting/news/singapore-rallied-fifth-spot-inlargest-foreign-investment-hubs
- Singapore Department of Statistics. (2017). Infographics - Singapore Economy 2017. Retrieved from singstat.gov: https://www.singstat.gov.sg/~/media/Files/visuali sing\_data/infographics/economy/singaporeeconomy22032018.pdf
- Smart PLS. (2018). Smart PLS. Retrieved 21 January, 2019, from Bootstrapping: https://www.smartpls.com/documentation/algorit hms-and-techniques/bootstrapping
- Spithoven, A., Vanhaverbeke, W., & Roijakkers, N. (2013). Open innovation practices in SMEs and large enterprises. Small Business Economics, 41(3), 537-562.
- Straitstimes.com. (6 October, 2018). New programme to spur digitalisation in local firms. Retrieved 8 October, 2018, from Straitstimes.com: https://www.straitstimes.com/business/newprogramme-to-spur-digitalisation-in-localfirms?xtor=CS3-18

- Szczepańska-Woszczyna, K. (2014). The Importance of Organizational Culture for Innovation in the Company. Forum Scientiae Oeconomia, 2(3), 27-39.
- Tan, K., & Phang, S. (2005). From Efficiency-Driven to Innovation-Driven Economic Growth: Perspectives from Singapore. Singapore: Singapore Management University.
- The Business Times. (3 October, 2018). The family business that lasts. Retrieved 16 October, 2018, from

https://www.businesstimes.com.sg/magazines/we alth-october-2017/the-family-business-that-lasts

- The Family Business Consulting Group. (2019). Innovation in Family Business. Retrieved 22 January, 2019, from thefbcg.com: https://www.thefbcg.com/Innovation-in-Family-Business/
- The Straits Times. (27 January, 2015). How Singapore's family firms can go the distance. Retrieved 16 October, 2018, from https://www.straitstimes.com/opinion/howsingapores-family-firms-can-go-the-distance
- Tidd, T., & Bessant, J. (2013). Managing Innovation: Integrating Technological, Market and Organizational Change (5th Edition ed.). Chickester: Wiley.
- Timperio, G., Tan, K., Fratocchi, L., & Pace, S. (2016). The impact of ethnicity on luxury perception: the case of Singapore's Generation Y. Asia Pacific Journal of Marketing and Logistics, 28(2), 217-233.
- Tsang, E. W. (2004). Toward a Scientific Inquiry into Superstitious Business Decision-Making. Organization Studies, 25(6), 923-946.
- Tuan, N., Nhan, N., Giang, P., & Ncog, N. (2016). The effects of innovation on firm performance of supporting industries in Hanoi, Vietnam. Journal of Industrial Engineering and Management, 9(2), 413-431.
- Ulusoy, G., Gunday, G., Alpkan, L., & Kilic, K. (2008). Effects of organizational innovations on firm's production performance. 2008 IEEE International Conference on Industrial Engineering and Engineering Management. Singapore: IEEE. doi:10.1109/IEEM.2008.4737883
- United Nations. (2018). World Economic Situation Prospects. Retrieved 8 October, 2018, from United Nations: https://www.un.org/development/desa/dpad/wpcontent/uploads/cites/45/publication/WFSD2018\_F

content/uploads/sites/45/publication/WESP2018\_F ull\_Web-1.pdf

- US News. (2018). Singapore. Retrieved 16 October, 2018, from https://www.usnews.com/news/bestcountries/singapore
- Van de Vrande, V., De Jong, J., Vanhaverbeke, W., & de Rochemont, M. (2009). Open innovation in SMEs: Trends, motives and management challenges. Technovation, 29, 423-437.
- Vinzi, E., Chin, W., Henseler, J., & Wang, H. (2010). Handbook of Partial Least Squares: Concepts, Methods and Applications. Heidelberg : Springer -Verlag.
- Voordeckers, W., Van Gils, A., & Van den Heuvel, J. (2007). Board Composition in Small and Medium-Sized Family Firms. Journal of Small Business Management, 45(1), 137-156.
- Ward, P. T., & Duray, R. (2000). Manufacturing strategy in context: Environment, competitive

Koh, A. Kong, E. Timperio, G. (2019). An Analysis of Open Innovation Determinants: the Case Study of Singapore based Family owned Enterprises. *European Journal of Family Business*, 9(2), 85-101.

strategy and manufacturing strategy. Journal of Operations Management, 18(2), 123-138.

- West, J. (2003). How open is open enough? Melding proprietary and open source plartform strategies. Research Policy, 32, 1259-1285.
- WIPO. (10 July, 2018). Global Innovation Index 2018: China Cracks Top 20. Top Rankings: Switzerland, Netherlands, Sweden, UK, Singapore, U.S. Retrieved 30 November, 2018, from WIPO: https://www.wipo.int/pressroom/en/articles/201 8/article\_0005.html
- Wolff, J. A., & Pett, T. L. (2006). Small-Firm Performance: Modeling the Role of Product and Process Improvements. Journal of Small Business Management, 44(2), 268-284.
- Wong, C. (2008). A knowledge-based economy: the case of Singapore. International Journal of Knowledge, Culture and Change Management, 8(6), 169-180.
- Wonglimpiyarat, J. (2011). Government programmes in financing innovations : comparative innovation system cases of Malaysia and Thailand. Technology in Society, 33(1-2), 156-164.
- Wynarczyk, P., Piperopoulos, P., & Mcadam, M. (2013). Open innovation in small and mediumsized enterprises: An overview. International Small Business Journal, 31(3), 240-255.
- Yadav, M. S., Prabhu, J., & Chandy, R. K. (2007). Managing the Future: CEO Attention and Innovation Outcomes. Journal of Marketing, 71(4), 84-101.
- Yang, J. (2010). The knowledge management strategy and its effect on firm performance: A contingency analysis. International Journal of Production Economics, 125(2), 215-223.
- Yeoh, A. (2004). Cosmopolitanism and its exclusions in Singapore. Urban Studies, 41(12), 2431-2445.
- Zahra. (2005). Entrepreneurial Risk Taking in Family Firms. Family Business Review, 18(1), 23 - 40.
- Zhu, Y., Wittmann, X., & Peng, M. W. (2011). Institution-based barriers to innovation in SMEs in China. Asia Pacific Journal of Management, 1-12.

Koh, A. Kong, E. Timperio, G. (2019). An Analysis of Open Innovation Determinants: the Case Study of Singapore based Family owned Enterprises. *European Journal of Family Business*, 9(2), 85-101.



EUROPEAN JOURNAL OF FAMILY BUSINESS

0

http://www.revistas.uma.es/index.php/ejfb

# Collaborative innovation in the family SME: conceptualization, goals, and success factors

Unai Arzubiaga<sup>a\*</sup>, Amaia Maseda<sup>a</sup>, Amaia Uribarri<sup>a</sup>, Jesús Manuel Palma-Ruiz<sup>b</sup>

<sup>a</sup>University of the Basque Country, UPV/EHU (Spain) <sup>b</sup>Universidad Autónoma de Chihuahua (Mexico)

Received 28 September 2018; accepted 19 December 2019

JEL CLASSIFICATION M19, O31, O32

KEYWORDS Collaborative innovation, Family SME; Innovative capabilities; Innovative attitudes; Management team

CÓDIGOS JEL M19, O31, O32

PALABRAS CLAVE Innovación colaborativa; PYME familiar; capacidades innovadoras; actitudes innovadoras; equipo directivo **Abstract** In a constantly changing environment, collaborative innovation enables the knowledge creation and new product designs, the improved efficiency of the production process, and the reduction of time-to-market. However, the achievement of such results in the family SME depends mainly on the unique characteristics of this type of organization, which in turn, represent the most widespread kind of business worldwide. Therefore, the objective of this article is to analyze how the composition of the management team, the factors related to the capabilities -cognitive factors, absorptive capacity, and innovative trajectory- and the attitudes -preservation of SEW and intra-organizational behavior- of the decision makers, mainly influenced by the family, affect when designing and implementing collaborative innovation processes in a successful way.

## La innovación colaborativa en la pyme familiar: conceptualización, objetivos y factores de éxito

**Resumen** En un entorno en constante cambio, la innovación colaborativa permite la creación de conocimiento y de nuevos diseños, la mejora de la eficiencia del proceso de producción y la reducción de tiempo para la comercialización de los nuevos productos. Sin embargo, la consecución de tales resultados en las pymes familiares depende en buena medida de las características propias de este tipo de organizaciones, que a su vez representan el tipo de empresa más extendida a nivel mundial. Por lo tanto, el objetivo de este artículo es analizar como la composición del equipo directivo, los factores relacionados con la capacidad -factores cognitivos, capacidad absorptiva y trayectoria innovadora- y las actitudes -preservación del legado socio-emocional y comportamiento intra-organizacional- de los decisores, en buena medida influenciados por la familia, afectan al momento de diseñar e implementar los procesos de innovación colaborativa de manera exitosa.

#### http://dx.doi.org/10.24310/ejfbejfb.v9i2.5265

Copyright 2019: Unai Arzubiaga, Amaia Maseda, Amaia Uribarri, Jesús Manuel Palma-Ruiz. European Journal of Family Business is an open access journal published in Malaga by UMA Editorial. ISSN 2444-8788 ISSN-e 2444-877X This work is licensed under a Creative Commons Atribution-NonCommercial-ShareAlike 4.0 International License (CC BY-NC-SA 4.0).

This study is based on the Master's Thesis titled "Innovation in the Basque family business" of the Master's Degree in Business Management from an Innovation and Internationalization Perspective of the University of the Basque Country (UPV/EHU), developed by the student Amaia Uribarri and supervised by the directors Amaia Maseda and Unai Arzubiaga

## Introduction

In today's dynamic and global environment, in which the demands of the different agents that intervene in the market change at a dizzying speed and the development of new technologies is continuous, companies are forced to adapt to new scenarios to offer innovative answers (Paunov, 2012). It is in this context that one can easily understand the reason why studies on innovation processes have boomed in recent vears (Holt and Daspit, 2016; Kraiczy, Hack, and Kellermans, 2014). Far from being conceived as a delimited and automatic linear, process, innovation is considered as a changing process, with no apparent limits and, above all, dynamic (Chang, Hughes, and Hotho, 2011). This process allows the experience and knowledge of different people and organizations to interact, that is to say, that the know-how flows between the various agents, favoring its feedback (Jensen, Johnson, Lorenz, and Lundvall, 2007). Thus, the mechanisms that allow interaction within the organizations themselves (collaboration between different units, or the participation of the company's personnel in the innovation processes) and the networks with which the company relates to its environment (other companies, universities, research, and technology centers) are gaining increasing prominence (Öberg, 2016). In this context, De Massis, Frattini, and Lichtenthaler (2012) argued that given the interaction between agents has a significant impact on the future of innovation, and collaborative innovation will have a very prominent role soon, both internally (intraorganizational collaboration) and externally (inter-organizational collaboration). Collaborative innovation is defined as voluntary

agreements among independent firms, who exchange and share capital, information, knowledge, and technology to achieve a common innovation goal (Feranita, Kotlar, and De Massis, 2017; Un, Cuervo-Cazurra, and Asakawa, 2010). It is a particularly interesting strategic option for small and medium-sized enterprises (SMEs) since it enables the development of new resources and capabilities to maintain and improve their competitiveness in the market (Muñoz-Bullón, Sanchez-Bueno, and De Massis, 2019). Besides, it allows to have resources that could not be obtained otherwise, or that would imply an excessive cost, and all this without having to give up the desire to be creative and innovative (Miles, Miles, and Snow, 2005). However, this activity is not without risks, given the complexity of the process and the numerous agents and factors that intervene. This complexity is accentuated by the nature of family-owned SMEs, which represent between 80-90% of commercial companies and are responsible for 70% of the employment generated in the private sector (Instituto de Empresa Familiar de España, 2016). In this regard, the lack of studies on collaborative innovation in the field of family businesses is particularly striking (Casprini et al., 2017; De Massis, Frattini, and Lichtenthaler, 2012; Feranita et al. 2017). Thus, the research carried out on the collaborative innovation process has focused mainly on the study of large companies (Spithoven, Vanhaverbeke, and Roijakkers, 2013), leaving the smaller ones relegated to a second stage, even though these smaller companies, as previously stated, represent the critical element of economic and social development.

For all the above reasons, this article focuses on the collaborative innovation processes in the family SME. For this purpose, a conceptualization of this phenomenon is presented by analyzing the factors that affect the success of the family SME when dealing with collaborative innovation processes, as well as the perceived benefits. In this way, this article makes at least two theoretical contributions. Firstly, a contribution to the innovation literature is formed by carrying out a conceptualization of the collaborative innovation process, aiming to deepen in its solid foundation to delve into and inspire a more rigorous approach. Secondly, a comprehensive approach of collaborative innovation processes in the context of the family business is offered, by identifying their distinctive characteristics and how such can influence this type of strategy.

From a practical point of view, this study also contributes to those responsible for designing and executing public policies in the field of innovation. It is expected for public institutions to act as facilitators of business innovation processes (Kontinen and Ojala, 2011), with the difficult task of distributing the limited public resources among companies that want to pursue innovative activities (Zúñiga-Vicente, Alonso-Borrego, Forcadell, and Galán, 2014). In this regard, two of the main distinctive features of family businesses are their long-term orientation and their close ties with the communities where they are based (Lumpkin, Brigham, and Moss, 2010), the fact of highlighting the benefits that family SMEs attain from innovation processes enables public institutions to assess the effect of the invested public resources more precisely. The remainder of this article is structured as follows. Next section describes a conceptualization of collaborative innovation, highlighting which are objectives sought through the the implementation of this type of processes. Subsequently. the main distinctive characteristics of family SMEs are briefly discussed. Afterward, the different elements that make up the theoretical model of

Arzubiaga, U., Maseda, A., Uribarri, A., Palma Ruiz, J.M. (2019). Collaborative innovation in the family SME: conceptualization, goals, and success factors. *European Journal of Family Business*, *9*(2), 12-114.

collaborative innovation processes in family SMEs are thoroughly explained, focusing on the composition of the management team as well as on the differential factors related to capacity and attitude of family SMEs. In the last section, the contributions of this study are summarized, and a series of relevant aspects are outlined for future studies.

## Collaborative Innovation

## Conceptualization

Collaborative innovation is defined as the creation of innovations beyond the limits of the company, and even the industry or sector, through the exchange of ideas, knowledge, experiences. and opportunities (Ketchen. Ireland, and Snow, 2007). It refers to a process of creation and development that involves multiple actors, from outside and from within organizations, working together in order to generate ideas, concepts or solutions in the form of product, process or service (Skippari, Laukkanen, and Salo, 2017) for business or for their own use (Haefliger, 2012). During its development and regularly, the collaborating agents reveal the results of their individual and collective efforts with the agreed partners (Baldwin and von Hippel, 2011).

Social capital is defined by Bourdieu (1986: 248) as "the sum of real or potential resources linked to the possession of a lasting network of relations of knowledge and mutual recognition." It is considered a key element for strategic collaborations, increasing the probability of successful collaborations due to the trust and willingness to share resources among the partners (Hitt, Ireland, Camp, and Sexton, 2001; Nahapiet and Ghoshal, 1998; Siebert, Kraimer, and Liden, 2001). In this sense, Galán and Castro (2004: 108) pointed out that "confidence can lead to joint efforts and, for this reason, it is considered as an antecedent and an extraordinary lubricant of collaboration." Also, they added that "when two units begin to trust one another the willingness to share resources increases without worrying about the advantages that the other party will incur."

In this way, the existence of inter-organizational trust implies excellent coordination of tasks among the companies that sustain relationships or transactions, providing relevant knowledge about their norms, routines, and procedures (Gulati, Nohria, and Zaheer, 2000). In this regard, Davis and Eisenhardt (2011) emphasized the significant interactions among the several members of the supply chain, which is one of the main productive ecosystems where collaborative innovation occurs, through the search for complementary partners with the resources needed (Venkatesh and Yadav, 2011).

## Objectives of collaborative innovation

Companies committed to collaborative innovation pursue several specific goals that can be grouped in three major groups: enabling knowledge creation and new product designs, improved efficiency of the production processes, ad reduction of time-to-market (Skippari et al., 2017).

## Enabling knowledge creation and new product designs

Collaboration among different organizations or agents in the innovation chain, from the idea generation to its conversion into a product or service, stimulates the cross-fertilization of shared knowledge and experiences (Swink, 2006), which leads to a higher number of initiatives on new products or services (Faems, van Loov, and Debackere, 2005). These benefits can be especially valuable in the case of technological innovations by facilitating staff involved in R&D activities with greater access to information and experiences (Roy and Sivakumar, 2010). For example, the collaboration between different organizations increases the quality of product design solutions, thus increasing their attractiveness to customers (Skippari et al., 2017).

Collaborative innovation can also be exciting when companies interact with either potential or current customers (Haefliger, 2012). On this regard, one of the most novel phenomenon developed in recent years is the co-design of products, which is the result of the collaborative work between companies and consumers (Fuchs and Schreier, 2011). This collaboration allows consumers to benefit from improvements in the products they usually consume, ensuring that these products will have better acceptance in the market, thus reducing the likelihood of better understanding rejection by what customers value (Tsai, 2009). In this search and identification of new markets, it is also worth noting that customized products and services tailored to niche audiences, which are willing to pay more if they can design the product themselves (Franke, Schreier, and Kaiser, 2010).

## Improved efficiency of the production process

Collaborative innovation can also be beneficial concerning efficiency and cost reduction in the development and production stages (Min et al., 2005). Thus, the collaboration between different companies allows managing learning and knowledge of the product creation processes (Öberg, 2016). A partnership facilitates cost reduction and enables the maximization and shared use of product platforms, global product designs, and generations of such products, among others (Swink, 2006).

A clear example of this practice is that carried out by the Spanish family supermarket chain Mercadona with its suppliers (Negocios en Navarra, 2016). Thus, Mercadona establishes collaborative ties with its inter-suppliers both in processes (reducing electricity and water consumption, minimizing waste, and optimizing logistics) and products. Mercadona, aware of its customers' needs and tastes, transfers them to its suppliers specialized in manufacturing, with whom it maintains stable and long-term relationships. commercial Thus. this joint consideration provides richer insights on network innovation output, producing in 2015 alone more than 100 new process improvements and at the same time, establishing solid foundations to collaborate in product and process development shortly.

## Reduction of time-to-market

Collaborative innovation can also result in a reduction in the time needed to commercialize a new product, a factor that allows companies to extend their market participation (Davis and Eisenhardt. 2011). Multi-organizational innovation teams tend to find solutions more guickly since they have a full range of knowledge sources (Ganesan et al., 2009), which allows faster and more numerous iterations of designs (Holmen, Aune, and Pedersen, 2013). On the other hand, collaboration encourages the reuse and better use of previous design and development work (Street and Cameron, 2007). Besides, the partnership allows the development process to begin without the need to fully complete the last phase design since the most relevant information is accessible to those responsible for making decisions during the product development phase (Swink, 2006).

Finally, although collaborative innovation, in general, leads to the development of new products and services (Rumball, 2007), how the process is developed and even the results acquired vary depending on the type of company that carries it out (Filip, Hansen, and Frölunde, 2016). SMEs have fewer resources than necessary to carry out basic research (Roxas, Piroli, and Sorrentino, 2011). Thus, collaborative innovation processes of SMEs often include as collaborating agents to universities, vocational training institutes and communities, groups and business clusters arising around some of the aspects to be developed as well as to different companies involved in the innovation process (Von Hippel and Von Krogh, 2003).

## The Family SME and its unique characteristics

A family business is identified as such by the participation of the family in the company. The familv influence is determined regarding ownership, government management, and (Steiger, Duller, and Hiebl, 2015; Mazzi, 2011). In addition to family involvement, the behavior and desire to be a family business is undoubtedly another distinguishing feature of this type of organization (Dawson and Mussolino, 2014; Chrisman, Chua, and Sharma, 2005). In this sense, one of the most recognized definitions of family business is the one proposed by Chua, Chrisman, and Sharma (1999), as that entity where the government and management falls on a dominant coalition controlled by members of the same family or a small number of families, whose desire is the sustainable maintenance of the business for future family generations.

The actions of family businesses are based on the dynamic interaction between family and business subsystems with a transgenerational expectation; that is, a desire to keep the company under the family control throughout different generations Reeb, 2003; (Anderson and Habbershon, Nordqvist, and Zellweger, 2010), differentiating nonfamily counterparts from (Zellweger, Eddleston, and Kellermanns, 2010). Thus, family businesses tend to exhibit a clear long-term orientation in their strategic decisions (Le Breton-Miller and Miller, 2006). In this sense, Goel and Jones (2016) pointed out that the need to balance and align the interests of the family and the company means that family businesses have resources and unique governance that directly affects their strategic decision-making. The desire to maintain control of the company in the long-term can translate into a more conservative behavior to avoid risk exposure (Gómez-Mejía, Haynes, Núñez-Nickel, Jacobson, and Moyano-Fuentes, 2007). This fact is known in the literature as the desire to maintain the socio-emotional wealth (SEW), which is one of the priority objectives of family businesses. SEW refers to a set of intangible elements such as the feeling of belonging, the perpetuation of family values, the preservation of the family dynasty, or family altruism, among others (Gómez-Mejía et al., 2007).

The distinctive cultural elements of the family business, long-term orientation, and risk aversion as a result of their desire to preserve the socioemotional wealth have a positive effect on strategic decision-making, and therefore, on the adoption of innovation strategies (Arzubiaga, 2019). These strategies are driven by the management team, who are usually influenced by the opinion of the family, given that family members are usually included in the top management team or the board of directors (Minichilli, Corbetta, and MacMillan, 2010). The

Arzubiaga, U., Maseda, A., Uribarri, A., Palma Ruiz, J.M. (2019). Collaborative innovation in the family SME: conceptualization, goals, and success factors. *European Journal of Family Business*, 9(2), 12-114.

fact that the management team is composed of family and non-family members of different generations and that there might be managers who are also owners means that both businessand family objectives have to be considered simultaneously (Kraiczy et al., 2014; Zellweger, 2007).

## Determining factors in collaborative innovation in the Family SME

The collaborative innovation process has a series of unique characteristics when it takes shape in

the family SME. In line with the above, the factors related to the capabilities and attitude of these companies affect their collaborative innovation processes. Likewise, these factors are also influenced by the characteristics of the management teams, in which the familial element can have a significant influence (Rondi, De Massis, and Kotlar, in press), farther in the case of SMEs given the limited number of personnel in the top management (Figure 1).



Figure 1 Factors related to the success of collaborative innovation in the family SME.

Based on the scheme presented in Figure 1, the following sections elucidate on each of the elements that contribute to and affect the collaborative innovation process of the family SME.

#### The management team composition

The top management team composition and its diversity are aspects with a strong influence on the behavior and decision-making of the organization (Ling and Kellermanns, 2010). In addition to the diverse elements common to any company, family SMEs have two diversity factors, which are considered unique and differentiated (Kraiczy et al., 2014): the ratio of family members in the management team and the number of generations involved in the management team (Arzubiaga, Maseda, and Iturralde, 2017).

Concerning the presence of family members in top management, their influence as a group stems from an educational base and common organizational culture, sharing experience, and knowledge acquired over time (Lozano-Posso and Urbano, 2017; Minichilli et al., 2010). Besides, they have unique values such as commitment, long-term orientation, and customer service, which gives them a more robust organizational culture and values (Chrisman, Chua, Pearson, and Barnett, 2012). However, a high percentage of family members in the management team can also minimize the broad-mindedness and knowledge of other organizations (Kraiczy et al., 2014), a pivotal point to achieve high efficiency in innovation processes. In this sense, the presence of non-family managers usuallv provides more diverse external knowledge and perspectives (Talke, Salomo, and Rost, 2010), due to their different managerial skills acquired outside the family business (Veider and Matzler, 2015) and better network contacts with external advisors. These advisors can provide experiences and technical knowledge in various areas (Classen, Van Gils, Bammens, and Carree, 2012), which can be vital to establishing relationships

Arzubiaga, U., Maseda, A., Uribarri, A., Palma Ruiz, J.M. (2019). Collaborative innovation in the family SME: conceptualization, goals, and success factors. *European Journal of Family Business*, 9(2), 102-114.

with third parties to design and implement collaborative innovation projects.

The number of family members can also affect the attitude of managers when making decisions about innovation in general (Minichilli et al., 2010), and collaborative innovation in particular (Magistretti, Dell'Era, De Massis, and Frattini, 2019). Thus, non-family managers may need to demonstrate that their employment is justified (Hiebl, 2015), so they will seek to increase their managerial impact and leave their professional imprint through the design and implementation of more risky projects (Casillas, Moreno, and Barbero, 2011). Also, this fact accentuates the difference in family members concerning the attitude toward risk, minimizing the risk of investment in innovation, with the ultimate goal of preserving the socioemotional wealth of the family in the organization (Gómez-Mejía et al., 2010).

Regarding the number of family generations involved in the top management team, such inter-generational presence is considered a vital diversity factor when making decisions about the innovation processes (Kellermanns and Eddleston, 2006). Thus, the involvement of different generations in the management team allows to diversify the knowledge as a result of the various educational backgrounds, experiences (Talke et al., 2010), different perspectives, and even, different network contacts (Chirico, Sirmon, Sciascia, and Mazzola, 2011). In this way, it is easier to identify the needs of new clients and markets, and the innovation processes can be more efficient by combining the new knowledge provided by the new generations with the tacit knowledge contributed by previous generations (Litz and 2001). However, Klevsen, knowledge combinations across different generations also require a flexible attitude that allows integrating this knowledge (De Clercq and Belausteguigoitia, 2015), shelving inter-generational tensions as to how to address the innovation in collaboration with third parties.

In short, the two primary sources of diversity in the management team composition play a prominent role in the factors related to the capabilities and attitudes of family SMEs when designing and implementing collaborative innovation processes.

## Factors related to capabilities

The competence of the decision-making bodies in the strategic area is also of particular importance when launching innovation projects (Talke et al., 2010). Thus, in the field of collaborative innovation, cognitive factors stand out (Skippari et al., 2017), the absorption capacity (Filip et al., 2016) and the innovative trajectory (Hibbert and Huxham, 2010) as critical factors to success in collaborative innovation processes.

## **Cognitive factors**

The cognitive factors of the agents involved in collaborative innovation processes play a crucial role in the design and development of innovations (Corsaro, Cantú, and Tunisini, 2012) and depend, to a large extent, on the knowledge acquired, the experiences lived and the unique social interactions experienced by individuals or teams (Marcel, Barr, and Duhaime, 2010).

As noted above, in family SMEs, most of the times, family members are responsible for leading and making decisions about aspects related to innovation (Sciascia, Mazzola, and Chirico, 2013). Usually, those family members share similar elements such as academic background, business know-how, and business culture acquired over the years (Lozano-Posso and Urbano, 2017; Minichilli et al., 2010). Such excessive homogeneity of cognitive factors, derived from a high proportion of family members in positions of responsibility for innovation processes, can result in a lack of knowledge diversity and diverse perspectives (Chrisman, Fang, Kotlar, and De Massis, 2015). Thus, generating a mental rigidity in the cognitive maps of these decision-makers (König, Kammerlander, and Enders, 2013). It is generally acknowledged that non-family members are the ones contributing more knowledge and new perspectives (Arzubiaga, Iturralde, Maseda, and Kotlar, 2018; Talke et al., 2010), different management capabilities, and better access to external network contacts (Veider and Matzler, 2015). This knowledge diversity and skills promotes the use of external information, thus reinforcing the absorption capacity of the company (Classen et al., 2012).

Nonetheless, the concurrence of new family generations in the decision areas can help alleviate the excessive mental rigidity in the cognitive maps of the management teams with a large proportion of family members of the same generation (Cruz and Nordqvist, 2012). Younger generations will contribute new ideas and successfully network contacts to design collaborative innovation projects (Litz and 2001). example. Klevsen. For the implementation of new technologies and tools (Fang, Kotlar, Memili, Chrisman, and De Massis, 2018), which represents essential knowledge to meet the challenges of a dynamic market (Sciascia et al., 2013). In this sense, family SMEs with high heterogeneous teams into innovation will have а greater tendency towards collaborative innovation and a higher probability of success.

Arzubiaga, U., Maseda, A., Uribarri, A., Palma Ruiz, J.M. (2019). Collaborative innovation in the family SME: conceptualization, goals, and success factors. *European Journal of Family Business*, 9(2), 12-114.

## Absorptive capacity

The absorption capacity refers to the ability of the organization to assess, assimilate, and apply new knowledge from collaborators (Cohen and Levinthal, 1990). It is based on a series of routines and organizational processes through which companies acquire, assimilate, transform, and exploit new knowledge (Zahra and George, 2002). Therefore, absorption capacity plays a vital role when collaborating with different agents in collaborative innovation processes (Reagans, Zuckerman, and McEvily, 2004).

In the family SME, the absorption capacity varies according to the capabilities of the people who lead the innovation processes (Kotlar, De Massis, Frattini, and Kammerlander, 2019; Kraiczy et al., 2014). Thus, SMEs with a more significant proportion of non-family members in the company, represents a diversity factor regarding knowledge, skills, and expertise (Veider and Matzler, 2015), with a higher probability of success in collaborative innovation processes. In this sense, mixed teams have a higher potential absorption capacity that allows them to be more receptive to the acquisition of external knowledge and more effective when assimilating that know-how (Alexiev, Jansen, Van den Bosch, and Volberda, 2010). In the same way, a more significant proportion of non-family members also increases the real absorption capacity of family SMEs, that is, to transform and recombine the information and knowledge acquired from different sources for later exploitation (Rodan and Galunic, 2004).

Given that family SMEs are reluctant to incorporate managers from outside the family, the absorption capacity can be driven by the inclusion of new generations in the decisionmaking processes on innovation projects and (Kellermanns, collaborative innovation Eddleston, Barnett, and Pearson, 2008). Thus, teams responsible for designing and the implementing collaborative innovation projects that enjoy a higher absorption capacity will be more effective as they have a higher ability to the rents resultant from this exploit collaboration.

## Innovative trajectory: depth and breadth

The know-how and expertise accumulated during the organizational life cycle are also influential factors when it comes to success in collaborative innovation processes (Filip et al., 2016). Thus, a long history of innovation leads to a wide range of recombinations of knowledge and experience, which will be of greater importance to the extent that the innovative trajectory has been more profound, in terms of accumulation of knowledge in a specific area, and more broadly, in terms of knowledge diversity in several areas (Davis and Eisenhardt, 2011). Family SMEs with a more substantial proportion of family members tends to have deeper innovative trajectories, acquiring tacit knowledge about a specific area over the years in the company. This gives them an advantage of knowing how to discriminate, within this scope, which collaborative innovation projects will be more likely to succeed, and identify from the first moment those that should be abandoned (Katila and Ahuja, 2002). However, the lack of knowledge diversity, expertise, and network contacts of those teams composed mostly by family members lessen the innovation trajectory (Arzubiaga, Kotlar, De Massis, Maseda, and Iturralde, 2018), which results in less knowledge and experience about the recombinations of novel elements (Ahuja and Katila, 2004) and translates into less effective collaborative innovation processes.

Consequently, balanced teams concerning family and non-family members, in principle, seem to be in a more advantageous situation to address diverse collaborative innovation projects. Relatedly, the presence of new generations in those bodies responsible for collaborative innovation projects can help to alleviate, in a certain way, the limited breadth of ideas and knowledge that senior management teams of family firms usually portray (Sciascia et al., 2013).

## Factors related to attitudes

The second axis on which the collaborative innovation processes of family SMEs pivot deals with the attitudes of the teams responsible for designing and implementing those processes, as shown in Figure 1. In this regard, two different characteristics can be distinguished that affect the collaborative innovation processes, such as the preservation of socio-emotional wealth (SEW) and risk aversion (Gómez-Mejía et al., 2007), as well as intra-organizational behavior (Nordqvist, Sharma, and Chirico, 2014).

## Preservation of SEW

The influence of the familial factor on family businesses has been related both to economic results, including competitive advantage and wealth creation, and to results unrelated to the financial scope, including the preservation of tradition, the strengthening of family ties, and the value creation across generations (Pearson, Carr, and Shaw, 2008). Consequently, decisionmaking also has this twofold facet, pursuing economic objectives, as well as those objectives closely linked to the family (Mahto et al., 2010; Souder, Zaheer, Sapienza, and Ranucci, 2016), such as the preservation of the socio-emotional wealth (Gomez-Mejía et al., 2007). In this sense, family businesses tend to estimate

In this sense, family businesses tend to estimate at all times how different strategic decisions can

Arzubiaga, U., Maseda, A., Uribarri, A., Palma Ruiz, J.M. (2019). Collaborative innovation in the family SME: conceptualization, goals, and success factors. *European Journal of Family Business*, 9(2), 102-114.

affect the business family (Chua, Chrisman, and De Massis, 2015; Vieira, 2014), prioritizing those processes and initiatives that avoid, as far as possible, the assumption of risks for the future of the business family (Gomez-Mejía, Makri, and Kintana, 2010; Kotlar, De Massis, Wright, and Frattini, 2018). Given that collaborative innovation brings uncertainties typical of this type of operations, family SMEs may not be willing to assume certain risks. On the one hand, innovation processes are inherent to chance, since they do not offer certainty of the results (Veider and Matzler, 2015). This uncertainty about the investment recovery, both in financial terms and intangible resources, can jeopardize innovation initiatives in general (Brinkerink and Bammens, 2018; Naldi, Nordqvist, Sjöberg, and Wiklund, 2007). On the other hand, family SMEs are often very reluctant to share knowledge with other collaborators outside the organization (Ireland and Webb, 2007). These companies, many of them located in the industrial sector, have developed for years know-how based on learning-by-doing (Chirico, 2008), resulting in tacit knowledge about a series of specialized products (Duran, Kammerlander, Van Essen, and Zellweger, 2016).

In general, SMEs usually have limited access to resources and lack of specific knowledge and technical expertise within a particular area (George, 2005), leading to difficulties when entering collaborative innovation due to their reluctance to lose experience and not seize knowledge opportunities from the collaborators. Thus, faced with the challenge of disclosing their know-how in addition to not being able to take advantage of what has been contributed by the other participants in the innovation processes, the participation of family SMEs in collaborative innovation with third parties is constrained (Debicki, Kellermanns, Chrisman, Pearson, and Spencer, 2016). Putting at risk the competitive advantage that implies having the differential know-how, in exchange for not obtaining clear benefits.

Given that these issues may affect the future viability of the organization, family SMEs will be conservative when taking part in collaborative innovation processes (De Massis, Chirico, Kotlar, and Naldi, 2013). This conservative attitude will be more accentuated in the case of family SMEs with a large proportion of family members amongst those who make strategic decisions (Kraiczy et al., 2014). This conservative attitude of family members can be weakened with more diversified management teams (Zahra, 2005). Thus, the heterogeneity due to the inclusion of non-family members and the concurrence of different family generations will help to create a prone attitude towards collaboration with third parties in innovation projects.

### Intra-organizational behavior

Innovation is fundamentally a collaborative effort between people who share ideas, perspectives, and values (Adler and Kwon, 2002). That is why social capital, understood as the set of values, norms, and attitudes that foster collaborative dynamics is a resource that favors the exchange of knowledge and information (Nahapiet and Ghoshal, 1998), as well as innovation processes (Sánchez-Famoso, Iturralde, and Maseda, 2015). The family business, given its peculiar characteristics linked to the family with solid business values and strong social ties (Hall, Melin, and Nordqvist, 2001), are companies with essential reserves of social capital in which interand intra-organizational relations have a strong influence on their behavior and dynamics (Sánchez-Famoso, Maseda, and Iturralde, 2017). In family SMEs, family members in charge of innovation decision-making tend to consider internal ideas and perspectives of higher value than those coming from outside the organization (Menon and Pffefer, 2003). Even though, there may be certain misgivings by family members to the recognition of the ideas and advice of the non-family group inside the organization since it can be understood as a transfer of power to those non-family members (Alexiev et al., 2010). Therefore, the internal social capital that the company possesses or the network of external relations acquired (external social capital), can determine to a large extent the predisposition towards collaborative innovation. Hence, family SMEs in which the inclination for external knowledge prevails will encourage collaborative inter-organizational innovation, while those that value internal cooperative dynamics more will opt for intra-organizational innovation. In this sense, greater participation of new generations in the decision-making process of collaborative innovation can play a unifying role between family groups and those of non-family members of the company (Casillas, Moreno, and Barbero, 2010).

## Conclusions and future lines of research

article, deepening the In this а in conceptualization of collaborative innovation strategy has been made as an increasingly common and widespread phenomenon, outlining the three main objectives pursued by this strategy: the knowledge creation and new product designs, the improved efficiency of the production process and the reduction of time-to-However, achievement market. the of collaborative innovation in the family SMEs depends mostly on the unique characteristics of this type of organization. Thus, factors such as the management team composition -the proportion of family members or the number of generations involved in management- factors related to capabilities -the cognitive factors, the absorption capacity, and the innovative trajectory in terms of depth and breadth, as well as those factors referred to preferences conservation of the SEW and inter-organizational behavior- primarily mediated by the influence of the family in all cases, play a crucial role in the successful design and implementation of collaborative innovation.

The main contributions of this work refer to deepen in the solid foundations that allow in the future to delve into the academic study of collaborative innovation and the identification of differentiating characteristics of family SMEs that affect this process, which give rise to future lines of research in this field. On the one hand, a significant advance in this matter would require an empirical study of the model presented in this article, to assess the theoretical development presented here. In this sense, it would be of particular interest to consider the possible moderating effects of the size of the company and sector variables, which would allow refining the impact of the variables of this model. On the other hand, the empirical testing of the model using longitudinal data would shed light on critical decision-making whether or not to be involved in the collaborative innovation process, e.g., the time required for collaborative innovation to bring about benefits for the company.

## References

- Adler, P. S., and Kwon, S. W. (2002). Social capital: Prospects for a new concept. Academy of Management Review, 27(1), 17-40.
- Ahuja, G., and Katila, R. (2004). Where do resources come from? The role of idiosyncratic situations. *Strategic Management Journal*, 25(8-9), 887-907.
- Alexiev, A. S., Jansen, J. J., Van den Bosch, F. A., and Volberda, H. W. (2010). Top management team advice seeking and exploratory innovation: The moderating role of TMT heterogeneity. *Journal* of Management Studies, 47(7), 1343-1364.
- Anderson, R. C., and Reeb, D. M. (2003). Foundingfamily ownership and firm performance: evidence from the SandP 500. *The Journal of Finance*, *58*(3), 1301-1328.
- Arzubiaga, U., Iturralde, T., Maseda, A., and Kotlar, J. (2018). Entrepreneurial orientation and firm performance in family SMEs: the moderating effects of family, women, and strategic involvement in the board of directors. International Entrepreneurship and Management Journal, 14(1), 217-244.
- **Arzubiaga, U. (2019).** Gobernanza corporativa y orientación emprendedora de las pymes familiares: un análisis de la influencia de la implicación familiar a distintos niveles. *Cuadernos de Gestión*, *19*(1), 119-145.
- Arzubiaga, U., Kotlar, J., De Massis, A., Maseda, A., and Iturralde, T. (2018). Entrepreneurial orientation and innovation in family SMEs:

unveiling the (actual) impact of the Board of Directors. Journal of Business Venturing, 33(4), 455-469.

- Arzubiaga, U., Maseda, A., and Iturralde, T. (2019). Exploratory and exploitative innovation in family businesses: the moderating role of the family firm image and family involvement in top management. *Review of Managerial Science*, 13(1), 1-31.
- Baldwin, C., and von Hippel, E. (2011). Modeling a paradigm shift: From producer innovation to user and open collaborative innovation. *Organization Science*, 22(6), 1399-1417.
- Bourdieu, P. (1986). The forms of capital. In J. Richardson (Ed.), Handbook of Theory and Research for the Sociology of Education. New York: Greenwood, 241-258.
- Brinkerink, J., and Bammens, Y. (2018). Family influence and RandD spending in Dutch manufacturing SMEs: The role of identity and socioemotional decision considerations. *Journal of Product Innovation Management*, 35(4), 588-608.
- Casillas, J. C., Moreno, A. M., and Barbero, J. L. (2011). Entrepreneurial orientation of family firms: Family and environmental dimensions. *Journal of Family Business Strategy*, 2(2), 90-100.
- Casillas, J. C., Moreno, A. M., and Barbero, J. L. (2010). A configurational approach of the relationship between entrepreneurial orientation and growth of family firms. *Family Business Review*, 23(1), 27-44.
- Casprini, E., De Massis, A., Di Minin, A., Frattini, F., and Piccaluga, A. (2017). How family firms execute open innovation strategies: the Loccioni case. Journal of Knowledge Management, 21(6), 1459-1485.
- Chang, Y., Hughes, M., and Hotho, S. (2011). Internal and external antecedents of SMEs' innovation ambidexterity outcomes. *Management Decision*, 49(10), 1658-1676.
- Chirico, F. (2008). Knowledge accumulation in family firms evidence from four case studies. International Small Business Journal, 26(4), 433-462.
- Chirico, F., Sirmon, D. G., Sciascia, S., and Mazzola, P. (2011). Resource orchestration in family firms: Investigating how entrepreneurial orientation, generational involvement, and participative strategy affect performance. Strategic Entrepreneurship Journal, 5(4), 307-326.
- Chrisman, J. J., Chua, J. H., Pearson, A. W., and Barnett, T. (2012). Family involvement, family influence, and family-centered non-economic goals in small firms. *Entrepreneurship Theory and Practice*, 36(2), 267-293.
- Chrisman, J. J., Chua, J. H., and Sharma, P. (2005). Trends and directions in the development of a strategic management theory of the family firm. *Entrepreneurship Theory and Practice*, 29(5), 555-576.
- Chrisman, J. J., Fang, H., Kotlar, J., and De Massis, A. (2015). A note on family influence and the adoption of discontinuous technologies in family firms. *Journal of Product Innovation Management*, 32(3), 384-388.
- Chua, J. H., Chrisman, J. J., and De Massis, A. (2015). A closer look at socioemotional wealth: Its flows, stocks, and prospects for moving forward. *Entrepreneurship Theory and Practice*, 39(2), 173-182.

Arzubiaga, U., Maseda, A., Uribarri, A., Palma Ruiz, J.M. (2019). Collaborative innovation in the family SME: conceptualization, goals, and success factors. *European Journal of Family Business*, 9(2), 102-114.

- Chua, J. H., Chrisman, J. J., and Sharma, P. (1999). Defining the family business by behavior. Entrepreneurship: Theory and Practice, 23(4), 19-19.
- Classen, N., Van Gils, A., Bammens, Y., and Carree, M. (2012). Accessing resources from innovation partners: The search breadth of family SMEs. *Journal of Small Business Management*, 50(2), 191-215.
- Cohen, W. M., and Levinthal, D. A. (1990). Absorptive capacity: A new perspective on learning and innovation. Administrative Science Quarterly, 35, 128-152.
- Corsaro, D., Cantù, C., and Tunisini, A. (2012). Actors' heterogeneity in innovation networks. Industrial Marketing Management, 41, 780-789.
- **Cruz, C., and Nordqvist, M. (2012).** Entrepreneurial orientation in family firms: A generational perspective. *Small Business Economics, 38*(1), 33-49.
- Davis, J. P., and Eisenhardt, K. M. (2011). Rotating leadership and collaborative innovation: Recombination processes in symbiotic relationships. Administrative Science Quarterly, 56(2), 159-201.
- Dawson, A., and Mussolino, D. (2014). Exploring what makes family firms different: Discrete or overlapping constructs in the literature?. Journal of Family Business Strategy, 5(2), 169-183.
- De Clercq, D., and Belausteguigoitia, I. (2015). Intergenerational strategy involvement and family firms' innovation pursuits: The critical roles of conflict management and social capital. Journal of Family Business Strategy, 6(3), 178-189.
- De Massis, A., Chirico, F., Kotlar, J., and Naldi, L. (2013). The temporal evolution of proactiveness in family firms: The horizontal S-curve hypothesis. *Family Business Review*, 27(1), 35-50.
- De Massis, A., Frattini, F., and Lichtenthaler, U. (2012). Research on technological innovation in family firms: Present debates and future directions. Family Business Review, 26(1), 10-31
- Debicki, B. J., Kellermanns, F. W., Chrisman, J. J., Pearson, A. W., and Spencer, B. A. (2016). Development of a socioemotional wealth importance (SEWi) scale for family firm research. Journal of Family Business Strategy, 7(1), 47-57.
- Duran, P., Kammerlander, N., Van Essen, M., and Zellweger, T. (2016). Doing more with less: Innovation input and output in family firms. Academy of Management Journal, 59(4), 1224-1264.
- Faems, D., van Looy, B., and Debackere, K. (2005). Interorganizational collaboration and innovation: Toward a portfolio approach. *Journal of Product Innovation Management*, 22, 238-250.
- Fang, H., Kotlar, J., Memili, E., Chrisman, J. J., and De Massis, A. (2018). The pursuit of international opportunities in family firms: Generational differences and the role of knowledge-based resources. *Global Strategy Journal*, 8(1), 136-157.
- Feranita, F., Kotlar, J., and De Massis, A. (2017). Collaborative innovation in family firms: Past research, current debates, and agenda for future research. *Journal of Family Business Strategy*. 8(3), 137-156.

- Filip, D., Hansen, B. D., and Frølunde, T. T. (2016). Interaction enablers, drivers and barriers of collaborative innovation projects between small firms and academia. 7 *Challenging traditional leadership behaviors in a distributed and ambiguous knowledge-intensive work environment*, 25.
- Franke, N., Schreier, M., and Kaiser, U. (2010). The "I designed it myself" effect in mass customization. *Management Science*, 56(1), 125-140.
- Fuchs, C., and Schreier, M. (2011). Customer empowerment in new product development. *Journal of Product Innovation Management*, 28(1), 17-32.
- Galán, J. L., and Castro, I. (2004). Las relaciones interorganizativas como fuente de capital social, *Universia Business Review*, 2(2), 104-117.
- Ganesan, S., George, M., Jap, S., Palmatier, R. W., and Barton, W. (2009). Supply chain management and retailer performance: Emerging trends, issues, and implications for research and practice. *Journal* of *Retailing*, 85, 84-94.
- Gedajlovic, E., Lubaktin, M. H., and Schulze, W. S. (2004). Crossing the Threshold from Founder Management to Professional Management: A Governance Perspective. Journal of Management Studies, 41(5), 899-912.
- George, G. (2005). Slack resources and the performance of privately held firms. Academy of Management Journal, 48(4), 661-676.
- Gersick, K. E., and Feliu, N. (2014). Governing the family enterprise: Practices, performance, and research. In L. Melin, M. Nordqvist, and P. Sharma (Eds.), SAGE handbook of family business. Thousand Oaks, CA: SAGE
- Goel, S., and Jones III, R. J. (2016). Entrepreneurial exploration and exploitation in family business: A systematic review and future directions. *Family Business Review*, 29(1), 94-120.
- Gómez-Mejía, L. R., Haynes, K. T., Núñez-Nickel, M., Jacobson, K. J., and Moyano-Fuentes, J. (2007). Socioemotional wealth and business risks in family-controlled firms: Evidence from Spanish olive oil mills. *Administrative Science Quarterly*, 52(1), 106-137.
- Gomez-Mejia, L. R., Makri, M., and Kintana, M. L. (2010). Diversification decisions in family-controlled firms. *Journal of Management Studies*, 47(2), 223-252.
- Gulati, R., Nohria, N., and Zaheer, A. (2000). Strategic networks. *Strategic Management Journal*, 21(3), 203-215.
- Habbershon, T. G., Nordqvist, M., and Zellweger, T. (2010). Transgenerational entrepreneurship. Transgenerational entrepreneurship: Exploring growth and performance in family firms across generations, 1-38.
- Haefliger, S. (2012). Collaborative Innovation Strategy, Technology, and Social Practice (Habilitation Thesis). ETH Zurich Research Collection.
- Hall, A., Melin, L., Nordqvist, M. (2001). Entrepreneurship as Radical Change in the Family Business: Exploring the Role of Cultural Patterns, Family Business Review, 14(3), 193-2008.

Arzubiaga, U., Maseda, A., Uribarri, A., Palma Ruiz, J.M. (2019). Collaborative innovation in the family SME: conceptualization, goals, and success factors. *European Journal of Family Business*, 9(2), 12-114.

- Hibbert, P., and Huxham, C. (2010). The past in play: Tradition in the structures of collaboration. *Organization Studies*, *31*(5), 525-554.
- Hiebl, M. R. (2015). Family involvement and organizational ambidexterity in later-generation family businesses: A framework for further investigation. *Management Decision*, 53(5), 1061-1082.
- Hitt M. A., Ireland R. D., Camp S. M., and Sexton D. L. (2001). Strategic entrepreneurship: entrepreneurial strategies for wealth creation. *Strategic Management Journal*, 22(6-7), 479-491.
- Holmen, E., Aune, T. B., and Pedersen, A. C. (2013). Network pictures for managing key supplier relationships. *Industrial Marketing Management*, 42(2), 139-151.
- Holt, D. T., and Daspit, J. J. (2016). Diagnosticar la disposición para la innovación en las empresas familiares. *Harvard Deusto Business Review*, (257), 36-50.
- Instituto de Empresa Familiar de España (2016). <u>http://www.iefamiliar.com/cifras/1</u> (accessed 12.03.17)
- Ireland, R. D., and Webb, J. W. (2007). Strategic entrepreneurship: Creating competitive advantage through streams of innovation. *Business Horizons*, 50(1), 49-59.
- Jensen, M. B., Johnson, B., Lorenz, E., and Lundvall, B. Å. (2007). Forms of knowledge and modes of innovation. *Research Policy*, 36(5), 680-693.
- Katila, R., and Ahuja, G. (2002). Something old, something new: A longitudinal study of search behavior and new product introduction. Academy of Management Journal, 45(6), 1183-1194.
- Katz, R., and Allen, T. J. (1982). Investigating the Not Invented Here (NIH) syndrome: A look at the performance, tenure, and communication patterns of 50 R and D Project Groups. *RandD Management*, 12(1), 7-20.
- Kellermanns, F. W., and Eddleston, K. A. (2006). Corporate entrepreneurship in family firms: A family perspective. *Entrepreneurship Theory and Practice*, 30(6), 809-830.
- Kellermanns, F. W., Eddleston, K. A., Barnett, T., and Pearson, A. (2008). An exploratory study of family member characteristics and involvement: Effects on entrepreneurial behavior in the family firm. Family Business Review, 21(1), 1-14.
- Ketchen, D. J., Ireland, R. D., and Snow, C. C. (2007). Strategic entrepreneurship, collaborative innovation, and wealth creation. *Strategic Entrepreneurship Journal*, 1(3-4), 371-385.
- König, A., Kammerlander, N., and Enders, A. (2013). The family innovator's dilemma: How family influence affects the adoption of discontinuous technologies by incumbent firms. *Academy of Management Review*, 38(3), 418-441.
- Kontinen, T., and Ojala, A. (2011). Network ties in the international opportunity recognition of family SMEs. *International Business Review*, 20(4), 440-453.
- Kotlar, J., and De Massis, A. (2013). Goal setting in family firms: Goal diversity, social interactions, and collective commitment to family-centered goals. Entrepreneurship Theory and Practice, 37(6), 1263-1288.
- Kotlar, J., De Massis, A., Frattini, F., and Kammerlander, N. (2019). Motivation gaps and

implementation traps: The paradoxical and time-varying effects of family ownership on firm absorptive capacity. Journal of Product Innovation Management, 1-31

- Kotlar, J., De Massis, A., Wright, M., and Frattini, F. (2018). Organizational goals: antecedents, formation processes and implications for firm behavior and performance. *International Journal* of Management Reviews, 20, S3-S18.
- Kraiczy, N. D., Hack, A., and Kellermanns, F. W. (2014). New product portfolio performance in family firms. *Journal of Business Research*, 67(6), 1065-1073.
- Le Breton-Miller, I., and Miller, D. (2006). Why do some family businesses out-compete? Governance, long-term orientations, and sustainable capability. Entrepreneurship Theory and Practice, 30(6), 731-746.
- Ling, Y., and Kellermanns, F. W. (2010). The effects of family firm-specific sources of TMT diversity: The moderating role of information exchange frequency. *Journal of Management Studies*, 47(2), 322-344.
- Litz, R. A., and Kleysen, R. F. (2001). Your old men shall dream dreams, and your young men shall see visions: Toward a theory of family firm innovation with help from the Brubeck family. *Family Business Review*, 14(4), 335-351.
- Lozano-Posso, M., and Urbano, D. (2017). Relevant Factors in the Process of Socialization, Involvement, and Belonging of Descendents in Family Businesses. *Revista Innovar Journal Revista de Ciencias Administrativas y Sociales*, 26(63), 61-76.
- Lumpkin, G. T., Brigham, K. H., and Moss, T. W. (2010). Long-term orientation: Implications for the entrepreneurial orientation and performance of family businesses. *Entrepreneurship and Regional Development*, 22(3-4), 241-264.
- Magistretti, S., Dell'Era, C., De Massis, A., and Frattini, F. (2019). Exploring the relationship between types of family involvement and collaborative innovation in design-intensive firms: insights from two leading players in the furniture industry. *Industry and Innovation*, 26(10),1121-1151.
- Mahto, R. V., Davis, P. S., Pearce, I. I., John, A., and Robinson Jr, R. B. (2010). Satisfaction with firm performance in family businesses. Entrepreneurship Theory and Practice, 34(5), 985-1001.
- Marcel, J. J., Barr, P. S., and Duhaime, I. M. (2010). The influence of executive cognition on competitive dynamics. *Strategic Management Journal*, 32, 115-138.
- Mazzi, C. (2011). Family business and financial performance: Current state of knowledge and future research challenges. *Journal of Family Business Strategy*, 2(3), 166-181.
- Menon, T. and Pfeffer, J. (2003). 'Valuing internal vs. external knowledge: explaining the preference for outsiders.' *Management Science*, 49, 497-513.
- Miles, R. E., Miles, G., and Snow, C. C. (2005). Collaborative entrepreneurship: How communities of networked firms use continuous innovation to create economic wealth. Stanford University Press.

Min, S., Roath, A. S., Daugherty, P. J., Genchev, S. E., Chen, H., Arndt, A. D., and Richey, R. G. (2005). Supply chain collaboration: What's

Arzubiaga, U., Maseda, A., Uribarri, A., Palma Ruiz, J.M. (2019). Collaborative innovation in the family SME: conceptualization, goals, and success factors. *European Journal of Family Business*, 9(2), 102-114.

happening? International Journal of Logistics Management, 16, 237-256.

- Minichilli, A., Corbetta, G., and MacMillan, I. C. (2010). Top management teams in familycontrolled companies: 'familiness,' 'faultlines,' and their impact on financial performance. Journal of Management Studies, 47(2), 205-222.
- Muñoz-Bullón, F., Sanchez-Bueno, M. J., and De Massis, A. (2019). Combining Internal and External R&D: The Effects on Innovation Performance in Family and Nonfamily Firms. Entrepreneurship Theory and Practice, 1042258719879674.
- Nahapiet, J., and Ghoshal, S. (1998). Social capital, intellectual capital, and the organizational advantage. Academy of Management Review, 23(2), 242-266.
- Naldi, L., Nordqvist, M., Sjöberg, K., and Wiklund, J. (2007). Entrepreneurial orientation, risk-taking, and performance in family firms. *Family Business Review*, 20(1), 33-47.
- Negocios en Navarra (2016). Available online: <u>http://www.negociosennavarra.com/mercadona-</u> impulsa-la-innovacion-colaborativa-proveedores/
- Nordqvist, M., Sharma, P., and Chirico, F. (2014). Family firm heterogeneity and governance: A configuration approach. *Journal of Small Business Management*, 52(2), 192-209.
- Öberg, C. (2016). What creates a collaboration-level identity? Journal of Business Research, 69(9), 3220-3230.
- Paunov, C. (2012). The global crisis and firms' investments in innovation. *Research Policy*, 41(1), 24-35.
- Pattikawa, L. H., Verwaal, E., and Commandeur, H. R. (2006). Understanding new product project performance. *European Journal of Marketing*, 40(11/12), 1178-1193.
- Pearson, A.W., Carr, J. C., and Shaw, J. C. (2008). Toward a theory of familiness: a social capital perspective. Entrepreneurship Theory and Practice, 32(6), 949-969.
- Reagans, R., Zuckerman, E., and McEvily, B. (2004). How to make the team: Social networks vs. demography as criteria for designing effective teams. Administrative Science Quarterly, 49(1), 101-133.
- Rodan, S., and Galunic, C. (2004). More than network structure: How knowledge heterogeneity influences managerial performance and innovativeness. Strategic Management Journal, 25(6), 541-562.
- Rondi, E., De Massis, A., and Kotlar, J. (forthcoming). Unlocking innovation potential: A typology of family business innovation postures and the critical role of the family system. *Journal of Family Business Strategy*. 1-13.
- Roxas, S.A., Piroli, G., and Sorrentino, M. (2011). Efficiency and evaluation analysis of a network of technology transfer brokers, *Technology Analysis* and Strategic Management, 23(1), 7-24.
- Roy, S., and Sivakumar, K. (2010). Innovation generation in upstream and downstream business relationships. *Journal of Business Research*, 63(12), 1356-1363.
- Rumball, D. (2007). Case studies of collaborative innovation in Canadian small firms. Small Business Policy Branch, Industry Canada.

- Sánchez-Famoso, V., Iturralde, T., and Maseda, A. (2015). The Influence of Family and Non-Family Social Capital on Firm Innovation: Exploring the Role of Family Ownership, European Journal of International Management, 9(2), 1-15.
- Sánchez-Famoso, V., Maseda, A., and Iturralde, T. (2017). Family involvement in top management team: impact on relationships between internal social capital and innovation, *Journal of Management and Organization*, 23(1), 136-162.
- Sciascia, S., Mazzola, P., and Chirico, F. (2013). Generational involvement in the top management team of family firms: Exploring nonlinear effects on entrepreneurial orientation. *Entrepreneurship Theory and Practice*, 37(1), 69-85.
- Seibert, S. E., Kraimer, M. L., and Liden, R. C. (2001). A social capital theory of career success. Academy of Management Journal, 44(2), 219-237.
- Skippari, M., Laukkanen, M., and Salo, J. (2017). Cognitive barriers to collaborative innovation generation in supply chain relationships. *Industrial Marketing Management*, 62, 108-117.
- Souder, D., Zaheer, A., Sapienza, H., and Ranucci, R. (2016). How family influence, socioemotional wealth, and competitive conditions shape new technology adoption. *Strategic Management Journal*, 38(9), 1774-1790.
- Spithoven, A., Vanhaverbeke, W., and Roijakkers, N. (2013). Open innovation practices in SMEs and large enterprises. Small Business Economics, 41(3), 537-562.
- Steiger, T., Duller, C., and Hiebl, M. R. (2015). No consensus in sight: an analysis of ten years of family business definitions in empirical research studies. *Journal of Enterprising Culture*, 23(1), 25-62.
- Street, C. T., and Cameron, A. F. (2007). External relationships and the small business: A review of small business alliance and network research. *Journal of Small Business Management*, 45(2), 239-266.
- Swink, M. (2006). Building collaborative innovation capability. Research-Technology Management, 49, 37-47
- Talke, K., Salomo, S., and Rost, K. (2010). How top management team diversity affects innovativeness and performance via the strategic choice to focus on innovation fields. *Research Policy*, 39(7), 907-918.
- Tsai, S. P. (2009). Modeling strategic management for cause-related marketing. *Marketing intelligence* and planning, 27(5), 649-665.
- Un, C. A., Cuervo-Cazurra, A., and Asakawa, K. (2010). R&D collaborations and product innovation. *Journal of Product Innovation Management*, 27(5), 673-689.
- Veider, V., and Matzler, K. (2015). The ability and willingness of family-controlled firms to arrive at organizational ambidexterity. *Journal of Family Business Strategy*, 7(2), 105-116.
- Venkatesh, S., and Yadav, M. S. (2011). Innovations in retailing. *Journal of Retailing*, 875(1), S1-S2.
- Vieira, E. S. (2014). Corporate risk in family businesses under economic crisis. Revista Innovar Journal Revista de Ciencias Administrativas y Sociales, 24(53), 61-73.

Arzubiaga, U., Maseda, A., Uribarri, A., Palma Ruiz, J.M. (2019). Collaborative innovation in the family SME: conceptualization, goals, and success factors. *European Journal of Family Business*, 9(2), 12-114.

- Von Hippel, E., and Krogh, G. V. (2003). Open source software and the "private-collective" innovation model: Issues for organization science. *Organization Science*, 14(2), 209-223.
- Zahra, S. A. (2005). Entrepreneurial risk-taking in family firms. *Family Business Review*, 18(1), 23-40.
- Zahra, S. A., and George, G. (2002). Absorptive capacity: A review, reconceptualization, and extension. Academy of Management Review, 27(2), 185-203.
- Zellweger, T. (2007). Time horizon, costs of equity capital, and generic investment strategies of firms. *Family Business Review*, 20(1), 1-15.
- Zellweger, T. M., Eddleston, K. A., and Kellermanns, F. W. (2010). Exploring the concept of familiness: Introducing family firm identity. *Journal of Family Business Strategy*, 1(1), 54-63.
- Zúñiga-Vicente, J. Á., Alonso-Borrego, C., Forcadell, F. J., and Galán, J. I. (2014). Assessing the effect of public subsidies on firm RandD investment: a survey. Journal of Economic Surveys, 28(1), 36-67.

Arzubiaga, U., Maseda, A., Uribarri, A., Palma Ruiz, J.M. (2019). Collaborative innovation in the family SME: conceptualization, goals, and success factors. *European Journal of Family Business*, 9(2), 102-114.



EUROPEAN JOURNAL OF FAMILY BUSINESS

http://www.revistas.uma.es/index.php/ejfb



## Filippo Ferrari<sup>a\*</sup>

<sup>a</sup>School of Economics, Management and Statistics. University of Bologna (Italy)

Received 11 December 2018; accepted 19 December 2019

JEL Abstract Current literature suggests that family involvement has an impact on firms' CLASSIFICATION innovation propensity, but it does not yet seem able to explain in which conditions. 0310 Adopting a curvilinear model, (Olson's circumplex model of family) this research demonstrates that the family's cohesion and flexibility levels play a significant role in the **KEYWORDS** relationship between family involvement and firm innovation propensity. Family Business; Moreover, by investigating a sample of SMEs (N=125) quantitatively, this paper also Innovation highlights that a curvilinear model shows a better fit (in comparison to linear models) for propensity: explaining the family involvement impact on innovation propensity. Finally, implications Circumplex Model; for both scholars and practitioners are discussed. Family Cohesion; Family flexibility CÓDIGOS JEL ¿Demasiado amor obstaculiza la innovación? Participación familiar e innovación de las 0310 empresas en pequeñas y medianas empresas (PYME) de propiedad familiar. Resumen La literatura actual sugiere que la participación familiar tiene un impacto en la PALABRAS CLAVE innovación empresarial, pero no puede explicar en qué condiciones. Adoptando un Empresa Familiar; modelo curvilíneo (el modelo de familia circumplex de Olson) esta investigación propensión a la demuestra que los niveles de cohesión y flexibilidad de la familia juegan un papel Innovación; importante en la relación entre la participación familiar y la propensión a la innovación Modelo empresarial. Circumplejo; Además, al investigar cuantitativamente una muestra de PYME (N = 125), este trabajo Cohesión familiar;

también muestra que un modelo curvilíneo muestra un mejor ajuste (en comparación con los modelos lineales) para explicar el impacto de la participación familiar en la propensión a la innovación. Finalmente, se discuten las implicaciones académicas y prácticas.

http://dx.doi.org/10.24310/ejfbejfb.v9i2.5388

#### Copyright 2019: Filippo Ferrari

Flexibilidad familiar

European Journal of Family Business is an open access journal published in Malaga by UMA Editorial. ISSN 2444-8788 ISSN-e 2444-877X This work is licensed under a Creative Commons Atribution-NonCommercial-ShareAlike 4.0 International License (CC BY-NC-SA 4.0).re licensed under a Creative Commons Atribution-NonCommercial-ShareAlike 4.0 International License (CC BY-NC-SA 4.0).

Corresponding author E-mail: filippo.ferrari5@unibo.it

## Introduction

Family firms have always shown an ambivalent relationship with innovation dynamics. Some scholars found a negative relationship between family business and innovation propensity (Block, 2012; Chen, Hsu, 2009; Chrisman, Patel, 2012); others conversely found a positive relationship (Gudmundson et al., 2003; Kim et al., 2008; Llach, Nordqvist, 2010). Therefore, literature is still inconclusive and seldom suggests an explanation of the relationship between family involvement and innovation levels. On one side, both long-term orientation (Munoz-Bullon et al., 2011; Dieguez-Soto et al., 2016) and social capital could foster the development of new products and processes (Llach, Nordqvist, 2010; Chen, Hsu, 2009; De Massis et al., 2015). On the other side, risk aversion (Chen, Hsu, 2009) and reluctance to make 'horizontal' partnerships (Dohennels, Froling, 1999) could negatively impact the overall innovation rate.

Occasionally literature is even contradictory: for instance, non-family members' exclusion from strategic decisions seems to lead to both negative (Zahara, 2005) and positive (Madanoglu et al., 2016) outcomes in innovation rate. Finally, as suggested by Sciascia et al., (2013), innovation propensity seems to change over time (Zelleweger, Sieger, 2012), but literature lacks of an explanation of this specific feature.

In general, current literature calls for new insights into to what extent, and how, the internal environment of the family affects the governance of the family firm and the family's pursuit of economic and non-economic goals (Arzubiaga et al., 2019; Daspit et al., 2017; Madanoglu et al., 2016). Although some authors (De Massis et al., 2013; Sirmon & Hitt, 2003) highlight the influence of family involvement on firm's innovation rate, such influence is one of the less understood determinants in the innovation propensity of family firms (Duran et al., 2016). Recently, De Massis and colleagues (2015) affirm that the conditions in which the family has an impact on innovation are still unclear, and research has until now led to inconclusive findings (Craig, Moores, 2006; Morck et al., 2000). Some authors (see for instance Carnes, Ireland, 2013) underline the heterogeneity of families; in that different family firms have different innovative outcomes, but they have been (until now) unable to explain. An as of yet unanswered question is "if familiness [i.e. the outcome of family-firm overlapping] helps both stabilizing and enriching processes, but these have opposite effects on innovation, under what conditions does each process prevail?" (Penney, Combs, 2013, 1422). Given this theoretical scenario, the purpose of this paper is to explore how and to what extent different levels of family involvement affect business innovation. In order to achieve this goal, limitations shown by the current organizational literature suggest the use of other theoretical approaches. For instance, family science, and social psychology of the family especially, can help to explain family processes performances (such and as innovation propensity), as claimed by previous literature (James et al., 2012, Dyer, Dyer, 2009). To date, this approach has had a limited impact on family business literature (for a review, see Daspit et al., 2017). By applying Olson's circumplex model to the family firms (Olson, 2000; 2011), this research posits that the family's cohesion and flexibility levels play a significant role in the relationship between family involvement and firm innovation propensity. In doing so, this paper highlights a curvilinear relationship between family involvement and firm innovation propensity, as theoretically supposed by previous literature (Penney, Combs, 2013; Sciascia et al., 2013; Daspit et al., 2017).

This paper contributes to the current literature in a threefold way.

Highlighting a curvilinear relationship between family involvement and propensity for innovation, this paper provides an explanation of the limitations shown by previous theoretical 'linear' approaches in explaining this relationship.

Furthermore, highlighting the different ways and intensity of family involvement in the business, this paper addresses family firms' heterogeneity, as advocated by current literature (Melin, Nordqvist, 2007; Sciascia et al., 2013).

Finally, the circumplex model is widely used by counsellors and practitioners in their professional activity. By showing that this approach is also suitable for investigating family firms dynamics academically, this paper builds a bridge between the currently (and too often) separate worlds of scholars and practitioners.

## Theoretical background

Family business literature has frequently highlighted the impact of family' involvement on firm innovation rate (for a literature review see De Massis et al., 2013). Unfortunately, research has until now failed to find a linear correlation between involvement and innovation: several scholars acknowledge this theoretical gap (Padilla-Melendez al., 2015; et Wright &Kellermans, 2011), in spite of the fact that many theories have attempted to support such a correlation.

For example, 'familiness' (Habbershon *et al.*, 2003), is a variable that can differentiate and characterize the firm, resulting in a competitive

advantage, as suggested by the resource-based view (RBV) (Habbershon et al., 2003; Sirmon and Hitt, 2003; Habbershon and Williams, 1999; Barney, 1991). Recently, Arzubiaga and colleagues (2019) found that family involvement decreases the positive impact on exploratory innovation, but does not improve the impact on exploitative innovation, unlike expected. Cassia et al (2011) found that shared family values, the desire to defend family reputation and high level of communication among family members are enabling factors for innovation. Conversely, a higher level of risk aversion, less professional management and closeness to the external environment seem to be obstacles in innovation development. However, these authors also affirm that "a number of the proposed factors do not appear to discriminate successful from unsuccessful New Product Development processes" (Cassia et al., 2011, 10). Unfortunately, they are not able to explain these findings, advocating for a better understanding of the family dynamics which underline the innovation processes.

Research based on stewardship theory (Davis, Schoorman, and Donaldson 1997) suggests that family members develop a strong sense of belonging and identification with their own family, and thus they are seldom engaged in opportunistic behaviours. The distinctive stewardship orientation of family versus nonfamily firms may idiosyncratically affect the characteristics of the product innovation process (De Massis et al., 2015). Following this approach, Kellermans and colleagues (2012) introduced the concept of family members reciprocity. In their view (see also Astrachan et al., 2002; Klein et al., 2005), reciprocity, by triggering altruistic dynamics and knowledge sharing among family members (Eddleston, Kellermans, 2007), fosters innovation rate. However, their research does not fully support the hypothesis. Their conclusion is that the influence of family involvement on innovation propensity is a complex phenomenon, which does not follow a linear relation, and could result both in positive and negative outcomes.

Literature suggests that also Agency Theory does not show a good fit for explaining high/low innovation rate. On one hand, this approach suggests that a typical value that is considered to significantly influence the dynamics of family firms is *altruism* (Dyer, 2003; Shulze et al., 2001). As argued by Shulze et al. (2001), altruism leads family members to be considerate of one another, promote and sustain the family bond; and this in turn promotes loyalty, commitment to the family firm's long-term prosperity and knowledge sharing. On the other hand, in a family firm, the Principal and the Agent are often the same person, and altruism could affect strategic choices about innovation. For instance, hiring a next generation member who is not properly skilled and, as a consequence, undermining knowledge improvement and eventually hindering the innovation rate. Moreover, literature investigating the decentralization of strategic decisions to nonfamily members is still inconclusive: this kind of delegation could have both negative (Zahara, 2015) and positive (Madanoglu et al., 2016) consequences for innovation.

Finally, research based on *Behavioural Theory* suggests that family firms are strongly focussed on maintaining strategic control among family members. For instance, some authors (Astrachan and Jaskiewicz 2008; Zellweger and Astrachan 2008), highlight the relevance of non-economic goals as development and conservation of socioemotional wealth (SEW) for the family (Gomez-Mejia et al., 2007). Berrone and colleagues affirm that "family firms are typically motivated by, and committed to, the preservation of their SEW, referring to non-financial aspects or "affective endowments" of family owners" (Berrone et al., 2012, p. 259). However, literature suggests that there is also a dark side of involvement that can lead to opportunism, complacency and blind faith (Eddleston & Kidwell, 2012; Ferrari, forthcoming; Steier, 2001; Sundaramurthy, 2008), and can also affect proactive negatively stakeholder engagement (Kellermans et al., 2012). More recently, Dieguez-Solo et al. (2016), analyse the relationship between family involvement and innovation through the SEW lens. They find that this relationship is not entirely clear, leading to negative results in short-term exploration, but positive in long-term exploitation.

In summary, organizational literature hasn't yet answered the question: in a family firm, in which conditions does family involvement foster (or hinder) firm innovation propensity? This paper suggests that current literature fails to explain the innovation process because it searches for a direct relationship between family involvement and innovation levels. Chrisman et al. (2014b) posit that the difference in outcomes could be due to a complex system of factors like continuity, command, community and connections, requiring a non-linear approach in order to explain the outcomes of family involvement. An alternative, psychological approach would therefore suggest that the relationship between family involvement and firm innovation propensity could be curvilinear rather than linear: too high or too low cohesion/ flexibility levels could undermine the propensity to innovation. Thus, drawing on social psychology of the family, this research adopts the circumplex model (Olson, 2011; see Figure 1 below), a curvilinear model of the family involvement.

## The Circumplex model

Social psychology of the family literature shows that family flexibility/adaptability plays a prominent role in ensuring some desired organizational outcomes such as reciprocity, knowledge development and, more in general, balanced relationships between family members (Olson, 2000; Michael-Tsabari, Lavee, 2012).

Every situation experienced by a family can be defined by assessing two different and variable family features: its *cohesion* (how self-oriented the family is) and its *flexibility* or adaptability (how changeable and adaptable the relationships within the family are; for an assessment of these dimensions, see also Beavers & Hampson, 1995; Olson, 2000; 2011; Michael-Tsabari, Lavee, 2012). Family *cohesion* is defined as "the emotional bonding that family members have towards one another" (Olson, 2000: 145), and family *flexibility* refers to the "amount of change in [the family's] leadership, role relationships and relationship rules" (Olson, 2000: 147).

	Low Family High						
High	Chaotically Disengaged	Chaotically Separated	Chaotically Connected	Chaotically Enmeshed			
Family	Flexibly Disengaged	Flexibly Separated	Flexibly Connected	Flexibly Enmeshed			
Flexibility	Structurally Disengaged	Structurally Separated	Structurally Connected	Structurally Enmeshed			
	Rigidly Disengaged	Rigidly Separated	Rigidly Connected	Rigidly Enmeshed			

Figure 1 The circumplex model (source: Olson, 2000).

Regarding the different levels of cohesion, a family could be: disengaged, separated, connected, or enmeshed. A very low cohesion level (disengaged family) is characterized by lack

of commitment, antagonism among family members, and a general family trend for expulsion and/or escape. Disengaged family systems are composed of highly independent members who have little or no attachment or commitment to the family. Members often "do their own thing" without seeking support or guidance from other family members (Olson, 2000: 147).

A low or moderate cohesion level (separated family) is characterized by strong individualism, competition for the resources, and infrequent communication among members (both qualitative and quantitative). A medium-high or high cohesion level (connected family) is characterized by commitment but at the same time respect for individual needs and career/life paths and full support (both economical and emotional) of members' autonomy. Finally, a very high cohesion level (enmeshed family) is characterized by the collective prevailing over individuals; strict resource sharing; continuous boycotting and undermining of members' autonomy.

Moreover, regarding the different levels of internal relationships, flexibility/adaptability, a family could be: strict/rigid, structured, flexible/versatile, or chaotic. A very low flexibility level (rigid family) is characterized by un-modifiable role/iob descriptions. and recursive dvnamics (both affective and relational). In a family characterized by low or moderate flexibility level (structured family) the family system, especially the senior, strictly controls the affective and relational features, restricting and norming them. A high flexibility level (flexible family) is characterized by a flowing relational system, and is well suited to different situations. The leadership is circulating, and makes the most of the juniors (for example, their skills or attitudes). A very flexibility level (chaotic high family) is characterized by a lack of reference points, the relational system is neglected, and the seniors do not provide guidance.

Each dimension/variable (both cohesion and flexibility) could be described in a curvilinear graph, whose extreme values are negative (dysfunctional). Due to the aforementioned dimensions/variables, familiar given the different values of both cohesion and flexibility, each combination could result in 16 different kinds of family. Among these 16 types, family businesses experience the highest likelihood of positive outcomes in families which are balanced, that is at the same time both connected/separated and flexible/structured (Michael-Tsabari, Lavee, 2012).

The circumplex model has been previously applied in order to investigate several family features and family firms' performance (for a review, see Daspit et al., 2017). Lee (2006)

Ferrari, F. (2019). Does too much love hinder innovation? Family involvement and firms' innovativeness in family-owned Small Medium Enterprises (SMEs). *European Journal of Family Business*, 9(2), 115-127.

investigated the impact of balanced cohesion and flexibility on job satisfaction; Nosé, Korunka, Frank, & Danes (2015) found that balanced family structures reduce relationship conflict within the family. Additionally, circumplex research shows that balanced levels of cohesion and flexibility help the family firm to survive through business transmission (Labaki, 2011). More generally, balanced family structures lead to positive outcomes in family performance (Zody, Sprenkle, MacDermid, & Schrank, 2006) even ensure success over multiple and generations (Michael-Tsabari & Lavee, 2012). As a conclusion, ongoing literature supports the application of the circumplex model to family business. However, as remarked on by Daspit and colleagues "these studies tend to address the impact of only one or two extreme family structures on family firms [...], thus potentially overlooking family system effects that a more detailed and nuanced application of circumplex theory might uncover" (2017, 13).

Regarding firm innovation levels, as a curvilinear model, Olson's circumplex model seems well suited for explaining the impact of family involvement on the firm's innovation propensity: as suggested by previous literature, unbalanced/dysfunctional situations could lead to negative outcomes, also for innovation propensity.

Thus, it is possible to set the following hypothesis:

H1: Family firms in an unbalanced situation (rigid, chaotic, disengaged or enmeshed) show lower levels of innovation propensity compared to firms in balanced situations.

## The research

The survey was carried out on a sample of Italian These family SMEs (N=125). firms are characterized by a full overlapping between ownership, family, and management; furthermore, in these firms (at least in the smallest ones) the owners are often directly involved in production (Ward, 1987; Gersick et al., 1997; Tagiuri, Davis, 1996). This kind of firm was chosen due to the fact that they are the most widespread type in the Italian context (Bugamelli et al., 2012; Pellegrino and Zingales, 2014; see also the general framework of Curimbaba, 2002).

The sample

In order to select the participants, the research was carried out with the help of a young entrepreneurs association ('Giovani Impenditori Confindustria'). Starting from the association's database, an exploratory mailing list was formed in order to collect the consensus to participate in the research. Following this, a second more specific mailing was done in order to find the firms with two specific characteristics:

1. A single family must share at least 50% of the ownership;

2. The strategic decisions must be managed by the family

The data gathering started at the beginning of November, 2017, and was completed by the end of April 2018; 125 questionnaires were collected.

## The methodology

Innovation propensity was measured using the Organisational Innovativeness Questionnaire (Wang, Ahmed, 2004). This tool is a well-developed and validated measurement instrument of organisational innovation propensity, based on five factors:

• Product innovativeness, defined as "the novelty and meaningfulness of new products introduced to the market in a timely fashion" (Wang, Ahmed, 2004, 304). Item example: "In new product and service introductions, our company is often first-to-market".

• Market innovativeness, defined as "the newness of approaches that companies adopt to enter and exploit the targeted market" (Wang, Ahmed, 2004, 305). Item example: "In comparison with our competitors, our products' most recent marketing programme is revolutionary in the market".

• Process innovativeness, defined as "an organisation's ability to exploit their resources and capabilities, and most importantly, the ability to recombine and reconfigure its resources and capabilities to meet the requirement of creative production" (Wang, Ahmed, 2004, 305). Item example: "We are constantly improving our business processes".

• Behavioural innovativeness, "demonstrated through individuals, teams and management that enable the formation of an innovative culture, the overall internal receptivity to new ideas and innovation" (Wang, Ahmed, 2004, 305). Item example: "In our company, we support individuals who do things in a different way", and

• Strategic innovativeness, defined as "an organisation's ability to identify external opportunities in a timely fashion and match external opportunities with internal capabilities in order to deliver innovative products and explore new markets or market sectors" (Wang, Ahmed, 2004, 306). Item example: "Key executives of the firm are willing to take risks to seize and explore "chancy" growth opportunities".

Family cohesion and flexibility was measured with the short Italian version of Olson's FACES IV

(Loriedo et al., 2013). This tool provides six different sub-scales for each family dimension:

• Disengaged (item example: "When we are at home, each of us seems to avoid the others");

• Separated/Connected (item example: "Each of us feels involved in the other family members' lives");

• Enmeshed (item example: "We spend too much time together");

• Strict/Rigid (item example: "In family relationships, each of us plays a strictly defined role");

• Structured/Versatile (item example: "Our family explores different work-related problem-solving methods/ ways to solve problems) and

• Chaotic (item example: "In our family we always seem disorganized").

The relationship between family situation and innovation propensity was measured with a correlation test. Moreover, statistically significant relationships were also measured with a regression analysis test.

Empirical evidence (Zahra, 2005; Duran et al., 2016; Zellweger, Sieger, 2012) suggests that when the first generation is still active in the family business, the propensity to innovate is lower. Literature suggests that innovation is often perceived as a threat to the family firm past history (Dyer & Whetten, 2006), and this fact could even lead to organizational failure (Haveman, Khaire, 2004). However, further evidence suggests that the type of involvement matters more than the involved generation (Arzubiaga et al., 2019). Given this ambiguous empirical evidence, the relationship between the family generation and the innovation propensity

was controlled using an ANOVA test.

#### Findings

Statistical analysis based on the available 125 returned, correct, and complete questionnaires shows the following results.

These findings support a negative correlation between the overall organizational innovation propensity and both enmeshed families (-.275) and chaotic families (-.402). Enmeshed families show also a stronger negative correlation with Product Innovation (-.435). Chaotic families also show a negative correlation both with Strategic Innovation (-.330) and Process Innovation (-.373). Disengaged families show a significant negative correlation with Strategic Innovation (-.300). Finally, *rigid* families show no correlation with the overall organizational innovation propensity level. For each statistically significant correlation, a linear regression was calculated to predict innovation level based on specific family situation.

Findings suggest a significant regression equation was found for *enmeshed* business families in affecting overall Innovation ((F 1, 124) = 19.537, p>.05), with an R<sup>2</sup> of .0129; constant = 3.327 -0.10 (enmeshment)) and Product Innovation especially ((F 1, 124 = 29.016, p>.05), with R<sup>2</sup> of 0.1830; constant = 3.529 - 0.187 (enmeshment)). For each point of enmeshment level measured with the Italian short version of Olson's FACES IV, overall family firm Innovation decreased by 0.10 and Product Innovation level decreased by 0.19. Findings also show that a significant regression equation was found for *chaotic* business families in affecting overall Innovation ((F 1, 124) = 23.938, p>.05), with an R<sup>2</sup> of .0.1550; constant =

Table 1 Correlation matrix between variables												
	Enmeshed	Chaotic	Disengaged	Rigid	Cohesion	Flexibility	Innovation	Product	Market	Strategic	Process	Behavioural
Enmeshed	1											
Chaotic	0,3072	1										
Disengaged	0,1186	0,5137	1									
Rigid	-0,0053	-0,2114	-0,1508	1								
Cohesion	-0,2609	-0,408	-0,4732	0,3227	1							
Flexibility	-0,2728	-0,3722	-0,3153	0,0064	0,2776	1						
Innovation	-0,2753***	-0,4023*	-0,169	-0,1003	0,0268	0,1997	1					
Product	-0,4355*	-0,0759	0,1237	-0,1397	-0,2076	-0,1593	0,4691	1				
Market	-0,1735	-0,2214	0,0047	-0,1257	-0,0189	-0,0882	0,5644	0,1634	1			
Strategic	-0,0667	-0,3309***	-0,3006***	0,0337	0,0305	0,2435	0,4102	-0,0045	-0,0286	1		
Process	-0,2347	-0,3733***	-0,2622	0,0667	0,1848	0,3261***	0,7268	0,2224	0,0577	0,3148	1	
Behavioural N=126: *** si	0,1045	-0,2011 α = .05. * sig	-0,0731	-0,0852	0,1257	0,3017***	0,7007	0,0179	0,3732	0,1098	0,487	1

Ferrari, F. (2019). Does too much love hinder innovation? Family involvement and firms' innovativeness in family-owned Small Medium Enterprises (SMEs). *European Journal of Family Business*, 9(2), 115-127.

3.362 -0.135 (chaotic situation)). Strategic Innovation ((F 1, 124 = 15.249, p>.05), with R<sup>2</sup> of 0.1023; constant = 3.284 - 0.183 (chaotic situation)), and Process Innovation ((F 1, 124 = 20.075, p>.05), with  $R^2$  of 0.1323; constant = 3.3845 - 0.213 (chaotic situation)). For each point of chaotic situation level measured with the Italian short version of Olson's FACES IV, family firm Innovation decreased 0.16, Strategic Innovation level decreased by 0.10 and Process Innovation level decreased by 0.21.

Furthermore, findings show that a significant regression equation was found for disengaged business families in affecting Strategic Innovation ((F 1, 124) = 12.321, p > .05), with an R<sup>2</sup> of .083; constant = 3.218 -0.149 (disengaged situation). For each point of disengaged situation level measured with the Italian short version of Olson's FACES IV, family firm Strategic Innovation decreased by 0.08.

In general, family cohesion and flexibility show no significant correlation with the organizational innovation propensity level. However, flexibility shows a positive correlation with Process and Behavioural Innovation (.33 .30 and respectively). Once more, for these statistically significant correlations, a linear regression was calculated to predict innovation level based on flexibility level of the family situation.

Findings show that a significant regression equation was found for *flexible* business families in affecting Process Innovation ((F 1, 124 = 14.753, p>.05), with R<sup>2</sup> of 0.10; constant = 2.412 + 0.353 (flexibility)) and Behavioural Innovation ((F 1, 124 = 12.413, p > .05), with R<sup>2</sup> of .083;constant = 2.246 + 0.339 (flexibility)). For each point of flexibility level measured with the Italian short version of Olson's FACES IV, family firm Process Innovation increased by 0.35 and Behavioural Innovation level increased by 0.34.

Given these results, H1 (Family firms in an unbalanced (rigid, chaotic, disengaged or enmeshed) situation show lower levels of innovation compared to firms in balanced situations) is supported with the exception of a rigid situation. Moreover, the sample shows no

 Table 2 Organizational Innovativeness controlling for generation- average values\*.

difference in organizational innovation propensity controlling for the current owner generation (see Table 2).

### Discussion

### Theoretical contributions

Due to the limitations previously shown by organizational literature in explaining how family involvement affects the firm's innovation rate, this paper fills a theoretical gap shifting the the organizational focus from to the psychological dimension. However, a circumplex model does not replace previous theoretical approaches, but provides a complementary explanation for the relationship between family involvement innovation and propensity. Resource-Based View and Behavioural Theory in particular are effectively complemented by the circumplex model.

Previous research suggested that the Resource Based View does not show a good fit for explaining negative outcomes (Cassia et al., 2011). Literature (Arzubiaga et al. 2019; Minichilli et al. 2010) suggests that family involvement may have a negative impact on innovation propensity due to the level of heterogeneity of managerial knowledge and skills (see also Cruz, Nordqvist, 2012), but the dynamic behind this impact is not yet clear. This research suggests that only extreme scores on the cohesion dimension of the circumplex model (i.e. enmeshed and chaotic family), are negatively associated with positive outcomes due to a overinvolvement or an insufficient involvement of the family in the firm,. Hence, such unbalanced situations could explain the lack of social capital (e.g. network with stakeholders) and human capital (e.g. skills and knowledge) development, dimensions considered strategic by RBV in innovation propensity. For example, enmeshed and strict/rigid situations could hinder the construction of a network of relationships and strategic partnership (social capital), and boycott the next generations' work experience outside the family firm) and their (e.g.

Current	Product	Market	Strategic	Process	Behavioural			
generation	Innovation	Innovation	Innovation	Innovation	Innovation	Innovativeness		
First	3.4	3.1	3.3	3.5	3.6	3.4		
Second	3.2	2.6	2.9	3.6	3.3	3.2		
Third or next	2.9	2.7	2.8	3.6	3.1	3.1		
Sample	3.2	2.8	3.0	3.6	3.3	3.2		

all values are expressed in a range from 1 to 5; a= .05; p= 0.204

Ferrari, F. (2019). Does too much love hinder innovation? Family involvement and firms' innovativeness in family-owned Small Medium Enterprises (SMEs). European Journal of Family Business, 9(2), 115-127.

In addition, this research suggests that the circumplex model helps in explaining the limitations of Behavioural Theory. For instance, striving for control maintenance could undermine next generation's autonomy the and (Ferrari, 2019; 2017). Recent development literature adopting a SEW approach in investigating family firm innovation (see for instance Dieguez-Soto et al., 2016), found a mixed impact (both positive and negative) from familv involvement. Moreover. previous literature describes but does not explain heterogeneity in striving for socioemotional wealth (Chrisman & Patel, 2012), for instance analyzing how socioemotional wealth evolves over generations (Berrone et al., 2012). This study suggests that a limitation in SEW literature could be that it does not consider different levels and types of family involvement. Indeed, cohesion levels which/that are too high, resulting in an enmeshed family, could explain negative effects on the relational system, and consequently generating differences in the extent to which family and non-family firms invest in innovation and the way they manage innovation process, as suggested by the literature (De Massis et al., 2015). In other words, this research suggests that too much or too little attention to socioemotional wealth leads to negative outcomes in family firm innovation propensity.

Beyond the adopted theoretical approach, one major criticism of family business literature is that family firms have been treated as a homogeneous population, as highlighted by some authors (Melin, Nordqvist, 2007; Sciascia et al., 2013). Literature often reflects the underlying heterogeneity of family businesses (e.g., Chrisman, Chua, Pearson, & Barnett, 2012), heterogeneity that forces researchers to go beyond a simple family versus non-family dichotomy to explain why the overlapping between the family and the business has different effects in different family firms. This paper highlights that, considering the levels of cohesion and flexibility, the heterogeneity of family businesses can result in (up to sixteen) different situations, each potentially different from the others regarding the impact on business performance.

In summary, previous theoretical approaches have offered suitable explanations for innovation propensity in balanced situations (separated, connected, structured, flexible/versatile family and all combinations of these), while the circumplex model shows a better fit in also explaining the (scarce) innovation propensity in unbalanced situations (enmeshed, disengaged, and chaotic). Furthermore, by specifically measuring the levels of involvement and flexibility of entrepreneurial families, the circumplex model seems to be more accurate in addressing firm heterogeneity, as advocated by recent (and current) literature (Madanoglu et al., 2016; Arzubiaga et al., 2019).

## Specific empirical contributions

This paper provides evidence that unbalanced families show the lowest innovation propensity levels: accordingly with Hypothesis 1, balanced levels of family cohesion and flexibility show no with significant correlation the overall organizational innovation propensity level. Furthermore, by identifying specific correlations between factors, it is therefore possible to design targeted interventions in order to improve innovation (see below for practical implications). For instance, flexibility shows a positive correlation with Process and Behavioural Innovation (.33 and .30 respectively). This fact explained by considering could be that innovation in processes and behaviours demands new ways to face organizational routines: hence, flexibility seems to show a good fit in ensuring that organizational outcomes are achieved. The effects of some flexibility-fostering managerial practices, (e.g. job rotation: Ortega, 2001) and other similar human resource innovations (e.g. horizontal rather than vertical internal career), performance have been documented on (Ichniowsky et al. 1996, 1997, 1999).

In contrast, cohesion shows a weak, but negative correlation with Product Innovation (-.21). 'Product innovativeness' is most often referred to as 'perceived newness, novelty, originality, or uniqueness of products' (Henard and Szymanski, 2001): all these features are competence-based and derived from an intimate knowledge of materials and their characteristics. Thus, product innovation is also grounded on knowledge sharing and training activities with non-family members and/or consultants, and cohesion could hinder that processes. Therefore, these findings suggests that, even in a balanced familv. cohesion undermines 'product innovativeness'.

Disengaged families show a significant negative correlation with Strategic Innovation (-.30), and a weak negative correlation with Process Innovation (-.26). Disengaged family systems consist of family members who are not cohesive and have little or no family loyalty, thus undermining 'behavioural innovativeness' (both at individual and team levels), which in contrast demonstrates management's willingness to change, and commitment to encouraging new ways of doing things, as well as the willingness to foster new ideas (Rainey, 1999). At the same time, 'process innovativeness' is fostered by contributions in terms of ideas, new ways of doing things, desire for both exploitative and explorative dynamics. Individuals who fulfil these

Ferrari, F. (2019). Does too much love hinder innovation? Family involvement and firms' innovativeness in family-owned Small Medium Enterprises (SMEs). *European Journal of Family Business*, 9(2), 115-127.

ambidextrous roles might face tensions in terms of different kinds of cognitive orientation requested by contradictory activities, such as efficiency-oriented versus variability-increasing tasks (Bonesso, Gerli, Scapolan, 2014; Swart & Kinnie, 2007). Hence, a scarce sense of belonging could easily undermine individual ambidexterity and eventually result in low Process Innovation Enmeshed families show a negative correlation with both the overall innovation propensity level

(-.275) and with Product Innovation (-.43). These findings seem to support previous empirical research, although it was carried out on strategic management rather than innovation propensity specifically. For instance, Vozikis et al. (2013) underline the relationship between family cohesion and preference for family in firm-level decisions. Extremely high levels of cohesion in the family translates to preference for family in the family firm, thus hindering the non-family members' contributions in terms of ideas, knowledge and skills, all issues at the base of new product development. Thus, as suggested by Daspit and colleagues (2017, 17), "if non-family members are indeed less likely to be hired and promoted, the organization is likely to suffer from a lack of specialized skills and diverse resources gained from knowledge the employment and advancement of non-family members".

In this sample, chaotic families show a negative correlation both with Strategic Innovation (-.32) and Process Innovation (-.37). The correlation matrix (see Table 1) also shows a significant correlation (.31) between Strategic and Process Innovation, suggesting the prominent role played by these intertwined features and their effect on firm performance.

Becker (2004) suggested that the chaotically linked system hinders intra-firm coordination and stability due to the lack of synergistic routines that develop in the chaotic context, thus undermining the development of both process and strategic innovation. Moreover, Becker also suggested (2004) that because the internal environment is inconsistent due to the chaotic family structure, the firm suffers from a lack of order, efficiency, and certainty, which may undermine all aspects of firm performance (Chrisman, Memili, & Misra, 2013).

In summary, these empirical findings suggest that unbalanced family situations (enmeshed, chaotic, disengaged) lead to negative outcomes in innovation propensity, thus supporting what is being hypothesized by ongoing theoretical literature (see, for instance, Daspit et al., 2017). Hence, the circumplex model provides rich suggestions on the role played by extreme cohesion and flexibility levels on family firm innovation levels/ innovation propensity (levels).

## Conclusions

The circumplex model seems to be suitable for overcoming the limitations demonstrated by other organizational approaches. This paper suggests that an over-involved/flexible and/or insufficiently involved/flexible family in the firm both lead to negative outcomes in innovation propensity. Therefore, FACES IV is a suitable tool for investigating if and in which conditions family relationships are to the detriment of firm performance, in particular to the innovation propensity. In conclusion this research, by providing empirical evidence supporting the circumplex model's application to family firms, has advanced several streams of research suggested by recent theoretical literature (Daspit et al., 2017).

First, this paper integrates current literature, offering insights into how the family system is a source of family firm heterogeneity, identifying why differences in innovation propensity exist rather than simply highlighting the existence of those differences. Indeed, theoretical models such as Resource Based View, Agency Theory, Stewardship Theory and Behavioural Theories show significant limitations in explaining different family firms' performance in innovation propensity. Drawing on Psychology of the Family literature, this research suggests that these limitations are due to the fact that cited approaches posit a linear relationship between family involvement and innovation propensity. Instead, the circumplex curvilinear model shows a better fit for explaining the family involvement impact on innovation propensity.

A further contribution of this paper is to support a theoretical model which is well-suited for defining the different situations in which family involvement leads to negative or positive outcomes in term of innovation level/ innovation propensity. More specifically, this paper provides evidence that some specific conditions play a role in fostering such innovation: enmeshed and mostly chaotic families show lower innovation propensity levels. In doing so, this paper responds to the call for integrating familyspecific insights into family business studies (Jennings, Breitkreuz, & James, 2014).

Third, previous literature has tended to address the impact of only one or two extreme family structures on family firms (e.g., Michael-Tsabari & Lavee, 2012; Penney & Combs, 2013): in contrast, by applying a curvilinear model, this paper addresses its investigation to *all* family system dimensions, highlighting the specific impact of each dimension on family firm outcomes.

Finally, focusing on organizational innovation propensity, this paper responds to the call for further empirical research which investigates 124

specific firm outcomes by applying the circumplex model, (Daspit et al., 2017). **Practical implications** 

As suggested by Daspit and colleagues (2017), the circumplex model offers a guiding framework that can diagnose the extent to which family systems are balanced and how the effects of balanced or unbalanced family dynamics affect the family firm. In addition, focusing on five different 'innovativeness' dimensions (product, market, strategy, behaviour, process), this paper offers a suitable approach in order to identify the specific weaknesses at the base of negative results in one or more of such dimensions. this approach Consequently. also helps psychologists and, more in general, practitioners in designing the proper interventions (e.g. training activities, or organizational re-design etc.) in order to foster organizational innovation propensity. The complex model applied to innovation thus shows real potential in bridging the academic world and the consulting world, as advocated in this Special Issue (EJFB, is forthcoming).

Limitations and suggestions for future research Beyond contributions and practical implications, this paper also presents several limitations.

Findings from this study could be due to a country bias in the sampling process. In fact, the Italian economy consists of small family-managed firms, with low innovation rate, and low capabilities in human resources management or for improving human and financial capital (Eurostat, 2013). Further research would have to involve samples from different organizational populations.

Moreover, it could be very insightful to investigate the specific innovation propensity level of family firms in extreme unbalanced situations (chaotically disengaged, rigidly enmeshed, etc.): due to its limited dimension (few cases belong to these dysfunctional categories), this sample does not allow for this kind of analysis.

Finally, from a methodological point of view, this research gathered data using the short Italian version of the Olson's FACES questionnaire (Loriedo et al., 2013): hence, these findings should be validated in future research by also collecting data with the original full-lengthversion of the Olson questionnaire.

## References

- Arzubiaga, U., Maseda, A., & Iturralde, T. (2017). Exploratory and exploitative innovation in family businesses: The moderating role of the family firm image and family involvement in top management. Review of Managerial Science, 47(7), 1343.
- Astrachan, J. H., and P. Jaskiewicz (2008). Emotional Returns and Emotional Costs in Privately

Held Family Businesses: Advancing Traditional Business Valuation. Family Business Review 21(2), 139-149.

- Astrachan, J. H., Klein, S. B., & Smyrnios, K. X. (2002). The F-Pec scale of family influence: A proposal for solving the family business definition problem. Family Business Review, 15(1), 45-58.
- Barney J.B. (1991). Firm resources and sustained competitive advantage. Journal of Management, 171, 1, 99-120.
- Beavers W.R., Hampson R.B., (1995). Misurare la competenza famigliare: il modello sistemico di Beavers, tr. it. in Ciclo vitale e dinamiche familiari, a cura di F. Walsh, Franco Angeli, Milano.
- Becker, M. C. (2004). Organizational routines: A review of the literature. Industrial and Corporate Change, 13(4), 643-677.
- Berrone, P., Cruz, C., & Gomez-Mejia, L. R. (2012). Socioemotional Wealth in Family Firms: Theoretical Dimensions, Assessment Approaches, and Agenda for Future Research. Family Business Review, 25(3), 258-279.
- Berrone, P., Cruz, C.C., Gomez-Mejia, L.R., & Larraza Kintana, M. (2010). Socioemotional wealth and corporate response to institutional pressures: Do family-controlled firms pollute less? Administrative Science Quarterly, 55(1), 82-113.
- Block, J. H. (2012). R&D investments in family and founder firms: An agency perspective. Journal of Business Venturing. 27, 248-265.
- Bonesso S., Gerli F., & Scapolan A.. (2014). The individual side of ambidexterity: Do individuals' perceptions match actual behaviors in reconciling the exploration and exploitation trade-off? European Management Journal, 32 392-405
- Bugamelli M., Cannari L., Lotti F. & Magri S., (2012). Il gap innovativo del sistema produttivo italiano: radici e possibili rimedi, in A. Arrighetti A. Ninni (a di) La trasformazione 'silenziosa' cura Cambiamento strutturale e strategie d'impresa nell'industria italiana, Università di Parma, Collana Economia Industriale е di Applicata. http://dspace-

unipr.cineca.it/bitstream/1889/2565/1/La\_trasfor mazione\_silenziosa-A%20Arrighetti\_A%20Ninni.pdf

- Cadieux L., Lorrain J., & Hugron P. (2002). Succession in Women-Owned Family Businesses: A Case Study, Family Business Review, 15, 17.
- Carnes, C.M. & Ireland, R.D. (2013). Familiness and innovation: Resource bundling as the missing link. Entrepreneurship Theory and Practice, 37(6), 1399-1419.
- **Carney M. (2005).** Corporate Governance and Competitive Advantage in Family-Controlled Firms. Entrepreneurship Theory and Practice 29(3), 249-265.
- Cartwright, & D. Zander, A. (1953). Group Dynamics: Research and Theory. Row, Peterson, Evanston.
- Cartwright, D., & Zander, A. (1968). Origins of group dynamics. In D. Cartwright & A. Zander (Eds.), Group dynamics: Theory and research (3rd ed., pp. 3-21). New York: Harper & Row.
- Chen, H.-L., & Hsu, W.-T. (2009). Family ownership, board independence, and R&D investment. Family Business Review, 22, 347-362.
- Chrisman, J.J., Fang, H., Kotlar, J., & De Massis, A. (2014b). A note on family influence and the adoption of discontinuous technologies in family

Ferrari, F. (2019). Does too much love hinder innovation? Family involvement and firms' innovativeness in family-owned Small Medium Enterprises (SMEs). *European Journal of Family Business*, 9(2), 115-127.

F. Ferrari

firms. Journal of Product Innovation Management, 32(3), 384-388.

- Chrisman, J. J., & Patel, P. (2012). Variations in R&D investments of family and non-family firms: Behavioral agency and myopic loss aversion perspectives. Academy of Management Journal, 55, 976-997.
- Chrisman, J. J., Chua, J. H., De Massis, A., Frattini, F., & Wright, M. (2014a). The ability and willingness paradox in family firm innovation. Journal of Product Innovation Management, 32(3), 310-318.
- Chrisman, J. J., Memili, E., & Misra, K. (2013). Nonfamily managers, family firms, and the winner's curse: The influence of noneconomic goals and bounded rationality. Entrepreneurship Theory & Practice, 38(5), 1103-1127.
- Chrisman, J., Chua, J., Pearson, A., & Barnett, T. (2012). Family involvement, family influence, and familycentered noneconomic goals in small firms. Entrepreneurship Theory and Practice, 36, 267-293.
- Chrisman, J.J., Chua, J.H., Litz, R. (2004). Comparing the agency costs of family and nonfamily firms: Conceptual issues and exploratory evidence. Entrepreneurship Theory and Practice, 28, 335-354.
- **Craig, J. B. L., and K. Moores (2006).** A 10 Year Longitudinal Investigation of Strategy, Systems, and Environment on innovation in Family Firms. Family Business Review, 19(1), 1-10.
- **Cruz C., Nordqvist M. (2012).** Entrepreneurial orientation in family firms: a generational perspective. Small Business Economy 38(1), 33-49.
- Curimbaba, F. (2002). The dynamics of women's roles as family business managers. Family Business Review, 15(3), 239-252.
- Daspit, J.J, Madison K., Barnett T., & Long R.G. (2017), The emergence of bifurcation bias from unbalanced families: Examining HR practices in the family firm using Circumplex Theory, in Human Resource Management Review, Special Issue on "The Role of Family Science Theories for Human Resource Management in Family Firms". http://dx.doi.org/10.1016/j.hrmr.2017.05.003
- Davis, J. H., F. D. Schoorman, and L. Donaldson (1997). Toward a Stewardship Theory of Management. Academy of Management Review, 22(1), 20-47.
- Davis, P. S., Harveston, P. D. (1998). The influence of family on the family business succession process: A multi-generational perspective. Entrepreneurship Theory and Practice, 22(3), 31-53.
- De Massis A., Frattini F., Pizzurno E., & Cassia L. (2015). Product Innovation in Family versus NonfamilyFirms: An Exploratory Analysis. Journal of Small Business Management, 53(1), 1-36.
- De Massis, A. (2012), Family Involvement and Procedural Justice Climate Among Nonfamily Managers: The Effects of Affect, Social Identities, Trust, and Risk of Non-Reciprocity. Entrepreneurship Theory and Practice, 36, 6, 1227-1234.
- De Massis, A., Frattini, F., & Lichtenthaler, U. (2013). Research on technological innovation in

family firms: Present debates and future directions. Family Business Review, 26(1), 10-31.

- Diéguez-Soto, J., Manzaneque, M., & Rojo-Ramírez, A. A. (2016). Technological Innovation Inputs, Outputs, and Performance: The Moderating Role of Family Involvement in Management. Family Business Review, 29(3), 327-346.
- Duran, P., Kammerlander N., Van Essen M., & Zellweger T. (2016). Doing more with less: Innovation input and output in family firms. Academy of Management Journal, 59, 4, 1224-1264
- Dyer, Jr. W.G. (2003). The family: The missing variable in organizational research. Entrepreneurship Theory & Practice, 27, 4, 401-416.
- Dyer, W. & Dyer, W. (2009). Putting the family into family business research. Family Business Review, 22, 216-219.
- **Dyer, W. G. (1986).** Cultural change in family firms. San Francisco and London: Jossy-Bass.
- Eddleston, K. A., & Kidwell, R. E. (2012). Parentchild relationships: Planting the seeds of deviant behavior in the family firm. Entrepreneurship Theory and Practice, 36(2), 369-386.
- Eddleston, K., & Kellermanns, F. W. (2007). Destructive and productive family relationships: A stewardship theory perspective. Journal of Business Venturing, 22(4), 545- 565.
- EJFB, Call for Paper for the special Issue "Innovation in family firms: bridging the theory-¬practice gap".
- Ferrari, F. (2017). The tragedy of having a daughter: construction and reproduction of gender stereotype in Italian family SMEs. Reti, Saperi, Linguaggi. Italian Journal of Cognitive Sciences, 1/2017, 133-152.
- Ferrari, F. (2019). In the mother's shadow: exploring power dynamics in family business succession. Gender in Management: An International Journal, 34, 2, 121-139.
- Ferrari, F. (forthcoming). Exploring the side effects of socioemotional wealth. A multilevel analysis approach to the dysfunctional dynamics in family business succession. International Journal of Entrepreneurship and Small Business.
- Gagné, M., & E. L. Deci (2005). Self- Determination Theory and Work Motivation. Journal of Organizational Behavior, 26(4), 331-362.
- Gersick, K. E. (1997). Generation to generation: life cycles of the family business. Boston, A: Harvard Business Press.
- Gomez-Mejia, L. R., K. T. Haynes, M. Nunez-Nickel, K. J. L. Jacobson, & J. Moyano-Fuentes (2007). Socioemotional Wealth and Business Risks in Family- Controlled Firms: Evidence from Spanish Olive Oil Mills. Administrative Science Quarterly, 52(1), 106-137.
- Gudmundson, D., Tower, C. B., & Hartman, E. A. (2003). Innovation in small businesses: Culture and ownership structure do matter. Journal of Developmental Entrepreneurship, 8, 1-17.
- Habbershon T. G. & Williams M. L. (1999). A resource-based framework for assessing the strategic advantages of family firms. Family Business Review, 12, 1, 1-25.
- Habbershon T.G., Williams M. & MacMillan I.C. (2003). A unified systems perspective of family

firm performance. Journal of Business Venturing, 18, 4, 451-465.

- Harvey M., & Evans R. (1995). Life After Succession in the Family Business: Is It Really the End of Problems? Family Business Review, 8, 3, 2-16.
- Haveman H.A., & Khaire M.V. (2004). Survival Beyond Succession? The Contingent Impact of Founder Succession on Organizational Failure. Journal of Business Venturing, 19, 437-63.
- Henard, D.H. & Szymanski, D.M. (2001). Why some new products are more successful than others. Journal of Marketing Research, 38, 3, 362-75.
- Ichniowski, C., & Prennushi. G. (1997). The effect of human resource management practices on productivity. American Economical Revue, 87(3) 291-313.,
- Ichniowski, C., & Shaw. K. (1999). The effects of human resource management systems on economic performance: An international comparison of U.S. and Japanese plants. Management Science, 45(5) 704-721.
- Ichniowski, C., Kochan T., Levine D., Olson C., & G. Strauss. (1996). What works at work: Overview and assessment. Industrial Relations, 35(3) 299-333.
- James, A., Jennings, J., & Breitkreuz, R. (2012). Worlds apart? Rebridging the distance between family science and family business research. Family Business Review, 25, 87-108.
- Jennings, J. E., Breitkreuz, R. S., & James, A. E. (2014). Theories from family science: A review and roadmap for family business research. In L. Melin, M. Nordqvist, & P. Sharma (Eds.), SAGE Handbook of Family Business. London: Sage.
- Kellermanns, F., Eddleston, K., Sarathy, R., & Murphy, F. (2012). Innovativeness in family firms: a family influence perspective. Small Business Economics, 38(1), 85-101.
- Kets de Vries, M. F. R. (1996). The dynamics of family controlled firms: The good and the bad news. In C. E. Aronoff, J. H. Astrachan & J. L. Ward (Eds), Family business sourcebook, vol 1 (pp. 312-323). Marietta, GA: Business Owner Resources.
- Kim, H., Kim, H., & Lee, P. M. (2008). Ownership structure and the relationship between financial slack and R&D investments: Evidence from Korean firms. Organization Science, 19, 404-418.
- Klein, S. B., Astrachan, J. H., & Smyrnios, K. X. (2005). The F-PEC scale of family influence: Construct validation, and further implication for theory. Entrepreneurship Theory and Practice, 29(3), 321-339.
- Kraatz, M., & Block, E. (2008). Organizational implications of institutional pluralism. In R. Greenwood, C. Oliver, R. Suddaby,K.Sahlin-Andersson (Ed.), Handbook of organizational institutionalism: 243-275. London: Sage Publication.
- Labaki, R. (2011). The Nova Group case study: Family dynamics in a multigenerational French family business. International Journal of Management Cases, 13(1), 27-42.
- Lansberg, I. (1999). Succeeding generations: Realizing the dream of families in business. Boston, MA: Harvard Business School Press.
- Lee, J. (2006). Impact of family relationships on attitudes of the second generation in family business. Family Business Review, 19(3), 175-191.

- Llach, J., & Nordqvist, M. (2010). Innovation in family and non-family businesses: A resource perspective. International Journal of Entrepreneurial Venturing, 2: 381-399.
- Madanoglu, M., Altinay, L., & Wang, X. (2016). Disentangling the effect of family involvement on innovativeness and risk taking: The role of decentralization. Journal of Business Research, 69, 1796-1800.
- Meyer, J. P., & Herscovitch, L. (2001). Commitment in the workplace: Toward a general model. Human Resource Management Review, 11(3), 299-326.
- Michael-Tsabari N., Lavee Y. (2012). Too Close And Too Rigid: Applying The Circumplex Model Of Family Systems To First-Generation Family Firms. Journal Of Marital And Family Therapy, 38, 1, 105-116.
- Minichilli A., Corbetta G., MacMillan I.C. (2010). Top management teams in family-controlled companies: 'Familiness', 'faultlines', and their impact on financial performance. Journal of Management Studies, 47(2), 205-222.
- Morrison, E. W. (1994). Role definitions and organizational citizenship behavior: The importance of the employee's perspective. Academy of Management Journal, 37(6), 1543-1567.
- Nosé, L., Korunka, C., Frank, H., & Danes, S. M. (2015). Decreasing the effects of relationship conflict on family businesses. The moderating role of family climate. Journal of Family Issues, doi: 10.1177/0192513X15573869.
- **Ortega, J (2001).** Job Rotation as a Learning Mechanism. Management Science, 47, 10, 1361-1370.
- Padilla-Meléndez A., Dieguez-Soto J., Garrido-Moreno. A., (2015). Empirical research on Innovation in Family Business: literature review and proposal of an integrative framework. Revue of business. Management, 17, 56, 1064-1089.
- Pellegrino B., Zingales L. (2014), Diagnosing the Italian Disease, Chicago Booth Working Paper. http://faculty.chicagobooth.edu/luigi.zingales/pa pers/research/Diagnosing.pdf
- Penney, C.R. & Combs, J.G. (2013). Insights from Family Science: The Case of Innovation. Entrepreneurship Theory and Practice, 37, 6, 1421-1427.
- Ryan, R. M., & Deci E. L. (2000). Self- Determination Theory and the Facilitation of Intrinsic Motivation, Social Development, and Well-Being. American Psychologist, 55(1), 68-78.
- Salvato, C., Minichilli, A., & Piccarreta, R. (2012). Faster route to the CEO suite: Nepotism or managerial proficiency? Family Business Review, 25(2), 206-224.
- Scabini E., lafrate R., (2003), Psicologia dei legami familiari, Carocci, Roma.
- Schulze W.S., Lubatkin M.H., Dino R.N. & Buchholtz A.K. (2001). Agency relationships in family firms: Theory and evidence. Organization Science, 12, 2, 99-116.
- Schulze, W.S., Lubatkin, M.H., & Dino, R.N. (2003). Toward a theory of agency and altruism in family firms. Journal of Business Venturing, 18, 473-490.
- Sciascia, S., Clinton, E., Nason, R.S, James, A.E & Rivera-Algarin, J.O (2013). Family Communication and Innovativeness in Family Firms. Family Relations, 62, 3, 429- 442.

Ferrari, F. (2019). Does too much love hinder innovation? Family involvement and firms' innovativeness in family-owned Small Medium Enterprises (SMEs). *European Journal of Family Business*, 9(2), 115-127.

- Sharma P. & Manikutty, S. (2005). Strategic Divestments in Family Firms: Role of Family Structure and Community Culture. Entrepreneurship: Theory & Practice, 29, 3, 293-311.
- Sharma, P., & Irving, P. G. (2005). Four bases of family business successor commitment: Antecedents and consequences. Entrepreneurship Theory & Practice, 29(1), 13-33.
- Sharma, P., Chrisman, J. J., & Chua, J. H. (2003a). Predictors of satisfaction with the succession process in family firms. Journal of Business Venturing, 18(5), 667-687.
- Sharma, P., Chrisman, J. J., & Chua, J. H. (2003b). Succession planning as planned behavior: Some empirical results. Family Business Review, 16(1), 1-15.
- Sharma, P., Chrisman, J. J., Pablo, A., & Chua, J. H. (2001). Determinants of initial satisfaction with the succession process in family firms: A conceptual model. Entrepreneurship Theory and Practice, 25(3), 1-19.
- Shepherd, D.A. & Haynie, M.A. (2009). Family business, identity conflict, and an expedited entrepreneurial process: A process of resolving identity conflict. Entrepreneurship Theory and Practice, 33(6), 1245-1264.
- Sirmon D.G. & Hitt M.A. (2003). Managing resources: Linking unique resources, management, and wealth creation in family firms. Entrepreneurship Theory and Practice, 27, 4, 339-358.
- Steier, L. (2001). Family firms, plural forms of governance, and the evolving role of trust. Family Business Review, 14, 353-367.
- Stryker S. (1968). Identity Salience and Role Performance. Journal of Marriage and the Family, 4, 558-64.
- Sundaramurthy, C. (2008). Sustaining trust within family businesses. Family Business Review, 21, 89-102.
- Swart, J., & Kinnie, N. (2007). Simultaneity of learning orientations in a marketing agency. Management Learning, 38(3), 337-357.
- Tagiuri, R., & Davis, J. A. (1996). Bivalent attributes of the family firm. Family Business Review, 9(2), 199-208.
- Tajfel H., Turner J.C, (1986) The social identity theory of intergroup behaviour, in S. Worchel, W.G. Austin (Eds.), Psychology of intergroup relations, pp. 7-24, Nelson-Hall Publishers, Chicago
- Tidd, J., J. Bessant, and K. Pavitt (2001). Managing Innovation: Integrating Technological, Market and Organizational Change. Chichester, UK: John Wiley & Sons.
- Trott, P. (2008). Innovation Management and New Product Development. Harlow, UK: Financial Times/Prentice Hall.
- Uhlaner L.M., (2006), Business Family as Team: Underlying Force for Sustained Competitive Advantage, in Poutziouris P. (edited by), Handbook of Family Business Research, Edward Elgar Publishing.
- Vera C. F. & Dean, M. A. (2005). An Examination of the Challenges Daughters Face in Family Business Succession. Family Business Review, 18, 321-345.
- Vozikis, G. S., Weaver, K. M., & Liguori, E. W. (2013). Do family cohesion and family member

skill evaluation affect family business internal or external hiring decisions? Journal of Management Policy and Practice, 14(1), 75-89.

- Wagner, J., & Hollenbeck, J. (1995). Management of organizational behavior (2nd ed.). Englewood Cliffs, NJ: Prentice-Hall.
- Ward J. L. (1987). Keeping the family business healthy: How to plan for continuing growth, profitability, and family leadership. Jossey-Brass, San Francisco.
- Wright, M., & Kellermanns, F. W. (2011). Family firms: A research agenda and publication guide. Journal of Family Business Strategy, 2(4), 187-198.
- Zellweger T. M. & Astrachan, J. H. (2008). On the emotional value of owning a firm. Family Business Review, 21 (4), 347-363.
- Zellweger T., & Sieger, P. (2012). Entrepreneurial orientation in long-lived firms. Small Business Economics, 8(1), 67-84.
- Zody, Z., Sprenkle, D., MacDermid, S., & Schrank, H. (2006). Boundaries and the functioning of family and business systems. Journal of Family and Economic Issues, 27(2), 185-206.

Ferrari, F. (2019). Does too much love hinder innovation? Family involvement and firms' innovativeness in family-owned Small Medium Enterprises (SMEs). *European Journal of Family Business*, 9(2), 115-127.



EUROPEAN JOURNAL OF FAMILY BUSINESS

http://www.revistas.uma.es/index.php/ejfb



# Entrepreneurial orientation and product innovation. The moderating role of family involvement in management.

Taras Fredyna<sup>a\*</sup>, Daniel Ruíz-Palomo<sup>b</sup>, Julio Dieguez<sup>b</sup>

<sup>a</sup>Graduated in Management and Business and Law, University of Málaga (Spain) <sup>b</sup>Department of Accounting and Finance, University of Málaga (Spain)

Received 12 December 2018; accepted 19 December 2019

#### JEL CLASSIFICATION L26, O32

KEYWORDS Family involvement; Entrepreneurial Orientation; Product Innovation; Radical Innovation; Incremental Innovation

#### CÓDIGOS JEL L26, O32

PALABRAS CLAVE Implicación familiar; Orientación Emprendedora; Innovación en Productos; Innovación Radical; Innovación Incremental Abstract In this current research it has been examined, firstly, the Entrepreneurial orientation in relation with Product innovation, Incremental innovation and Radical innovation, and, secondly, the moderating effect of Family involvement in the management of companies in the relationship between Entrepreneurial orientation and Product innovation, Incremental innovation and Radical innovation. Using a sample of 634 Spanish family firms, the results found conclude that the Entrepreneurial orientation has a positive effect on Product innovation, Incremental innovation, and Radical innovation, and moreover, they reveal that the Family involvement in the management of companies has a moderating effect on these relations, witnessing that family firms with higher scale of Family involvement in the management of the companies reduces the effect of Entrepreneurial orientation on Product innovation, Incremental innovation and finally, Radical innovation.

## Orientación emprendedora e innovación en productos. El efecto moderador de la implicación familiar en la gestión.

**Resumen** En el presente trabajo se ha analizado, en primer lugar, la relación entre la Orientación emprendedora y la Innovación en productos, Innovación incremental e Innovación radical, y en segundo lugar, el efecto moderador que tiene la Implicación familiar en la gestión de la empresa sobre la relación entre la Orientación emprendedora y la Innovación en productos, Innovación incremental e Innovación radical. Para tal análisis se ha utilizado una muestra de 634 empresas familiares españolas, y los resultados obtenidos han demostrado que la Orientación emprendedora tiene un efecto positivo sobre la Innovación en productos, Innovación incremental e Innovación radical, y además, también han demostrado que la Implicación familiar en la gestión de la empresa tiene un efecto moderador en dichas relaciones, de tal forma que las empresas familiares con mayor Implicación familiar en la gestión reducen el efecto de la Orientación emprendedora sobre la Innovación en productos, Innovación incremental e Innovación radical.

http://dx.doi.org/10.24310/ejfbejfb.v9i2.5392

Copyright 2019: Taras Fredyna, Daniel Ruíz-Palomo, Julio Dieguez.

European Journal of Family Business is an open access journal published in Malaga by UMA Editorial. ISSN 2444-8788 ISSN-e 2444-877X This work is licensed under a Creative Commons Atribution-NonCommercial-ShareAlike 4.0 International License (CC BY-NC-SA 4.0).

## Introduction

The Entrepreneurial Orientation, it is understood as a strategic process whereby companies can identify new opportunities and carry out entrepreneurial actions to later take advantage of them (Dess and Lumpkin, 2005), is an important figure in the field of entrepreneurship, due to its Influence on the results, value, and growth of a company, it has been the subject of many studies in recent years. Most of these researches are based on a series of conceptual bases that, constructed on the definition of Miller's Entrepreneurial orientation (1983), were established by Lumpkin and Dess (1996), which are as follows:

- Entrepreneurial orientation includes five dimensions, which are innovation, assumption of risk, proactivity, competitive aggressiveness and autonomy.

- The dimensions of the entrepreneurial orientation are independent but related.

- The direct relationship between the entrepreneurial orientation and the performance may differ because of others mediating or moderating variables.

In the field of entrepreneurship research, there are several studies that have analysed the different relationships of it with other variables, such as business result (Smart and Conant, 1994; George, Wood and Khan, 2001; Rauch et al., 2009), or the growth of a company (Wiklund and Shepherd, 2005; Moreno and Casillas, 2008; Casilla and Moreno, 2010). Likewise, it has been studied the effect and importance of different variables on entrepreneurial orientation, such as the organizational structure (Covin and Slevin, 1988; Green, Covin and Slevin, 2008), degree of dynamism in the sector (Lumpkin and Dess, 2001; Wiklund and Shepherd, 2005), technological intensity in industry (Covin and Slevin, 1989), social capital (Stam and Elfring, 2008), as well as innovation (Elenurm, Ennulo and Laar, 2007; Rauch et al., 2009; Fernández-Mesa et al., 2012), highlighting not only the positive association between entrepreneurship and innovation (Fernández-Mesa et al., 2012), but also its importance on business performance (Newey and Zahra, 2009; Baker and Sinkula, 2009).

Accordingly, investigations indicate that enterprising companies, unlike other conservative companies, innovate more regularly and frequently, while taking risks, which are considered in their product market as strategies (Miller and Friesen, 1982). The concept of Innovation can be understood from different points of view; hence, a company can innovate by: (i) changes in its working methods; (ii) in the use of productive factors or (iii) in the ways of carrying out production. Following the Oslo Manual (2005), there are four types of innovations that cover a wide range of changes which can take place in the activities of a company: (i) Product innovation, (ii) Process innovation, (iii) Organizational innovation and (iv) Marketing innovation.

Bearing in mind the diverse types of innovation described above, most of the researches, as well as in this article, are about Technological Innovation (a concept closely related to Product innovation, according to the Oslo Manual (2005), which Souitaris (2003) justifies as any change in things, products or services offered by an organization (Product innovation) and any change in the way these are created and delivered (Process innovation). The interest in this kind of innovation is mainly due to its major influence industrial competitiveness and national on is development since it an important determinant of sustained performance (Blundell et al., 1999). Additionally, we can differentiate two other forms of innovation: Radical innovation and Incremental innovation (Clausen and Pohjola, 2013). Radical innovations are defined as innovations in products which are new and unknown to the market in which the while company operates: incremental innovations are new product innovations to the company but, not new to the market.

Literature reveals that several researches support the existence of a positive relationship between entrepreneurial orientation and innovation (Nasution et al., 2010). In the same way, there are other studies, which confirm a positive influence of entrepreneurial orientation on product innovation (Boso et al., 2013; Avlonitisa and Salavoub, 2007; Zhou et al., 2005) and on radical innovation (Zhou, Yim, and Tse, 2005). However, to the best of our knowledge, entrepreneurial orientation literature has not analysed the relation mentioned above. distinguishing between radical and incremental innovation. In order to attempts to fill this gap, this study analyses not only the relationship between entrepreneurial orientation and product innovation, but also distinguishes between radical and incremental innovation. We consider, in accordance with Sorescu and Spanjol (2008), that the influence of entrepreneurial orientation on such type of product innovation may differs due to the different levels of efforts which are needed to its implementation (risk-taking, investments or organizational and management capacities) and its impacts in the firm performance (growth, ROA,...).

Furthermore, the context of Family Firm, which plays a very important role in the world economy (Aranoff and Ward, 1995; La Porta et al., 1999; Neubauer and Lank, 1998), presents a special interest in the literature of entrepreneurial orientation (Zahra et al., 2004; Steier, 2007; Steier, 2009) and innovation (Craig and Moores, 2006; Pittino and Visintin, 2009).

Thus, within the field of entrepreneurship research, family firms present an unique and exceptional context. The singularity of family firms is manifested through its structure of resources and capacities (Sirmon and Hitt, 2003; Dyer 2006), which can ease or restrict its entrepreneurial activities (Habbershon, Williams and MacMillan, 2003; Howorth, 2007; Nordqvist, Habbershon and Melin, 2008). The studies carried out in this field show that there is a dual relationship between family firms and the entrepreneurial orientation (Nordqvist, Habbershon, and Melin, 2008). On the one hand, certain characteristics of family firms help to develop its entrepreneurship (Aldrich and Cliff, 2003), such as its culture (Hall, Melin, and Nordqvist, 2001; Nordqvist et al., 2008; Zahra et al., 2004) or its centralized structure, which also influences the proactive and innovative conduct of the company (Salvato, 2004). However, there are other factors of these types of companies, which have а reducing effect on its entrepreneurship, such as their greater resistance to change or their aversion to risk (Naldi et al., 2007; Zahra, 2005). Such duality, as well as its uniqueness, makes the relationships between entrepreneurial orientation and family firms may become an interesting field of study.

Likewise, in the field of Innovation, there is also much interest in family firms, as well as its relationship with innovation. Traditionally, family firms are considered more conservative (Sharma, Chrisman, and Chua, 1997), less keen to change (Kets de Vries, 1993; Ward, 1997), more risk-averse (Naldi et al., 2007), and with greater difficulty to get access to capital markets (Kets de Vries, 1993), therefore, some studies indicate that the relationship between family involvement in the company and innovation is negative (Chen and Hsu, 2009). Nevertheless, family firms should have incentives to innovate, since that innovation creates wealth and opens up new business opportunities. Hence, it is coherent with their survival instinct and concern for their long-term continuity (Sirmon et al., 2008). In this sense, other studies confirm the existence of a positive relationship between the family involvement in the company and innovation (Zahra 2005; Margaret, 2008; Casillas and Moreno. 2010). In addition, familv involvement in a company influences the process of making strategic decisions (Miller, Le Breton-Miller, and Lester, 2011), and therefore, has an impact on decisions related to the innovation. Furthermore, studies on innovation in family firms are usually focused on product (De Massis et al., 2013), process (Classen et al., 2014) or organizational (Madrid-Guijarro, García and Van Auken, 2009) innovation.

Despite the importance of family management to act as a driver of the willingness and ability to impact on product innovation, as far as we know, no one has empirically investigated the interaction effect of family management and entrepreneurial innovation on product innovation. Thus, we bridge an important gap in literature studying the moderating effect of family management in the entrepreneurial orientation-product innovation interplay. In addition, we have not only analysed the moderating effect in the mentioned relationship, but also we distinguish between radical and incremental innovation.

Therefore, we propose the following research questions: Has entrepreneurial orientation a positive impact on product innovation? Does this relationship differ if we distinguish between radical and incremental innovation? And, finally, are these relationships affected by family management?

Bearing in mind the above gaps identified and research questions, the aim of this article is to study the influence of entrepreneurial orientation on product innovation, distinguishing between incremental and radical innovation. Furthermore, this work aims to analyse the moderating effect of Family Involvement in the management of the company in the relationship Entrepreneurial Orientation between and Product Innovation. To this end, we integrate our arguments using Resource Based View as the theoretical background of reference.

For those purposes, an empirical study has been developed through a regression model, using data obtained from a survey, which comprises a total of 634 business managers located in the Region of Murcia (Spain).

The obtained results indicate that family management decreases the positive effect of entrepreneurial orientation on product innovation, radical innovation and incremental innovation. These results contribute to the family business literature in several ways. Firstly, whereas previous studies have mainly analysed the relationship between entrepreneurial orientation and product innovation, we further examine this relationship distinguishing between radical innovation and incremental innovation. In this way, we develop fine-grained understanding а more of entrepreneurial orientation-product innovation relationship. Secondly, we also study the moderating effect of family management in the relationship between entrepreneurial orientation and product innovation, radical innovation and incremental innovation, showing that family management has a negative moderating effect. Thus, this paper contributes to an enhanced understanding of the moderating role of family management in the obtaining of product innovation, giving a more nuanced picture of

Fredyna, T. Ruíz-Palomo, D. Dieguez, J. (2019). Entrepreneurial orientation and product innovation. The moderating role of family involvement in management. *European Journal of Family Business*, 9(2), 128-145.

entrepreneurial orientation-product innovation relationship.

The current study is divided into several sections. The next section offers the theoretical bases of the entrepreneurial orientation, product innovation and family concepts, as well as the existing relationships between them will be section explained. Then, in three the Methodology used will be described, showing the sample, the data and main the variables, followed by the presentation of the findings obtained. Finally, in section forth the main findings of the study will be exposed and commented.

## Theorical framework an Hypothesis

## The Entrepreneurial Orientation and Product Innovation

Miller (1983, pp.771) has been the pioneer defining the concept of entrepreneurial orientation, according to which entrepreneurial companies are "those that are geared towards innovation in the product - market field by carrying out risky initiatives, and which are the first to develop innovations in a proactive way in an attempt to defeat their competitors". Miller not only defined the concept of entrepreneurial orientation but also established three distinguishing dimensions of it, such as (i) innovativeness, (ii) risk-taking (iii) proactiveness. Later, Lumpkin and Dess (1996) suggested two more dimensions besides the three previous ones proposed by Miller (1983), such as (iv) competitive aggressiveness and (v) autonomy, establishing a total of five dimensions, but this conceptualization has not been widely adopted (Wales et al.. 2013), and it has been argued that the original three-dimension of entrepreneurial orientation is best from a conceptual point of view (George and Marino, 2011). Therefore, focused on Miller's (1983)original conceptualization, researchers agree on the three dimensions of entrepreneurial orientation (Covin and Wales 2018; Rauch et al. 2009; Wales et al. 2013): innovation, risk-taking, and proactiveness. Moreover, Covin and Slevin (1989, pp.79) have identified these three dimensions as "a basic, unidimensional strategic orientation" which implies that only companies that exhibit high levels of all three dimension should be regarded as entrepreneurial. Furthermore, these three characteristics were positioned by Miller (1983) as the heart of entrepreneurial orientation and are often combined to create a higher-order indicator of firm-level entrepreneurship (Rauch et al., 2009). Therefore, defined entrepreneurial we orientation as a firm-level construct of these three dimensions where innovativeness, risktaking, and proactiveness, as it has been used in literature (Wiklund and Shepherd, 2005; Rauch et al., 2009; Wales et al., 2013; Wales et al., 2018).

Thus, dimension of innovativeness а is understood, according to Lumpkin and Dess (1996, pp. 142), as "a firm's tendency to engage and support ideas, in new novelty, experimentation, and creative processes that may result in new products, services, or technological processes". As regards the dimension of proactiveness, according to Hughes and Morgan (2007), it represents a future perspective where companies try to develop new products or improvements in them, anticipating the changes and opportunities that appear in the business setting, promoting changes in current tactics and detecting future market trends. And also, by its own, the risk-taking dimension is understood as "the degree to which managers are willing to make large and risky resource commitments - i.e., those which have a reasonable chance of costly failures'' (Miller and Friesen, 1978, p. 932). Moreover, Wiklund and Shepher (2005), indicate that this dimension implies the allocation of significant resources by the company to exploit opportunities or carry out strategies whose results are uncertain in unfamiliar situations.

According to the aforementioned, it can be indicated that the entrepreneurial orientation of a company is not an action carried out in a timely or unitary way, but rather it is an activity that implies a continuous strategic performance over time (Covin and Slevin, 1991), which translates into a generating process of the entrepreneurial strategy that decision-makers use to disseminate organizational purposes,

maintaining their vision and creating sustainable competitive advantages.

Regarding the concept of innovation, Schumpeter (1934) demonstrated that, this takes place when a new good or change in its quality is introduced, a new production method, the opening to a new market, and the conquest of a new source of

supply of raw materials or the creation of a new company in any industry. According to what he exposed, the development of the economy is due to innovation, understood as the dynamic process through which updated technologies replace the old ones, a process called "creative destruction". According to this point of view, economic growth takes place because of the process of destruction in which the old structure of the industry - its products, its processes, or its organization changes continuously due to innovation (Link, 1980). In this sense, innovative activity is the main source of innovation and economic progress (Nelson, 1991).

In addition to this, the Oslo Manual (2005, pp. 46) defines innovation as "the implementation of a new or significantly improved product (good or service), or process, a new marketing method, or a new organizational method in business practices, workplace organization or external relations".

Both Schumpeter (1934) and the Oslo Manual (2005) differentiate several types of innovation, namely: (i) Product innovation, (ii) Process innovation, (iii) Organizational innovation and (iv) Marketing innovation. In relation with the different types of innovation indicated before, the present study deals with Product Innovation, not only taking into account its popularity in most of the works (Souitaris, 2003), but also because it is a relevant aspect of sustained performance (Blundell et al., 1999), as well as due to its importance for the survival of a company (Dyer and Song, 1998), its profitability (Ali et al., 1993), its growth and its expansion into new areas (Danneels and Kleinschmidt, 2001). Thus, a company is innovative in products when it provides a new or significantly improved product (good or service) regarding to its characteristics or previous uses (Oslo Manual, 2005; Laursen and Salter, 2006).

Yet, the aim of innovation can be based on a new knowledge or on the reconfiguration, in a new way, of the existing one (Schumpeter, 1934; Drucker, 1985). From an economic point of view, a product, service or production process may be considered as an innovation, but it does not have to be something new in a strict sense, but new in the market where it wants to be introduced (Koellinger, 2008). In this sense, we can differentiate other two concepts within innovation, such as Incremental innovation and Radical Innovation, depending on the degree of novelty of it (Schumpeter, 1934; Oslo Manual, 2005; Anderson and Tushman, 1990), so that the radical innovation consists in the contribution of goods and services that differ significantly in their characteristics or intended uses in relation to the products previously produced by the company and, the incremental innovation, which implies minor changes in the technical specifications of the products already existing in the company. It should also be noted that incremental and radical innovations bring distinct levels of risk and therefore, require organizational different and management capacities. Thus, radical innovations derive from the exploration of new capacities searching greater variations and novelties (March 1991). Companies introducing radical innovations need to substantially change their ways of operating by entering unknown markets and/or introducing new products based on technologies that are new to them. This type of innovation can give several results: on the one hand, it can lead to the destruction of competition (Menguc and Auh, 2010), and on the other hand, it can lead to cannibalization of current products and even changes in competition rules (Hurmelinna -Laukkanen, Sainio, and Jauhiainen, 2008). On top of that, incremental innovation is the result of the exploitation of current capacities and in search of continuous improvements (March, 1991). The research of this type of innovation is one of the simplest tasks (Bessant et al, 2010) since the decisions on that subject are taken within the framework of a trajectory or an established technological paradigm (Dosi, 1982). Therefore, radical innovation, as opposed to incremental innovation, represents a higher risk strategy, but also a higher result (Bessant, Birkinshaw and Delbridge 2004; Sorescu and Spanjol, 2008).

Considering both concepts, entrepreneurial orientation and innovation, these are considered crucial factors for the growth and economic performance of companies. Schumpeter (1961) already warned of the importance of innovation and entrepreneurship, as they are central figures of economic growth and development. Based on his work, recent researches in the field of strategic management and evolutionary economics sustain that companies must be entrepreneurial and put innovation at the

Fredyna, T. Ruíz-Palomo, D. Dieguez, J. (2019). Entrepreneurial orientation and product innovation. The moderating role of family involvement in management. *European Journal of Family Business*, 9(2), 128-145.

forefront of competition strategy (Teece, 2007). The main reason is because new products are the basis of the company's performance and profitability (Teece, 2007) and such innovations are fundamental for adapting to change market conditions (Nijssen, Hillebrand and Vermeulen, 2005; Bessant et al., 2005). Actually, according to Miles and Snow (1978) ideas, those companies that adopt an innovative orientation can respond quickly to changes in business setting and have a greater capacity to find and exploit new products and market opportunities. Furthermore, entrepreneurial orientation is an important factor not only for the survival of the company, but also for the improvement of its short and long-term results (Wiklund and Shepherd, 2005), since it implies "entrepreneurial strategy-making processes that key decision makers use to enact their firm's organizational purpose, sustain its vision, and create competitive advantage(s)" (Rauch et al., 2009, pp. 763). Moreover it involves a willingness to innovate to rejuvenate market offerings, take risks to try competitors toward new marketplace opportunities (Covin and Slevin, 1991). Thus, the implementation and integration of the strategy of entrepreneurial orientation, not only allows increasing the skills of the company to generate knowledge and provide solutions to fulfil the needs of current and potential consumers (Workman, 1993; Gatignon and Xuereb, 1997), but it also reduces their aversion to risk and thus, favours the development of knowledge generation mechanism and improves processes in product innovation (Miller, 1983; Lumpkin and Dess, 1996).

Considering the abovementioned, it would be considered that there are a positive relationship between entrepreneurial orientation and product innovation. The fundamental premise of Resource Based View theory is that firms' resources and capabilities are those that determining firm's capacity to innovate. Firms' resources can be tangible or intangible (Hall, 1992; Amit and Schoemaker, 1993). Tangible resources include capital, access to capital and consist location. Intangible resources of knowledge, reputation, skills, entrepreneurial orientation, etc. (Runyan et al., 2006). The considerations of entrepreneurial orientation like a strategy or process contribute to the creation

of opportunities and advantages, convert it into intangible resource and an important dimension of Resource Based View theory, which have impact in growth (Ferreira et al., 2011) and innovation (Poazi et al., 2017) of firms. From the strategic point of view, intangible resources may be more important, as soon as they are able to requirements for encourage the produce sustainable advantages which are more valuable, rare and inimitated by competitors (Bettis and Hitt, 1995; Barney, 1991), what mean that high stock of intangible resources, including entrepreneurial orientation, increases the probability of firm innovation (Nonaka, 1994).

In fact, Miller and Frisen (1982), just like more recently studies have been confirmed (Nasution et al., 2010), argued that more enterprising companies innovate more frequently.

In the same way, other authors confirm the positive relationship between entrepreneurial orientation and product innovation (Zhou et al., 2005), as well as with radical innovation (Zhou, Yim, and Tse, 2005). Similarly, Salavou and Lioukas (2003), in their study of Greek SMEs on the strategic sources of radical innovations in products, have concluded that entrepreneurial orientation has a positive effect on product innovations. In addition, more recent study of Boso et al. (2013), also confirmed the positive relationship between entrepreneurial orientation and product innovation. Likewise Avlonitisa and Salavoub (2007) showed that entrepreneurial companies, being more proactive and risky, develop and introduce more innovative products. Adding more relevance to the foregoing, the following hypotheses are proposed:

H11: Firm Entrepreneurial Orientation has a positive effect on Product Innovation.

H12: Firm Entrepreneurial Orientation has a positive effect on Incremental innovation.

H13: Firm Entrepreneurial Orientation has a positive effect on Radical Innovation.

## Family management

The qualification of Family Firms exists due to the determination of a vision and control mechanisms of the company by a family, and its contribution to the creation of resources and unique capabilities within them (Sharma, 2004; Chrisman, Chua and Linz, 2003). Astrachan, Klein and Smyrnios (2002) developed a system of scales that differentiate the degree of family involvement into a company, these are:

- The experience, which measures the involvement of a family through the number of members and generations of the family with an active participation into the company;

- Culture, which measures the involvement of a family in the values of the company, as well as the commitment to it;

- Control, which measures the influence of a family on the ownership, management, and governance of the company.

Villalonga and Amit (2006) also differentiate three ways in which families can be involved in a company, this is through management, ownership and control.

In Europe, the national associations of family businesses have adopted as a formal definition, the proposal made by the European Family Business Group (GEEF) in 2008 considers a family business to be one in which:

- Most of the votes are owned by the person or persons of the family that founded or acquired the company;

- Most votes can be direct or indirect;

- At least one representative of the family participates in the management or governance of the company;

- Listed companies are considered a family business if the person who founded or acquired the company, or their family members or descendants own 25% of the voting rights to which the share capital is entitled.

In the same way, it can be indicated that the family involvement in a company can be carried out in different ways, and in the present work it is understood by the family management, the participation of the family members in the management activities of a family firm, measured by the family participation in the management of the organization and the structure of the company (Zahra 2005).

In the context of entrepreneurship, family firms are an unique and relevant context to analyse, either by the configuration of resources and capabilities they have and their relationship with entrepreneurial behaviour (Sirmon and Hitt, 2003; Dyer, 2006; Eddleston, Kellermanns, and Sarathy, 2008) or the relevant effect of family management in the process of decision making (Nordqvist, Habbershon, and Melin, 2008).

In general, family-managed firms are characterized by their greater ability to develop an entrepreneurial performance, since their survival depends on their ability to enter new markets in which they can offer innovative products and services (Zahra, Hayton, and Salvato, 2004). This is due to a series of characteristics present in this type of company (Casillas and Moreno, 2010), such as: (i) its greater long-term orientation, which allows it to develop better long-term business strategies (Ward, 1997), including innovation strategies, since in this situation, they are more likely to support innovation as a source of growth and wealth (Zahra, Hayton and Salvato, 2004) and as а survival mechanism to protect their competitiveness over time; (ii) its greater degree of centralization, which makes the process of decision-making faster, and therefore, the implementation and development of innovation too; (iii) and the strength of the interests between the owners and the managers, since the greater degree of concentration between ownership and management, the innovation projects proposed by the family are more defended, and consequently, the allocation of the resources for its development is faster and more efficient, in addition to having greater control over the measures taken by management (Donckels and Fröhlich, 1991).

However, the resources available to the company are small, and considering their aversion to debt at the time of making capital investments (Galve Gorriz and Salas Fumas, 1996), family-managed firms have difficulties to develop proactive strategies. It may be noted that not only risk aversion affects the resources available to the family business, but also capital restrictions, which jointly limit the possibility of initiating costly innovation projects. For fear of losing control in decision making, family managed firms are slightly inclined to access capital markets (Kets de Vries 1993), what causes that they are restricted regarding their financial resources and thus, their opportunities to finance innovation activities as well as proactive projects.

Furthermore, family managed firms have a greater aversion to risk as a result of the superposition of the objective of survival of the company, the fortune and social reputation of the family, against the profitability objective (Zahra 2005, Gómez-Mejía et al., 2007), and its more conservative character of business assets, which represents the personal wealth of the family (Gómez-Mejía et al., 2007). Therefore, its aversion to indebtedness (Galve Gorriz and Salas Fumas, 1996) which reduces the availability of financial resources, that causes firstly, family managed firms have limited financial resources, and secondly, aim for safer projects (Wright et al., 2002).

Fredyna, T. Ruíz-Palomo, D. Dieguez, J. (2019). Entrepreneurial orientation and product innovation. The moderating role of family involvement in management. *European Journal of Family Business*, 9(2), 128-145.
The Resource Based View theory suggests that resources and capabilities are drivers in innovation of firms (Poazi et al., 2017). Rather, these resources must also be managed effectively and appropriately (Hitt, Ireland, and Hoskisson, 2001) in order to produce value and archive the proposed results. As it is said before, family managed firms rarely have all of the needed resources, particularly financial resources, since these firms avoid sharing equity with nonfamily members and also, they have some limitation to manage them, due to the prioritization of their particular goals, the retaining of interfamily relatedness and the superposition of the objective of survival of the company. This causes a loss of efficiency of the process of strategic decision-making, including those related to innovation (Le Breton-Miller and Miller, 2009), and a greater difficulty in developing proactive projects, what, as we have suggested, decreases the positive relationship between entrepreneurial orientation and product innovation (Choi et al., 2015).

In order to emphasise aforementioned and considering that entrepreneurial orientation is a resource-consuming strategy orientation (Covin and Slevin, 1991) which enhance to make large resource contributions, the following hypotheses are proposed:

H21: Family Management weakness the effect of the Firm Entrepreneurial Orientation on Product Innovation.

H22: Family Management weakness the effect of the Firm Entrepreneurial Orientation on Incremental Innovation.

H23: Family Management weakness the effect of the Firm Entrepreneurial Orientation on Radical innovation.



Figure 1 Hypotheses.

#### Methodology

#### Data collection and sample

The data collection for the realization of the study has been carried out through a questionnaire made to a total of 634 managers of companies located in the Region of Murcia (Spain), as a part of research activity promoted by the "Observatorio Económico de la PYME", a Research Centre for SME's funded by the Instituto de Fomento de la Region de Murcia<sup>1</sup>.

The Region of Murcia is a fairly representative territory of the national panorama, since, in relation to the Spanish national average (99,90%), it is composed of a total of 99,92% of micro, small and medium enterprises (SMEs). In this sense, SMEs are defined, according to Recommendation 2003/361/CE, to micro, small and medium enterprises that have less than 250 employees and whose annual turnover does not exceed 50 million euros or whose balance sheet annual does not exceed 43 million euros.

The sample selection process (design process) was designed in accordance with the objectives of the study and the characteristic structure of the region following the stratified sampling principles in finite populations. The population of firms was segmented by size and industry. The number of firms in each stratum was implemented according to the information contained in the Companies Registration Office following the criteria of the "Instituto Nacional de Estadística" (Spanish Statistical Office).

The resulting sample is formed by 634 family businesses. In turn, within the family businesses, there are family businesses managed by family members, 571 (90,06%), and companies managed by non-family members, 63 (9,94%), a similar proportion to other studies carried out (Huybrechts et al., 2013). On the other hand, family businesses have been classified, firstly, according to the sector in which they carry out their business activity, thus differentiating between the Industry (51,42%), Construction (10,88%) and Services ( 37,70%); and secondly, depending on their size, taking into account the number of employees working in them, differentiating between Small (83,75%), Medium (14,36%) and Large (1,89%).

The table 1 collects the final distribution of the sample.

<sup>&</sup>lt;sup>1</sup> The Instituto de Fomento de la Región de Murcia is the development agency of the Murcia's region. It is a public institution that belongs to the Manufacturing, Firm and Innovation Office.

Table 1 Descriptive statistics of the sample.							
	Ν	%					
Family-Managed firms	571	90,06%					
Non-Family Managed firms	63	9,94%					
Manufacturing	326	51,42%					
Construction	69	10,88%					
Services	239	37,70%					
Small firm (up to 50 employees)	531	83,75%					
Medium firm (from 50 to 250 employees)	91	14,36%					
Large firm (more than 250 employees)	12	1,89%					
Total	634	100,00%					

The estimation of the sample considers in the worst case (relative frequency of answers in a specific item is p = q = 0.5), to a maximum error of 4,06% at a confidence level of 95%. Companies that chose to not participate in the project were replaced with similar (random election) firms in same industry and geographic area. the Information was collected through personal interviews with firm managers during April 2010 2010, using and July а self-managed questionnaire addressed to firm's CEO. SME's managers are the most important decision makers (Van Gils, 2005) and managerial perceptions influence to a significant degree the firm's strategic behavior (O'Regan & Sims, 2008). Control tests were carried out during the elaboration process of the survey. To test the validity of the survey, the bias of non-response was analyzed. Due to the nature of the data, it is possible that the relations between the variables were inflated as a consequence of the common method variance, since the same source is used to gather data for both the independent and dependent variable. This is why the bias of potential problem of the variance of the common method was also analyzed. To test for nonresponse bias, we used late respondents as surrogates for non-respondents Nwachukw et al., 1997), as well as the responses of the companies that responded to the first round of interviews (81% of the sample) were contrasted with those late responses (19% of the sample), and the results obtained showed no significant differences between the two groups, taking into account the Student t and chi-squared statistics. Considering these outcomes, non-response and industry bias were not found. Likewise, to test the bias of potential problem of the variance of the common method, we used the Harman's single-factor test suggested by Podsakoff and Organ (1986). In this way, we have realized a factorial analysis including all the dependent variables and independent variables, achieving to explain a high amount of the variance (Christmann, 2000). In the factorial analysis executed in our study, three factors were obtained (KMO: 0.841; Bartlett sphericity test Sig. 0.000) which explained a 58,430% of the total variance. Between these factors, the first one collects the dependents variables explained

a 25,174% of the variance. These results suggested that the bias of the common method variance was not relevant in our study (Podsakoff y Organ, 1986).

#### Variables definition

#### Product innovation

A company is innovative in products when it introduces a new or significantly improved good or service with respect to its characteristics or previous uses (Manual de Oslo, 2005; Laursen y Salter, 2006).

There are two types of product innovation: (i) on the one hand, radical innovation, which consists of the contribution of goods and services that differ significantly in their characteristics or intended uses with respect to the products previously produced by the company; and (ii) on the other, incremental innovation, which consist of a series of minor changes in the technical specifications of the products already existing in the company (Oslo Manual, 2005).

Product innovation can be measured through two approaches, first, an objective approach, and secondly, subjective approach. one The objective approach is used to measure the innovation from quantitative data (number of patents, number of new products, investments cost), while the subjective approach is based on the firm's manager or owner's perception about the innovative activity of the firm. In this work, we have opted for the subjective approach because it is the most appropriate for the case of SMEs and avoids underestimating the innovative activity of SMEs, in particular incremental activity, which the objective approach may overlook (Hughes, 2001). Subjective measures were highly correlated with objective measures of innovation and allowed comparisons among firms (Frishammar and Hörte, 2005). Also, subjective measures are valid to identifying and monitoring obstacles that decrease innovation among SMEs (Kalantaridis and Pheby, 1999). Furthermore, the results of Denison and Mishra (1995, pp. 219) supported that "subjective measures of effectiveness are better suited for the comparison of a disparate set of firms than are the objective measures of effectiveness"

#### Entrepreneurial Orientation

Based on the conceptualization of Miller (1983), and other previous studies (Naman and Slevin, 1993; Wiklund and Shepherd, 2005), three dimensions of entrepreneurial orientation have been used in the present work to measure the entrepreneurial orientations, which are innovativeness, risk-taking, proactiveness. The dimension of innovativeness refers to the predisposition of the company to support research and technological development activities via R&D; the risk-taking dimension

Table 2 Variables definition.

implies keeping an active position in situations of uncertainty and carrying out risky projects; and the dimension of proactivity implies the maintenance of a prospective position in front of the competition, identifying new business opportunities in the market and carrying out actions in advance that are later imitated by the competition.

Considering a combination of three dimensions of entrepreneurial orientation, we obtained this variable through a factor analysis. The result obtained with the combination of the three dimensions can be seen in table 2.

#### Family management

The family management has been measured through the participation of family members in the management of the company. In this sense, family involvement is represented by the variable called Family Management, which has been measured as a binary variable, which takes value 1 when family members hold mostly the management positions (Vanderkerkhof et al., 2015), and the value 0 when the management positions are hold mostly by professionals from outside the family.

Table 2 summarizes the contents relating to the definition of the variables in the models

Variable	Definition	Measure
Family Management	Family versus non-family management	1: Family members. 0: Non - family members.
Entrepreneurial Orientation	Entrepreneurial orientation of the company	This variable has been obtained through a factorial analysis of the different questions of the survey related to the firm entrepreneurial orientation. The questions were: Please indicate the degree of agreement with the following statements (1: very unfavorable; 5: very favorable): 1- The managers of my company support research and technological development activities 2- My company is very inclined to take on risky projects with a lot of market potential 3- The actions carried out in my company are soon imitated by the competition 4- My company has a great ability to identify new business opportunities 5- My company adopts an active position in uncertainty situations The results were: - Explained variance: 48,5%; - $x^2(10) = 849,43;$ - Prob> $x^2 < 0,000;$ - KMO: 0,772; - Cronbach $\alpha$ ; 0,729.
Product innovation - Incremental innovation - Radical innovation	Existence and importance of Product innovation into the company	With a range 0-5, this variable was built as the average of two sets of dummy and complementary licker variable, where 0: no innovation; 1-5: the higher of the value, the higher the importance of the innovation. The questions were: Has the firm made any changes or improvements in its products in the past year considering (a): incremental innovation; (b): radical innovation? If so, indicate the degree of importance of these changes for your business.
Performance	Evolution of certain aspects of performance in the company.	This variable has been obtained through a factorial analysis of the different questions of the survey related to the performance of a company. The questions were: Please indicate how has the evolution been of the following aspects of your business in the last two years (1: very unfavorable; 5: very favorable): 1- Improvements in market share; 2- Improvements in profitability; 3- Improvements in productivity. The results were: - Explained variance: 78,9%; - x <sup>2</sup> (3) = 1314,49; - Prob> x <sup>2</sup> < 0,000; - KMO: 0,728; - Cronbach q: 0.867.
Financial Position	Financial aspects in the company	This variable has been obtained through a factorial analysis of the different questions of the survey related to financial position of the company. The questions were: Please indicate how has the evolution been of the following aspects of your business in the last two years (1: very unfavorable; 5: very favorable): 1- Liquidity and cash; 2- Leverage indebtedness; 3- Debt service capacity; 4- Cost of debt; 5- Self-financing capability (to retain earnings). The results were: - Explained variance: 63,2%; - x <sup>2</sup> (10) = 1877,13; - Prob> x <sup>2</sup> < 0,.000; - KMO: 0,838; - Cronbach c: 0,853.
Size	Number of employees	Logarithm (ln) of the number of employees of the company
Age	The number of years since the firm started the business	Number of years since the firm was created
Industry	Sector of the activity	1. Manufacturing. 2. Construction. 3. Services.

existence of multicollinearity has been contrasted through the variance inflation factor

(VIF), obtaining values between 1,12 and 3,45, depending on the estimated model. Therefore, we discarded the presence of multicollinearity. Also, all the models have been estimated applying the robust estimator of White, for which consistent standard errors have been obtained, and, in accordance with that, we ruled out the problem of the heteroscedasticity. Finally, the obtained statistically significant F values confirm the overall validity of the estimated models and significance of the the ioint estimated parameters.

In order to make the estimations of the regression models, we have obtained different models of direct relationship between the Entrepreneurial Orientation and Family Management and Product Innovation (Models 1 and 2), Incremental Innovation (Models 4 and 5) and Radical Innovation (Models 7 and 8), adding in turn, to the previous models, a moderating variable, this is the Family Management, in order to estimate the possible effect of interaction of family involvement on the effect of the Entrepreneurial Orientation on Product Innovation (Model 3), Incremental Innovation (Model 6) and Radical Innovation (Model 9). Thus, the following steps have been followed to estimate each model: at first, Models 1, 4 and 7, the control variables into took only consideration; then, Models 2, 5 and 8, incorporated the direct relationship variables; and finally, Models 3, 6 and 9, incorporated the

10

						Interaction	1 effect va	anable		
Bef nec	ore describing essary to indi	the result icate that	s of each in all of	model, it them th	is e	interaction				
Tab	le 4 Correlations r	natrix.								
		1	2	3	4	5	6	7	8	Ģ
1	Product innovation	1								
2	Incremental innovation	0,8898***	1							
3	Radical innovation	0,8791***	0,5647***	1						
4	Entrepreneurial Orientation	0,3120***	0,2525***	0,3002***	1					
5	Family management	-0,1422***	-0,1232***	-0,1286***	-0,0916**	1				
6	Performance	0,1400***	0,1320***	0,1152***	0,3165***	-0,0847**	1			
7	Financial Position	0,1929***	0,2015***	0,1381***	0,2227***	-0,0861**	0,3950***	1		
8	Age	0,0959**	0,0983**	0,0708*	0,0379	-0,0867**	0,0188	0,1348***	1	
9	Size	0.1553***	0.1289***	0.1459***	0.1079**	-0.1954***	0.1461***	0.1691***	0.1841***	

Pearson's bivariate correlations \*\*\*p<0.01 \*\*p<0.05 \*p<0.10

-0,1127\*\*\*

2

-0,1644\*\*\*

3

-0,1560\*\*\*

1

10

Industry

Fredyna, T. Ruíz-Palomo, D. Dieguez, J. (2019). Entrepreneurial orientation and product innovation. The moderating role of family involvement in management. *European Journal of Family Business*, 9(2), 128-145.

-0,0901\*\*

4

0,0246

5

0,0428

6

-0,0330

7

-0,1749\*\*\*

8

-0,059

9

1

10

Table 3 summarizes the main descriptive statistics of the variables and reports the means, standart desviations, as well as minimum and maximum values of the variables for the sample. Beside, table 4 gathers the bivariate correlations between variables.

To verify the proposed hypotheses Ordinary Least Squares (OLS) general linear regression models were used. Nine models were estimated. Table 5 displays the resulting regression models.

Table 3	Descriptive	statistics.

	Mean	Standart desviation	Min	Max
Product innovation	3,039	3,321	0,000	10,00
Incremental innovation	1,702	1,918	0,000	5,000
Radical innovation	1,338	1,836	0,000	5,000
Entrepreneurial Orientation	0,000	1,000	-2,776	2,417
Family management	0,901	0,299	0,000	1,000
Performance	0,000	1,000	-2,074	2,270
Financial Position	0,000	1,000	-2,189	2,182
Age	22,98	16,505	0,000	159,0
Size	14,566	1,527	8,854	18,891
Industry	1,863	0,935	1,000	3,000

#### 138

Table 5 Regressions.									
	Pro	duct innovat	ion	Incre	Incremental innovation		Radical innovation		
	MODEL 1	MODEL 2	MODEL 3	MODEL 4	MODEL 5	MODEL 6	MODEL 7	MODEL 8	MODEL 9
Entrepreneurial Orientation (EO)		0.897	1,738		0,398	0,766		0,498	0,971
		(0.141)***	(0,337)***		(0,086)***	(0,201)***		(0,078)***	(0,199)***
Family management		-1.014	-0,879		-0,505	-0,446		-0,509	-0,433
		(0.441)**	(0,423)**		(0,263)*	(0,258)*		(0,242)**	(0,233)*
EO x Family management			-0,942			-0,412			-0,530
			(0,370)**			(0,220)*			(0,216)**
Performance	0,265	0,006	0,027	0,103	-0,015	-0,006	0,162	0,021	0,033
	(0,145)*	(0,144)	(0,144)	(0,086)	(0,087)	(0,087)	(0,082)**	(0,081)	(0,081)
Financial position	0,419	0,330	0,350	0,303	0,264	0,272	0,116	0,066	0,077
	(0,151)***	(0,139)**	(0,139)**	(0,088)***	(0,085)***	(0,085)***	(0,084)	(0,078)	(0,078)
Age	0,007	0,005	0,005	0,005	0,005	0,004	0,002	0,001	0,000
	(0,010)	(0,009)	(0,009)	(0,005)	(0,005)	(0,005)	(0,005)	(0,005)	(0,005)
Size	0,233	0,177	0,166	0,098	0,071	0,067	0,135	0,105	0,099
	(0,094)**	(0,093)*	(0,093)*	(0,055)*	(0,054)	(0,055)	(0,053)**	(0,052)**	(0,052)*
Industry	-0,494	-0,429	-0,431	-0,188	-0,162	-0,163	-0,305	-0,267	-0,268
	(0,150)***	(0,146)***	(0,145)***	(0,089)**	(0,087)*	(0,087)*	(0,082)***	(0,080)***	(0,080)***
Constant	0,443	2,111	2,142	0,533	1,348	1,362	-0,090	0,763	0,781
	-1,411	-1,559	-1,550	(0,818)	(0,883)	(0,878)	(0,792)	(0,864)	(0,861)
F	9,90***	16,37***	15,70***	8,49***	12,04***	11,13***	7,99***	14,16***	13,74***
R <sup>2</sup>	0,08	0,15	0,16	0,06	0,11	0,11	0,06	0,14	0,14
VIF	1,12	1,14	3,45	1,12	1,14	3,45	1,12	1,14	3,45
Ordinary Least Squares regressions, No standardized OLS coefficients reported (Robust standard errors in parentheses), ***p<0,01; **p<0,05; *p<0,10.									

Firstly, as we indicated, in order to measure the direct effect of Entrepreneurial Orientation and Family Management on Product Innovation, Models 2, 5 and 8 were estimated.

Yi = B0 + B1 EO +B2 Family management + B3

Performance + B4 Financial position + B5 Age + B6 Insize + B7 Sector + εi

These models not only measure the influence of Entrepreneurial Orientation Family and Management in Product Innovation (Model 2), but also differentiate such influence into Radical innovation (Model 5) and Incremental innovation (Model 8). In Model 2, the dependent variable (Yi) is Product Innovation, and in Models 5 and 8, the dependent variables are Incremental Innovation and Radical Innovation respectively. The global validity of the models was confirmed because the F value was significant (F=16,37\*\*; F=12,04\*\*; F=14,16\*\*). In all models we found a positive and significant Beta associated with Entrepreneurial Orientation (B1=0,897\*\*\*; B1=0,398\*\*\*; B1=0,498\*\*\*) and a negative and Beta associated with Family significant Management (B2=-1,014\*\*; B2=-0,5045\*; B2=-0,509\*\*). These results imply, on the one hand,

that Entrepreneurial Orientation positively affects Product Innovation, Incremental innovation and Radical Innovation, verifying H11, H12, H13, and, on the other hand, that Family Management has a negative effect on Product Innovation, Incremental innovation and Radical Innovation.

Also, Models 3, 6 and 9 have measured the moderating effect of Family Management on the relationship between Entrepreneurial Orientation and Product Innovation, Incremental innovation and Radical Innovation.

Yi = 60 + 61 EO + 62 Family management + 63 EO\*Family management + 64 Performance + 65 Financial position + 66 Age + 67 Insize + 68 sector + εi

In these models, the dependent variable (Yi) is Product Innovation (Model 3), Incremental Innovation (Model 6) and Radical Innovation (Model 9). The global validity of the models was confirmed because the F value was significant (F=15,70\*\*; F=11,13\*\*; F=13,74\*\*). Furthermore, in all models we found a positive and significant Beta (B3=-0,942\*\*; B3=-0,412\*; B3=-0,530\*\*), what indicates that Family Management

moderate the effect of Entrepreneurial Orientation on Product Innovation, Incremental innovation and Radical Innovation and verify the hypotheses H21, H22 y H23.

In addition, once the hypotheses H21, H22 y H23 were verified, we propose some graphics in figure 2 in order to better visualize and interpret

the moderating effect of Family Management on the relationship between Entrepreneurial Orientation and Product Innovation, Incremental innovation and Radical Innovation.



- + - Family management



Figure 2 Moderating effects.

In the graphics shown in figure 2 it can be seen that in family-managed firms, the positive effect of Entrepreneurial Orientation on Product Innovation, Incremental Innovation and Radical Innovation is lower in non-family managed firms. The table 6 gathers the quantitative marginal effects of the analyzed moderating effect.

**Table 6** Marginal effects of Entrepreneurial Orientation onInnovation depending on the family involvement in themanagement of the company.

	$\begin{array}{c} \text{Marginal effects of Entrepreneurial} \\ \text{Orientation} \\ (\frac{\partial \widehat{v}_i}{\partial \text{OREMP}_i}) \end{array}$					
	Product innovation	Incremental innovation	Radical innovation			
Family management	0,796	0,354	0,441			
Non - family management	1,738	0,766	0,971			

Some of concerning control variables were resulted statistically significant in the estimated models and, consequently, have an effect on Entrepreneurial Orientation and / or Product innovation.

#### **Discussion and Conclusions**

entrepreneurial The orientation can be considered as practice, philosophy or strategic process that leads companies to innovation (Miller and Friesen, 1982; Wiklund and Shepherd, 2005). Innovation is a relevant factor of economic progress (Nelson, 1991) and it is an important aspect for the sustained performance (Blundell et al., 1993), survival (Dyer and Song, 1998), profitability (Ali et al., 1993), growth and expansion (Danneels and Kleinschmidt, 2001) of the company. This is also true that family firm context presents big interest in the field of entrepreneurial orientation (Zahra et al.. 2004; Steier, 2007; Steier, 2009) and innovation (Craig and Moores, 2006; Pittino and Visintin, 2009). Thus, there are many researches that analyse the relationship between the abovementioned variables in different ways, but, unfortunately, the study of the relationship between entrepreneurial innovation and product innovation distinguishing between incremental and radical innovation has been practically nonexistent as far as we know.

Additionally, to complete our research, we also analysed the moderating effect of family managed firms on the abovementioned relationship.

To achieve our objective, we firstly studied the relationship between entrepreneurial orientation and product innovation, including radical innovation and incremental innovation. Using Miller´s (1983) three original dimensions characteristic of entrepreneurial orientation (Naman and Slevin, 1993; Wiklund and Shepherd, 2005), the obtained results supported the positive association between entrepreneurship and product innovation (Zhou et al., 2005; Nasution et al. 2010: Fernández-Mesa et al., 2012), and radical innovation (Avlonitisa and Salavoub, 2007; Zhou, Yim, and Tse, 2005; Salavou and Lioukas, 2003), as well as indicated favourable influence of entrepreneurial а orientation on incremental innovation. After these results, we could reaffirm and reinforce that entrepreneurial orientation has a positive effect on the product innovation, including incremental innovation and radical innovation.

Secondly, considering the importance that product innovation has for family firms (De Massis et al., 2013), we analysed the influence of the family involvement in the management of firm on the relationship between the entrepreneurial orientation and product innovation, also distinguishing between radical innovation and incremental innovation. Our findings showed that family managed firms relationship between weaken the entrepreneurial orientation and product innovation.

These results are consistent with the work of Chen and Hsu (2009) and the traditional vision of family forms as more conservative organizations (Sharma, Chrisman, and Chua, 1997), less keen to change (Kets de Vries, 1993; Ward, 1997), more risk-averse (Naldi et al., 2007), and with greater difficulty to get access to capital markets (Kets de Vries, 1993). Therefore, the obtained results reveal that family involvement in firms' management debilities the positive effect of entrepreneurial orientation on product innovation, including radical and incremental innovation.

Nevertheless, the above not mean that family managed firms do not innovate. Analysing the effect of family management on radical and incremental innovation, our results showed, not very deeply, how the family management has a major moderating effect on radical innovation versus incremental innovation. The aforementioned difference is due to the fact that incremental innovation is derived from exploiting current resources and capabilities and searching of continuous improvements of the firm and has less impacts on processes and methods of operating of the company (March, 1991), and it is

not as risky and does not require as much financial resources as radical innovation (Bessant, Birkinshaw and Delbridge 2004; Sorescu and Spanjol, 2008).

In conclusion, our study provides a new contribution in the literature of family business, particularly with the analysis how the family management moderates the effect of the entrepreneurial orientation on product innovation relationship, as far as we know, the previous literature has not analysed yet. The study's findings suggest that high level of family effect management reduces the of entrepreneurial orientation on product innovation, specifically in radical innovation. On the contrary, when family involvement on manager is low, the entrepreneurial orientation effect on product innovation, radical innovation and incremental innovation is higher. Therefore, the family management should be taken into account in the field of family firms due to its importance in innovation, since that creates wealth and opens up new business opportunities, and entrepreneurship, as they are central figures growth economic and development of (Schumpeter, 1961).

This conclusion is important for family managed firms due to (i) it could help stakeholders and policy-makers to make decisions about subsidies. investments, training, etc, which would help firms to improve their efforts on product innovation. In this sense, this negative moderator effect of family management could be corrected by the participation of non-family members in firms' management or with the professionalization and teaching of their own managers. Also, public institutions should facilitate family managed firms greater resources so that they can innovate more, what will not improve the development of family only businesses, but national also economy. Furthermore, due to that Product Innovation is a key factor of industrial competitiveness and national development, being an important determinant of sustained vield (Blundell et al., 1999).

This study is not free from limitations. Firstly, the little research exists on the innovation behaviour of family firms. Secondly, the study is limited to analyzing Spanish companies, specifically in Murcia, so their results might not be generalizable to companies from other regions or countries. Thirdly, this study treats family firms as a homogeneous category instead of taking into account the differences that exist between various types of family firms.

Several research extensions can be derived from this article. Firstly, future research should also theoretically and empirically study if our findings apply to other types of innovations, for example in (i) Process innovation, (ii) Organizational innovation or (iii) Marketing innovation. Secondly, the relationship between entrepreneurial orientation and innovation also could be analysed considering other forms of family involvement, such as in ownership or governance. Similarly, the role of new generations should be considered, as each generation will bring different management styles and objectives to the family firm. Finally, the expansion of the sample to the international arena would allow the generalization of these conclusions.

#### References

- Aldrich, H. E., and Cliff, J. E. (2003). The pervasive effects of family on entrepreneurship: Toward a family embeddedness perspective. Journal of Business Venturing, 18(5), 507-525.
- Ali A, Kalwani M.U, and Kovenock D. (1993). Selecting product development projects: pioneering versus incremental innovation strategies. Management Science, 39(3), 255-274.
- Amit, R. and Schoemaker, P. (1993). Strategic assets and organizational rent. Strategic Management Journal, 14, 33-46.
- Anderson, P., and Tushman, M. (1990). Technological Discontinuities and Dominant Designs: A Cyclical Model of Technological Change. Administrative Science Quarterly, 35, 604-633.
- Aronoff, C. E., and J. L. Ward (1995). Family-Owned Businesses: A Thing of the Past or a Model for the Future? Family Business Review, 8(2), 121-130.
- Astrachan, J. H., Klien, S. B., and Smyrnios, K. X. (2002). The F-PEC scale of family influence: A proposal for solving the family business definition problem. Family Business Review, 15(1), 45-58.
- Atuahene-Gima, K. (2001). An empirical investigation of the effect of market orientation and entrepreneurship orientation alignment on product innovation. Organization Science, 1(2), 54-74.
- Avlonitisa G. J., and Salavoub H. E. (2007). Entrepreneurial orientation of SMEs, product innovativeness, and performance. Journal of Business Research, 60(5), 566-575.
- Baker, W. E., and Sinkula, J. M. (2009). The Complementary Effects of Market Orientation and Entrepreneurial Orientation on Profitability in Small Businesses. Journal of Small Business Management, 47, 443-464.
- Barney, J., (1991). Firm Resources and Sustained Competitive Advantage. Journal of Management, 17, 99-120.
- Bessant, J., Birkinshaw, J., and Delbridge, R. (2004). Innovation as Unusual. Business Strategy Review, 15(3), 32-35.
- Bettis R.A., Hitt M.A. (1995). The New Competitive Landscape. Strategic Management Journal, Summer Special Issue, 16, 7-19.
- Blundell, R., Griffiths, R., and Van Reenen, J. (1999). Market share, market value and innovation in a panel of British manufacturing firms. Review of Economic Studies, 66(3), 529-554.
- Boso, N., Cadogan, J. W., and Story, V. M. (2013). Entrepreneurial orientation and market orientation as drivers of product innovation success: A study of exporters from a developing economy. International Small Business Journal, 31(1), 57-81.

- Calantone, R.G., Cavusgil, S.T., and Zhao, Y. (2002). Learning orientation, firm innovation capability and firm performance. Industrial Marketing Management, 31, 515-524.
- Casillas, J.C., and Moreno, A.M. (2010). The Relationship between Entrepreneurial Orientation and Growth: The Moderating Role of Family Involvement. Entrepreneurship & Regional Development, 23, 27-44.
- Casillas, J.C., Moreno, A.M., and Barbero, J.L. (2010). A configurational approach of the relationship between entrepreneurial orientation and growth of family firms. Family Business Review, 13(1), 27-44
- Chen, H. L., and Hsu, W. T. (2009). Family, Ownership, Board Independence, and R&D Investment. Family Business Review, 22(4), 347-362.
- Choi, Y. R., Zahra, S. A., Yoshikawa, T., and Han, B. H. (2015). Family Ownership and R&D Investment: The Role of Growth Opportunities and Business Group Membership. Journal of Business Research, 68(5), 1053-1061.
- Chrisman, J. J., Chua, J. H., and Litz, R. A. (2003). Commentary: A unified perspective of family firm performance: An extension and integration. Journal of Business Venturing, 18(4), 467-472.
- Chrisman, J. J., Chua, J., and Steier, L. P. (2003). An introduction to theories of family business. Journal of Business Venturing, 18(4), 441-448.
- Chrisman, J. J., J. H. Chua, and R. A. Litz. (2004). Comparing the agency costs of family and nonfamily firms: Conceptual issues and exploratory evidence. Entrepreneurship Theory and Practice, 28 (4): 335-54.
- Classen, N., Carree, M., Van Gils, A., and Petters, B. (2014). Innovation in family and non-family SMEs: an exploratory analysis. Small Business Economics, 42(3), 595-609.
- Clausen, T.H., and Pohjola, M. (2013). Persistence of product innovation: comparing breakthrough andincremental product innovation. Technology Analysis & Strategic Management, 25(4), 369-385.
- Craig, B. L., and Moores, K. (2006). A 10-Year Longitudinal Investigation of Strategy, Systems, and Environment on Innovation in Family Firms. Family Business Review 19(1), 1-10.
- Covin, J. G., and Slevin, D. P. (1989). Strategic management of small firms in hostile andbenign environments. Strategic Management Journal, 10, 75-87.
- **Covin, J. G., and Slevin, D. P. (1988).** The influence of organization structure on the utility of an entrepreneurial top management style. Journal of Management Studies, 25(3), 217-234.
- Covin, J. G., and Slevin, D. P. (1991). A Conceptual Model of Entrepreneurship as Firm Behavior. Entrepreneurship Theory and Practice, 16, 7-25
- Covin, J. G., and Wales, W. J. (2018). Crafting High-Impact Entrepreneurial Orientation Research: Some Suggested Guidelines. Entrepreneurship Theory and Practice. Entrepreneurship Theory and Practice, 43(1), 3-18.
- Damanpour, F. (1992). Organizational Size and Innovation. Organization Studies, 13(3), 375-402.
- Danneels E., and Kleinschmidt E.J. (2001). Product innovativeness from the firm's perspective: its dimensions and their relation with project

selection and performance. Journal of Product Innovation Management, 18(6), 357-373.

- Denison, D. R., A. K. Mishra. (1995). Toward a theory of organizational culture and effectiveness. Organ. Sci, 6(2) 204-223.
- De Massis, A., Frattini, F., and Lichtenthaler, U. (2013). Research on Technological Innovation in Family Firms: Present Debates and Future Directions. Family Business Review, 26, 10-31.
- Dess, G.G. Y Lumpkin, G.T. (2005). The role of entrepreneurial orientation in stimulating effective corporate entrepreneurship. Academy of Management Executive, 19, 147-156.
- Dyer, W. G., JR. (2006). Examining the "family effect" on firm performance. Family Business Review, 19(4), 253-273.
- Dyer, B., and Song, M.X. (1998). Innovation strategy and sanctioned conflict: a new edge in innovation? Journal of Product Innovation Management, 15(6), 505-519.
- Ferreira, J.J., Azevedo, G.S. and Fernández, R. (2011). Contribution of Resource-Based View and Entrepreneurial Orientation on Small Firm Growth, Cuadernos de Gestión, 11(1), 95-116.
- Fernández-Mesa, A., Alegre-Vidal, J., and Chiva-Gómez, R. (2012). Orientación Emprendedora, Capacidad de Aprendizaje Organizativo y Desempeño Innovador. Journal of technology management and innovation, 7(2), 157-170.
- Frishammar, J., and Hörte, S. (2005). Managing External Information in Manufacturing Firms: The Impact on Innovation Performance. Journal of Product Innovation Management, 22(3), 251-266.
- Frishammar, J. and Hörte, S. (2007). The role of market orientation and entrepreneurial orientation for new product development performance in manufacturing firms. Technology Analysis & Strategic Management 19(6): 765-788.
- Furtan, W.H., and Sauer, J. (2008). Determinants of food industry performance: survey data and regressions for Denmark. Journal of Agricultural Economics, 59 (3), 555-573.
- Galve Gorriz, C., Y Salas Fumas, V. (1996). "Ownership structure and firm performance: Some empirical evidence from Spain", Managerial and Decision Economics, 17, pp.587-594.
- Gatingnon H., and Xuereb J. M. (1997). Strategic orientation of the firm and new product performance. Journal of Marketing Research, 34(1), 77-90.
- George, B. A., & Marino, L. (2011). The epistemology of entrepreneurial orientation: conceptual formation, modeling, and operationalization. Entrepreneurship Theory and Practice, 35(5), 989-1024.
- George, G., Wood, D. R., and Khan, R. (2001). Networking strategy of boards: Implications for small and medium-sized enterprises. Entrepreneurship and Regional Development, 13(3), 269-285.
- Gómez-Mejía, L.R., Haynes, K.T., Núñez-Nickel, M., Y Moyano-Fuentes, H. (2007). Socioemotional wealth and business risk in family-controlled firms: Evidence from Spanish olive oil mills. Administrative Science Quarterly, 52, 106-137.
- Green, K. M., Covin, J. G., and Slevin, D. P. (2008). Exploring the relationship between strategic

reactiveness and entrepreneurial orientation: The role of structure-style fit. Journal of Business Venturing, 23(3), 356-383.

- Gudmundson, D., Hartman, E.A., and Tower, C.B. (1999). Strategic orientation: Differences between family and nonfamily firms. Family Business Review, 12(1), 27-39.
- Habbershon, T. G., Williams, M., and Macmillan, I. C. (2003). A unified systems perspective of family firm performance. Journal of Business Venturing, 18, 451-465.
- Hall, A., Melin, L., and Nordqvist, M. (2001). Entrepreneurship as a radical change in family business: Exploring the role of cultural patterns. Family Business Review, 14, 193-208.
- Hall, R. (1992). The strategic analysis of intangible resources. Strategic Management Journal, 13 (1), 135-144.
- Hart, S. L. (1992). An integrative framework for strategy-making processes. The Academy of Management Review, 17(2), 327-351.
- Hitt, M.A., Ireland, R.D., Camp, S.M., & Sexton, D.L. (2001). Strategic entrepreneurship: Entrepreneurial strategies for wealth creation. Strategic Management Journal, 22 (special issue), 479-491.
- Hughes, M., and Morgan, R.E. (2007). Deconstructing the relationship between entrepreneurial orientation and business performance at the embryonic stage of firm growth. Industrial Marketing Management, 36, 651-661.
- Hughes, A. (2001). Innovation and business performance: Small entrepreneurial firms in the UK and the EU. New Economy, 8(3), 157-163.
- Hurmelinna-Laukkanen, P., Sainio, L. M., and Jauhiainen, T. (2008). Appropriability Regime for Radical and Incremental Innovations, R&D Management, 38(3), 278-289.
- Huybrechts, J., Voordeckers, W., and Lybaert, N. (2013). Risk taking of private family firms: the influence of a nonfamily CEO and the moderating effect of CEO tenure. Family Business Review, 26, 161-179.
- Ireland, R. D., Hitt, M. A., and Sirmon, D. G. (2003). A model of strategic entrepreneurship: The construct and its dimensions. Journal of Management, 29(6), 963-989.
- Kalantaridis, C., and J. Pheby (1999). Processes of Innovation Among Manufacturing SMEs: The Experience of Bedfordshire. Entrepreneurship and Regional Development, 11(1), 57-78.
- Kets de Vries, M. F. R. (1993). The Dynamics of Family Controlled Firms: The Good and the Bad News. Organizational Dynamics, 21(3), 59-71.
- Laursen, K., and Salter, A. (2006). Open for innovation: the role of openness in explaining innovation performance among U.K. manufacturing firms. Strategic Management Journal, 27, 131-150.
- La Porta, R., F. Lopez-de-Silanes, A. Shleifer, and R. Vishny (1999). Corporate Ownership Around the World. Journal of Finance, 54(2), 471-517.
- Le Breton-Miller, I., and Miller, D. (2009). Agency vs Stewardship in Public Family Firms: A Social Embeddedness Reconciliation. Entrepreneurship Theory and Practice, 33(6), 1169-1191.
- Link, A.N. (1980). Firm size and efficient entrepreneurial activity: A reformulation of the

Schumpeter hypothesis. Journal of Political Economy, 88, 771-782.

- Lumpkin, G.T., and Dess, G.G. (1996). Clarifying the entrepreneurial orientation construct and linking it to performance. Academy of Management Review, 21(1), 135-172.
- Lumpkin, G.T., and Dess, G.G. (2001). Linking two dimensions of entrepreneurial orientation to firm performance: The moderating role of environment industry life cycle. Journal of Business Venturing, 16, 429-451.
- Madrid-Guijarro, A., Garcia, D., and Van Auken, H. (2009). Barriers to Innovation among Spanish Manufacturing SMEs. Journal of Small Business Management, 47(4), 465-488.
- March, J. G. (1991). Exploration and Exploitation in Organizational Learning. Organization Science, 2, 71-87.
- Margaret, M. (2008). The Role of Human Capital and Entrepreneurial Orientation on Radical Product Innovations in Small Scale Carpentry Workshops in Nairobi. Kenyatta University, Kenyatta.
- Menguc, B., and Auh, S. (2010). Development and Return on Execution of Product Innovation Capabilities: The Role of Organizational Structure. Industrial Marketing Management, 39(5), 820-831.
- Miles, R., Snow, C., Meyer, A., and Coleman, H. (1978). Organizational Strategy, Structure, and Process. The Academy of Management Review, 3(3), 546-562.
- Miller, D. (1983). The correlates of entrepreneurship in three types of firms. Management Science, 29, 770-792.
- Miller, D., and Friesen, P. H. (1982). Innovation in conservative and entrepreneurial firms: Two models of strategic momentum. Strategic Management Journal, 3, 1-25.
- Miller, D., Le Breton-Miller, I., and Lester, R. H. (2011). Family and Lone Founder Ownership and Strategic Behavior: Social Context, Identity, and Institutional Logics. Journal of Management Studies, 48(1), 1-25.
- Moreno, A.M., and Casillas, J.C. (2008). Entrepreneurial orientation and growth of SMEs: A causal model. Entrepreneurship Theory and Practice, 32(3), 507-528.
- Morris, M.H., and Paul, G.W. (1987). The relationship between entrepreneurship and marketing in established firms. Journal of Business Venturing, 2, 247-59.
- Naldi, L., Nordqvist, M., Sjöberg, K., and Wiklund, J. (2007). Entrepreneurial orientation, risk taking, and performance in family firms. Family Business Review, 20, 33-47.
- Naman, J.L., and Slevin, D.P. (1993). Entrepreneurship and the concept of fit: A model and empirical tests. Strategic Management Journal, 14, 137-153.
- Nasution, H., Mavondo, F., Jekanyika, M., and Oly, N., (2010). Entrepreneurship: Its relationship with market orientation and learning orientation and as antecedents to innovation and customer value. Industrial Marketing Management, 40, 336-345.
- Nelson, R.R. (1991). Why do firms differ, and how does it matter? Strategic Management Journal, 12, 61-74.
- Neubauer, F., and Lank, A. G. (1998). The family business: Its governance for sustainability. New York: Macmillan.

- Newey, L., and Zahra, S. (2009). The evolving firm: How dynamic and operating capabilities interact to enable entrepreneurship. British Journal of Management, 20(1), S81-S100.
- Nonaka, I. (1994). A dynamic Theory of Organizational Knowledge Creation. Organization Science, 5, 14-37.
- Nordqvist, M., Habbershon, T.G., and Melin, L. (2008). Transgenerational Entrepreneurship: Exploring Entrepreneurial Orientation in Family Firms, In Landström, H., Crijn, H. Laveren, E. &Smallbone, D., (Eds.) Entrepreneurship, Sustainable Growth and Performance: Frontiers in European Entrepreneurship Research, Northampton, Mass.: Edward Elgar Publishing Limited, 93-116.
- Nwachukwv, S., Vitell, S., Gilbert, F., and Barnes, J. (1997). Ethics and social responsibility in marketing: an examination of the ethics evaluation of advertising strategies. Journal of Business Research, 39(2), 107-118.
- **O'Regan, N., and Sims, M. (2008).** Identifying high technology small firms: A sectoral analysis. Technovation, 28, 408-423.
- **OECD, and Eurostat. (2005).** Oslo manual: Guidelines for collecting and interpreting innovation data (3rd ed.). Paris: Organisation for Economic Cooperation and Development, Statistical Office of the European Communities.
- Poazi, F., Tamunosiki-Amadi, J., Fems, M. (2017). The Resource-Base View of Organization and Innovation: Recognition of Significant Relationship in an Organization. World Academy of Science, Engineering and Technology, Open Science Index 123, International Journal of Economics and Management Engineering, 11(3), 697 - 704.
- Podsakoff, P.M., and Organ, D.W. (1986). Selfreports in organizational research: problems and prospects. Journal of Management, 12(4), 531-544.
- Rauch, A., Wiklund, J., Frese, M., and Lumpkin, G.T. (2009). Entrepreneurial orientation and business performance: An assessment of past research and suggestions for the future. Entrepreneurship Theory and Practice, 33, 761-781.
- Runyan, R., Huddleston, P. and Swinney, J. (2006). Entrepreneurial orientation and social capital as small firm strategies: a study of gender differences from a Resource-Based View. Entrepreneurship Management, (2), 455-477
- Salavou, H., and Avlonitis, G. (2008). Product innovativiness and performance: a focus on SMEs. Management Decision, 46 (7), 969-985.
- Salavou, H. and Lioukas, S. (2003). Radical Product Innovations in SMEs: The Dominance of Entrepreneurial Orientation. Creativity and Innovation Management, 12, 94-108.
- Salvato, C. (2004). Predictors of entrepreneurship in family firms. Journal of Private Equity, 27(3), 68-76.
- Schumpeter, J.A. (1934). The Theory of Economic Development. Cambridge, Massachusetts: Harvard University Press.
- Sharma, P. (2004). An Overview of the Field of Family Business Studies: Current Status and Directions for the Future. Family Business Review, 17(1), 1-36.
- Sharma, P., Chrisman, J. J., and Chua, J. H. (1997). Strategic Management of the Family Business: Past

Research and Future Challenges. Family Business Review 10, 1-36.

- Simpson, P.M., Sigaw, J.A., and Enz, C.A. (2006). Innovation orientation outcomes: the good and the bad. Journal of Business Research, 59, 1133-1141.
- Sirmon, D. G., Arrègle, J. L., Hitt, M. A., and Webb J. W. (2008). The Role of Family Influence in Firms' Strategic Responses to Threat of Imitation. Entrepreneurship Theory and Practice 32(6), 979-998.
- Sirmon, D.G., and Hitt, M.A. (2003). Managing resources: Linking unique resources, management, and wealth creation in family firms. Entrepreneurship Theory and Practice, 27, 339-358.
- Smart, D.T., and Conant, J.S.(1994). Entrepreneurial orientation, distinctive marketing competencies and organizational performance. Journal of Applied Business Research, 10, 28-38.
- Sorescu, A., and Spanjol, J. (2008). Innovation's Effect on Firm Value and Risk: Insights from Consumer Packaged Goods. Journal of Marketing, 72, 114-132.
- Souitaris, V. (2003), Determinants of technological innovation. Current research trends and future prospects in Shavinina, L (ed.), International Handbook on Innovation, Pergamon, 513-528.
- Stam. W., Elfring, T. (2008). Entrepreneurial Orientation and New Venture Performance; The moderationg role of Intra - and Extra - Industry Social Capital. Academy of Management Journal, 51, 97-111.
- Steier, L. (2007). New venture creation and organization: A familial sub-narrative. Journal of Business Research, 60, 1099-1107.
- Steier, L. (2009). Where Do New Firms Come From? Households, Family Capital, Ethnicity, and the Welfare Mix. Family Business Review, 22 (3), 273-278.
- Teece, D.J. (2007). Explicating dynamic capabilities: The nature and microfoundations of (sustainable) enterprise performance. Strategic Management Journal, 28, 1319-1350.
- Van Gils, A. (2005). Management and governance in Dutch SMEs. European Management Journal, 23(5), 583-589.
- Vandekerkhof, P., Steijvers, T., Hendriks, W., and Voordeckers, W. (2015). The Effect of Organizational Characteristics on the Appointment of Nonfamily Managers in Private Family Firms: The Moderating Role of Socioemotional Wealth. Family Business Review, 28(2), 104-122.
- Villalonga, B., and Amit, R. (2006). How do family ownership, control and management affect firm value? Journal of Financial Economics, 80(2), 385-417.
- Wales, W. J., Gupta, V. K., and Mousa, F.-T. (2013). Empirical research on entrepreneurial orientation: An assessment and suggestions for future research. International Small Business Journal, 31(4), 357-383.
- Ward, J. L. (1997). Growing the family business: Special challenges and best practices. Family Business Review, 10, 323-337.
- Westhead, P., and Howorth, C. (2007). Types of private family firms: An exploratory conceptual

and empirical analysis. Entrepreneurship and Regional Development, 19, 405-431.

- Wiklund, J., and Shepherd, D. (2005). Entrepreneurial orientation and small business performance: a configurational approach. Journal of Business Venturing, 20, 71-91.
- Workman J. P. (1993). Marketing's limited role in new product development in one computer systems firm. Journal of Marketing Research, 30, 405-421.
- Zahra, S. A. (2005). Entrepreneurial risk taking in family firms. Family Business Review, 18, 23-40.
- Zahra, S. A., Hayton, J. C., and Salvato, C. (2004). Entrepreneurship in family vs. non-family firms: A resource-based analysis of the effect of organizational culture. Entrepreneurship Theory and Practice, 28(4), 363-381.
- Zhou K. Z., Yim C. K., and Tse D. K. (2005). The effects of strategic orientations on technology and market-based breakthrough innovations. Journal of Marketing, 69(4), 42-60.

Fredyna, T. Ruíz-Palomo, D. Dieguez, J. (2019). Entrepreneurial orientation and product innovation. The moderating role of family involvement in management. *European Journal of Family Business*, 9(2), 128-145.



EUROPEAN JOURNAL OF FAMILY BUSINESS

http://www.revistas.uma.es/index.php/ejfb



# Private Equity focused on Family Firms & Small and Medium Sized Companies: Review and Science Mapping Analysis of the Recent Scientific Field.

Laura Arteche-Bueno<sup>a</sup>, Camilo Prado-Román<sup>a\*</sup>, Antonio Fernandez-Portillo<sup>b</sup>

<sup>a</sup> University Rey Juan Carlos (Spain)

<sup>b</sup> University of Extremadura. (Spain)

Received 10 August 2018; accepted 19 December 2019

JEL CLASSIFICATION F21, G11, G24, O16 KEYWORDS Private Equity; Family Firms; SMEs; Science Mapping; Bibliometrics	Abstract Private equity ("PE") is mostly invested in established firms, of which family firms ("FFs") are the dominant form. This article reports the recent evolution of the scientific research on the PE focused on FFs and small and medium-sized enterprises ("SMEs" or "the middle-market"). The purpose is to identify the main themes related to the field between 1992 and 2018 and to identify and analyze the major thematic areas throughout the period. The methodology applied is the science mapping analysis, which shows that: (i) published research on the field is concentrated in two main thematic areas: corporate governance-entrepreneurship and innovation-management, and; (ii) there has been an atomization of the research field during the last six years. Throughout this article, the authors develop a more complete understanding of the PE scientific field focused on family owned SMEs and provide suggestions for those looking for alternatives to traditional bank financing.
CÓDIGOS JEL F21, G11, G24, O16 PALABRAS CLAVE Capital Privado; Empresa Familiar; PyMEs; Mapeo Científico; Bibliometría	Capital privado centrado en empresas familiares y pequeñas y medianas empresas: revisión y análisis de mapeo científico del campo científico reciente. Resumen El capital inversión ( <i>private equity</i> ) se invierte principalmente en empresas establecidas, de las cuales las empresas familiares son la forma dominante. Este artículo analiza la reciente evolución de la investigación científica sobre <i>private equity</i> centrada en las Empresas Familiares y las pequeñas y medianas empresas (PYME). El propósito es identificar los temas principales relacionados entre 1992 y 2018 y analizar las principales áreas temáticas a lo largo del período. La metodología aplicada es <i>science mapping analysis</i> , que muestra que: 1. la investigación publicada en el campo se concentra en dos áreas temáticas principales: gobierno corporativo-emprendimiento y gestión de la innovación, y; 2. ha habido una atomización del campo de investigación durante los últimos seis años. La investigación realiza un análisis en profundidad para entender la literatura sobre <i>private equity</i> , centrandose en las PYME familiares, ofreciéndose indicaciones para aquellos que buscan alternativas al financiamiento bancario tradicional.

http://dx.doi.org/10.24310/ejfbejfb.v9i2.5285

Copyright 2019: Laura Arteche-Bueno, Camilo Prado-Román, Antonio Fernandez-Portillo.

European Journal of Family Business is an open access journal published in Malaga by UMA Editorial. ISSN 2444-8788 ISSN-e 2444-877X This work is licensed under a Creative Commons Atribution-NonCommercial-ShareAlike 4.0 International License (CC BY-NC-SA 4.0).

#### Introduction

The middle-market sector plays an important role in our financial system provided that, and according the European Commission, "SMEs are the backbone of Europe's economy: they represent 99% of all businesses in the region". In addition, SMEs have strong difficulties to obtain external funds for growth. The diversification of their financing sources is a key issue to allow room for growth and internationalization today and the PE is, in some cases, the main source of long term financing for them that has lots of advantages against other sources of financing. PE is an effective alternative to traditional financing for private SME as it provides with solid and sustainable business models to better deal with economic cycles. Many FFs are facing succession around the world (Shanker and Astrachan, 1996; Upton and Petty, 2000) and the challenge of ensuring succession of the business is a pressing global phenomenon (PWC, 2012); but PE has largely been ignored as a possible solution (Higashide & Birley, 2002; Howorth, Westhead and Wright, 2004).

FFs are of particular significance for the global economy (IFERA, 2003; Anderson and Reeb, 2003; Morck and Yeung, 2003; Astrachan and Shanker, 2003; Klein, 2000). With more than 14 million family businesses in Europe, their importance to the economy cannot be overestimated. In some countries, they represent anywhere from 55 to 90 percent of all businesses ("European Family Business Barometer", KPMG and EFB, 2017).

FFs are a heterogeneous group with varying degrees of family influence, differences in size, industry and geography (Chua, Chrisman, Steier and Rau, 2012; Chrisman, Chua, Pearson and Barnett, 2012; Tsang, 2002). Nowadays, ownership and management succession are one of the biggest challenges for FF. However, "many of them do not have the necessary resources and capabilities to grow or to manage generational succession" (Howorth et al., 2004; Shanker and Astrachan, 1996; Sirmon and Hitt, 2003; Upton and Petty, 2000). Succession is the most frequently studied topic in the family business literature (Chua, Chrisman and Sharma, 2003) but the exploration of nonfamily route to succession has not received much attention in the academic literature (Birley and Westhead, 1990; Howorth et al., 2004). According to the PwC Global Family Business Survey 2018, succession and access to financing are between the key challenges for FF over the next two years.

Despite the abovementioned, according to the European Central Bank (November of 2017) "banking products are the main source of financing for European SMEs, while other sources available in the market such as equity (which includes the PE) are hardly considered as a potential source of funds." But PE represents an alternative source to finance investment opportunities for a wide variety of firms (Martí, Menéndez-Requejo and Rottke, 2013). One possible solution to the succession problems is to open up the family firm's capital to PE investors (Dawson, 2011).

Several authors like Benavides-Velasco, Quintana-García and Guzmán-Parra (2013) and Voordeckers, Le Breton-Miller and Miller (2014) have shown that finance is not only one of the top areas in family business research but also a growing area. The importance is warrantied since the availability of sufficient financial resources is of critical importance for the FF's survival and growth (Koropp, Kellermanns, Grichnik and Stanley, 2014).

Simultaneously, the PE activity has become a major focus of study since the late 90s because of the increasing evidence on high performance PE funds, among other reasons. The positive effect generated by PE has been widely studied and demonstrated in the existing literature (Kaplan and Schoar, 2005; Metrick, and Yasuda, 2011; Haro De Rosario, 2013)

Given this evidence, the authors question whether, indeed, PE is a real alternative to bank financing for family owned SMEs and that, in addition, it offers great advantages over other sources of financing.

To solve this question, the authors first consider knowing what the historical evolution of the scientific field of PE focused on this type of companies has been. In order to understand what the most impactful and relevant topics in the field have been in the past and are nowadays, the authors develop in this research paper an empirical analysis of the field through a bibliometric analysis based on the analysis of scientific maps, developing a more complete understanding of the field and discovering current and future research areas relevant to both PEs and family owned SMEs.

This study aims also at making the family owned SMEs aware of the existence and advantages of the use of PE as an alternative to traditional financing and to promote it in the next years. This type of study was suggested by Michiels and Molly (2017).

in their review of Financing Decisions in Family Businesses. As described below, a high level of funds has been raised by middle-market PEs in 2017 and new fundraisings are expected to be closed in 2018, what results in an attractive opportunity for the family owned SMEs looking for speeding up growth in the next years.

Arteche Bueno, L.; Prado Román, C.; Fernandez Portillo, A(2019). Private Equity focused on Family Firms & Small and Medium Sized Companies: Review and Science Mapping Analysis of the Recent Scientific Field. *European Journal of Family Business*, 9(2), 146-158.

#### Evolution of the scientific PE field focused on FFS and the middle-market between 1992 and 2018

The word "PE" can mean risk capital invested in a wide range of companies and industries: from funds provided to start-ups and privately-owned SMEs to acquisitions of multinational companies publicly-traded even entire mature and companies (Gilligan and Wright, 2010). The scientific study of the PE sector with activity in the FFs SMEs market segment belongs to a relatively recent past: the first article indexed in the Web of Science ("WoS") appeared in year 1992 and significant volumes of high impact research did not appear until year 2007 (Cumming, Siegel and Wright, 2007; Cumming, 2007; Renneboog, Simons and Wright, 2007). Until then, years 2000 and 2001 were especially productive for the PE due to the boom of both the high-tech and the mergers and acquisitions sectors. Between 2001 and 2006, the European PE houses raised 193,786 million Euros in funds, being 2005 the peak year with 71,771 million Euros raised. The high level of activity of the period occurred in many nations (Wright, Amess, Weir and Girma, 2009a; Strömberg, 2008), culminating in a peak worldwide in 2007.

The onset of the financial crisis from 2008 resulted in a massive fall in deal value worldwide in 2009 as debt markets closed and PE firms set about restructuring troubled and overleveraged portfolio companies. Nevertheless, PE was not just a transitory phenomenon and PE firms have adapted to begin to build a new future (Wright, Jackson and Frobisher, 2010). Since 2010, worldwide PE showed signs of recovery, as the third quarter provided the strongest showing of the market since the financial crisis at an aggregate value of 66.7 billion US\$ (Preqin, 2010).

In the following years, there are signs that indicate that fundraising and deal making will be strong in the Europe, the Middle East and Africa region: leading European middle-market PE funds are right now in the process of fundraising. In the last few years fundraising among PE firms has hit record levels and most of surveys suggests this trend is to continue in 2018. The sector is, in addition, under a globalization process.

# The methodology: science mapping bibliometrics analysis through scimat software tool

In bibliometrics, science mapping analysis ("SMA") is designed to display the structural and dynamic aspects of scientific research, to determine the scope of a research field and to quantify and visualize the detected subfields by means of co-word analysis or document cocitation analysis. It is focused on monitoring a scientific field and delimiting research areas to determine its conceptual structure and scientific López-Herrera, evolution (Cobo, Herrera-Viedma, and Herrera, 2011b; Noyons, Moed, and van Rann, 1999b). In this article, the SMA is performed using the software Science Mapping Analysis Software Tool ("SciMAT") (Cobo, López-Herrera, Herrera-Viedma and Herrera, 2012b), designed and developed by the SECABA Laboratory at the University of Granada (Spain). SciMAT is based (Cobo, López-Herrera, Herrera-Viedma and Herrera, 2011a) on a co-word analysis (Callon, Courtial, Turner and Bauin, 1983) and the h-index (Hirsch, 2005), which are applied in a longitudinal framework.

# Data Sources

To obtain the publications of the journals and their citations, the bibliographic database WoS (property of Clarivate Analytics) is used. WoS is the world's leading scholarly literature database in the sciences and social sciences: it is a reference database that provides with the most complete current and retrospective quality coverage in the sciences and social sciences, going back to 1900 (Harzing and van der Wal, 2008). A database with this property is appropriate for developing a rigorous SMA of the PE field focused on SMEs with a longitudinal perspective.

# Sample

We focus on this analysis on articles dealing with all types of family businesses within the middlemarket, meaning that they can imply family involvement in various ways, and can be private or public firms. The sample for this study consists of 252 documents (and their citations) published in the WoS core collection during the 1992-2018 period. It was extracted with an advanced search as follows: ("private equit"" OR "venture capita") AND ("small and mediumsized" OR "small and medium sized" OR "SME\*" OR "middle market" OR "middle-market") AND ("family owned" OR "family-owned" OR "FF\*" OR "family business\*" OR "family compan\*" OR "family enterprise\*"). The sample includes 186 journals. The distributions of the documents by years, together with their aggregated number of citations and the list of core economic journals are shown in Figure 1 and Table 1, respectively. Table 1 shows the year of the first article included in the sample and the aggregated number of citations corresponding to the articles included in the sample. For each document, the complete information provided by the WoS was retrieved, that is, authors, affiliations, title, keywords, references, citations, abstract, source, and so on.

Arteche Bueno, L.; Prado Román, C.; Fernandez Portillo, A.(2019). Private Equity focused on Family Firms & Small and Medium Sized Companies: Review and Science Mapping Analysis of the Recent Scientific Field. *European Journal of Family Business*, 9(2), 146-158.

Table 1 List of Top 20 Journals in the PE Field within

FFs & SMEs between 1992 and 2018.

Journal	First Year Inclu ded	N° of Docume nts Include d	Total Citations
Small Business Economics	1997	11	296
Journal of Business Venturing	1992	5	251
Research Policy	2003	5	187
Journal of Management	2005	1	171
Organization Science	2003	1	129
Academy of Management Perspectives	2012	1	98
Journal of Economic Geography	2002	2	72
Regional Studies	2001	2	60
International Small Business Journal	2006	5	55
Entrepreneurship Theory and Practice	2011	3	55
Corporate Governance - An International Review	2005	4	52
Planning C-Government Policy	2001	4	47
Journal of Cleaner Production	2015	1	32
Applied Soft Computing	2013	1	31
Venture Capital	2015	6	21
Strategic Entrepreneurship Journal	2013	1	17
Journal of Family Business Strategy	2012	2	17
Journal of Small Business Management	2013	4	14
Journal of Banking & Finance	2014	1	7
Journal of Small Business and Enterprise Development	2016	4	4



**Figure 1** Distribution of documents included in the research by years and aggregated annual number of citations.

#### Procedure and Sample's Processing

The documents were downloaded from WoS as plain text and added to SciMAT in ISIWoS format. They are the knowledge base for further SMA. Thus, it contains the bibliographic information stored by WoS per each research document. To improve the data quality, a deduplicating process was applied: the most repeated keywords and words representing the same concept were grouped as unit of analysis. A total of 872 word groups were created: 10 top word groups ("VC", "SME", "PE", "Firm", "FFs", "Investments", "Market", "Finance", "Industries" and "Models") appearing in the majority of the documents were classified as "stop" groups (words with a very broad and general meaning) in the tool. Table 2 shows the top 20 keywords that were not classified as stop groups.

Table 2	Top 2	0 Keyword	s of	the	PE	Field	within	FFs	£
SMEs bet	ween	1992 and 2	018	•					

	Number of		Number of
Keywords	document	Keywords	document
	S		S
Performance	35	Ownership Structure	15
Innovation	29	IPOs	13
Corporate Governance	27	Strategies	12
R&D	18	Informatio n	12
Growth	18	Financing	11
Agency Costs	17	Firm Performan ce	10
Ownership	17	United Kingdom	10
Management	15	Start-Up	9
Entrepreneurship	15	Investor	9
Leveraged Buyouts	15	Managem ent Buyouts	9

Next, using the period manager of SciMAT, the periods of time of the longitudinal analysis were established. The whole time frame (1992-2018) was divided into three consecutive periods of time: 1992-2006, 2007-2012 and 2013-2018. In these periods of time, 33, 71 and 148 documents indexed in the WoS were found, respectively. The first period encompasses a greater number of years compared to the last two periods, but it was decided to make this distribution of years because: (i) in the early years of research there were few documents per year and, in order to detect correctly the themes of a discipline, it is necessary to define more or less homogeneous periods of time with respect to the number of documents (Cobo, López-Herrera, Herrera and

Arteche Bueno, L.; Prado Román, C.; Fernandez Portillo, A.(2019). Private Equity focused on Family Firms & Small and Medium Sized Companies: Review and Science Mapping Analysis of the Recent Scientific Field. *European Journal of Family Business*, 9(2), 146-158.

Herrera-Viedma, 2012a; Cobo et al., 2012b; López-Herrera, Herrera-Viedma, Cobo, Martínez, Kou and Shi, 2012), and; (ii) the experience from previous studies of SMA (Cobo, Chiclana, Collop, de Oña, and Herrera-Viedma, In Press, 2014; Cobo et al., 2011a, 2011b, 2012a, 2012b) indicates that an excessive number of periods of time hampers the mapping and interpretation of thematic areas.

The next step is to configure the analysis. To perform it, the following configuration in SciMAT was established: author's, source's and added words as the unit of analysis (all with a threshold of 2 times as minimum frequency), cooccurrence analysis as the tool to build the networks (again with an edge value reduction of 2 times as minimum), equivalence index as the similarity measure to normalize the networks, and the simple centers algorithm as the clustering algorithm to detect the clusters or themes (with a network size range of between 3 and 12 times). The bibliometric measures chosen were the h-index and the sum of citations calculated for the documents that were mapped each theme. Measures used for the to longitudinal maps were: (i) the inclusion index to detect conceptual nexus between research themes of different periods of time through the evolution maps, and; (ii) the Jaccard's Index (Peters and van Raan, 1993), which is a common similarity measure for the normalization process needed in bibliometrics, for the overlapping of the different detected clusters.

# SMA'S results through SCIMAT

Detection of Research Themes: The PE Field within the Family Owned SMEs Market Segment In order to analyze the most highlighted themes of the recent PE field focused on family owned SMEs for each period of time, a strategic diagram is provided. In each diagram, the sphere size is proportional to the number of citations associated with each research theme.

# First period (1992-2006).

According to the strategic diagram presented in Figure 2, the PE research activity pivots on 5 period. themes during this with entrepreneurship, corporate governance, management, decision-making and innovation as key motor themes. The performance measures of the motor themes are given in Table 3. Entrepreneurship is the major motor theme in terms of performance measures: 300 citations and h-index 2.

Corporate governance is a system of structures and processes to direct and control the functions of an organization by setting up rules, procedures and formats for managing decisions within an organization (Palaniappan, 2017). L. Arteche Bueno, C. Prado Román, A. Fernandez Portillo Management of corporate governance was identified together with other themes by Kaplan as one of the main sources for PEs' value enhancement (Jensen, Kaplan, Ferenbach, Feldberg, Moon and Davis, 2006); it was also suggested as a corner stone in value creation by many studies (e.g., Jensen et al., 2006; Millson and Ward, 2005; Nisar, 2005). It is accepted that, on average, PE backing exerts a positive effect on investee firms. But little attention has been paid in the literature to the effect of PE involvement in FFs. PE financing is regularly promoted to meet the need for finance and, in addition, provide managerial expertise to help businesses overcome some of the challenges associated with growth. However, to retain ownership and control over the family business, owner managers often rely on internally generated funds (Berger and Udell, 1998; 2001; Romano, Tanewski and Poutziouris, Smyrnios, 2001). FFs that avoid external influences may be reluctant to take on any form of external finance, including PE (Poutziouris, 2001; Upton and Petty, 2000) but this could constrain their ability to grow.

In this period, Kellermanns and Eddleston (2006) investigated how generational involvement, willingness to change, and the ability to recognize technological opportunities impact corporate entrepreneurship in FFs. Their findings suggest that willingness to change and technological opportunity recognition are positively related to corporate entrepreneurship in FFs.

Several studies focus on the use of PE and VC by FFs between 1992 and 2006. These sources may be preferred in many cases because of the opportunity it offers to fund the FF transition (Upton and Petty, 2000).

The low use of external equity financing by FFs has been a focus of research in the past. This is usually due to a higher preference for internally generated funds rather than external sources, or debt financing rather than external equity financing. These preferences are linked to approaches done by several theories, like the stewardship theory (Davis, Schoorman and Donaldson, 1997). In FFs, financing has been linked to strategic decisions such as the timing of succession (Kimhi, 1997) and the sale of family business (Bhattacharya and Ravikumar, 2001).

Innovation emerges as a key motor theme yet in this period with studies about the important role in the process of creative destruction of SMEs (Acs, Morck, Shaver and Yeung, 1997), with focus on the international diffusion of SMEs innovations.

Table 3 Performance Measures for the Motor Themes(1992-2006).

Arteche Bueno, L.; Prado Román, C.; Fernandez Portillo, A.(2019). Private Equity focused on Family Firms & Small and Medium Sized Companies: Review and Science Mapping Analysis of the Recent Scientific Field. *European Journal of Family Business*, 9(2), 146-158.

Motor Themes	Documents	Citations	h-Index
Entrepreneurship	2	300	2
Management	1	129	1
Corporate Governance	2	113	2
Decision-Making	2	102	2
Innovation	4	90	1



Figure 2 Strategic diagram for the 1992-2006 period.

#### Second period (2007-2012).

The research was focused on 5 themes (see Figure 3). In this case, 3 major themes can be identified (motor themes plus basic themes): corporate governance, innovation and empirical-analysis. Performance measures of the motor themes are shown in Table 4.

Corporate governance is closely linked to the use of investor funds to change corporate governance arrangements through buyouts of firms by PE firms in this period (Gilligan and Wright, 2010). PE professionals take into account family-specific criteria when selecting FFs to invest in, including human resources and opportunities to reduce agency costs. Furthermore, PE professionals prefer FFs that are already professionalized (Dawson, 2011).

Innovation appears as motor theme in this period. Bruque and Moyano (2007) studied the factors behind the intensity and speed of adoption of information technology in SMEs in which family or cooperative character play an important role. Their results indicate that there are a number of internal factors that influence the success of the adoption decision, on the one hand, and the implementation process, on the other hand. Puig and Perez (2009) studied

innovation related to internationalization: the dominant role played by large FFs in the internationalization of the Spanish economy. In contrast with other countries, foreign capital and technology and collective action at regional, national and international levels play a far more important role in the internationalization of large FFs.

Several studies indicate that family involvement appears to result in lower use of external equity. In general, the distance between family businesses and external investors is large, mainly due to the "empathy gap" between owners and investors (Poutziouris, 2011) or because of the preferred retention of control rather than firm's growth and development (Wu, Chua and Chrisman, 2007).

Studies of the use of PE and VC by FFs in this period suggest that these sources may be preferred in many cases because of the nonfinancial benefits that these types of investors can bring to the family such as managerial support, expertise, and contacts (Tappeiner, Howorth, Achleitner and Schraml, 2012; Martí, Menéndez-Requjo and Rottke, 2013).

In general, FF owners balance financial and nonfinancial resources of PE with the need to cede control rights: non-financial resources are valued more highly when resolving family issues (Tappeiner et al., 2012).



Figure 3 Strategic diagram for the 2008-2012 period.

Table 4Performance Measures for the Motor Themes(2007-2012).			
Motor Themes	Documents	Citations	h-Index

Arteche Bueno, L.; Prado Román, C.; Fernandez Portillo, A.(2019). Private Equity focused on Family Firms & Small and Medium Sized Companies: Review and Science Mapping Analysis of the Recent Scientific Field. *European Journal of Family Business*, 9(2), 146-158.

152			
Corporate Governance	11	257	8
Innovation	9	128	5

#### Third period (2013-2018).

The research conducted in this period is distributed in 12 PE themes (see Figure 4), with a clear atomization if compared to previous years. The performance measures of the main motor themes of the period are shown in Table 5.

Table 5Performance Measures for the main MotorThemes (2013-2018).

Motor Themes	Documents	Citations	h-Index
Agency Theory	12	55	2
Research & Development	9	44	2
Earnings- Management	6	43	4
Management Buyouts	15	13	2

Management buyouts ("MBOs") are a motor theme when they became phenomena of the 1980s: when no suitable family successor can be identified, FFs' owners may select an MBO exit route. Some studies suggest that a PE buyout is a governance mechanism that may sustain an entrepreneurial transition by realigning family interests and goals (Di Toma and Montarani, 2017). Secondary buyouts (leveraged sales from one PE fund to another) have been the fastest growing segment of PE deals in last decade and therefore highly studied (Degeorge, Martin and Phalippou, 2016).

Despite more and more FFs open their capital for investors, existing studies mainly outside conclude that family companies are more reluctant that others to hand over control to outside investors. Exploratory evidence from a sample of Belgian FFs is supportive of the hypothesis that family members who identify strongly with their firms are less willing to cede control to outside investors and, if they do cede control, have a stronger preference for investors who may readily identify with FFs, like family offices or high net worth individuals (like business angels), rather than investors who may not fit well with a familial identity, such as PE or VC sponsors (Neckebrouck, Manigart an Meuleman, 2017).

Relevant literature about the PE's positive effects on corporate governance and value creation has been developed in the last period (Acharya, Gottschalg, Hahn and Kehoe, 2013). The impact of PE on FFs' performance was studied thought the analysis of the productivity growth in a sample of PE-backed family companies in 2016. The study found that FFs accessing PE showed lower productivity growth before the initial PE round, which was driven by an imbalance between inputs and output, especially in founder-controlled firms. This analysis also confirmed the positive impact of PE involvement on productivity growth in foundercontrolled firms (Croce and Martí, 2016).



Figure 4 Strategic diagram for the 2013-2018 period.

# Thematic Evolution of the PE Scientific Field focused on FFs & SMEs (1992-2018)

Structural analysis of the evolution of the PE scientific field focused on FFs and SMEs between 1992 and 2018.

According to Figure 5, the research developed in the PE activity field presents a high cohesion level in the cases of the thematic areas "corporate governance-entrepreneurship" and "innovation-management".

The two main thematic areas present a growth pattern, because they have been growing in the number of themes discussed since their origin. However, the PE scientific community is dynamic and relatively recent as the number of scientific documents starts growing in the second period, only twelve years ago. MBOs and the UK are new scientific areas that emerge between 2013 and 2018 and that lead to new research areas related to the study of the role of information asymmetries (Dehlen, Zellweger, Kammerlander and Halter, 2014) and the value of FFs for PE investors (Ahlers, Hack and Kelleermanns, 2014), in the case of MBOs.

And also related to the UK market (Mason and Pierrakis, 2013; Mason and Harrison, 2015), in the case of the UK. The two main thematic areas are constant between the first and second periods and then they go through an atomization process since year 2013. Therefore, the scientific communication has resulted in a specialization of

Arteche Bueno, L.; Prado Román, C.; Fernandez Portillo, A.(2019). Private Equity focused on Family Firms & Small and Medium Sized Companies: Review and Science Mapping Analysis of the Recent Scientific Field. *European Journal of Family Business*, 9(2), 146-158.

the big historical thematic areas in niche research themes that become basic and transversal themes.



Figure 5 Thematic areas and evolution of the PE field focused on FFs & SMEs (1992-2018).

# Performance analysis of the evolution of the PE scientific field focused on FFs and SMEs (1992-2018).

In Table 6, performance measures of the main thematic areas are provided. Corporate governance-entrepreneurship area stands out over the rest in terms of citations: 893 citations across the 1992-2018 period. Relevant research in corporate governance started in 2003, with a study of the primary rationalities governing the exchange relationships in family investment decisions during the early stages of new venture creation (Steier, 2003).

In the second period, corporate governanceentrepreneurship has high impact papers about management practices (Bloom, Genakos, Sadun and Van Reenen, 2012), where relevant findings about the relation between ownership and management were done. In the case of entrepreneurship, studies about the internationalization processes (George, Wiklund and Zahra, 2005) and owners' succession (Wasserman, 2003) are the ones with the highest impact in the first period, with 171 and 129 citations, respectively,

Table 6	Performance Me	easures of th	e PE Field'	s Main
Thematic Areas (1992-2018).				
Main Thematic Areas	matic Aroas	Document	Citation	h-
	matic Aleas	S	S	Index

Corporate Governance - Entrepreneurship	78	893	15
Innovation - Management	97	715	14

#### **Discussion and applications**

This study aims at giving continuity and increase family owned SMEs' consciousness of the positive effects and the increasing availability of the PE as an alternative to traditional growth funding, which has been widely demonstrated in past scientific research (Paglia and Harjoto, 2014). The coding phase of the analysis led to three interesting findings:

1. The quantity of PE scientific publications within the FFs and SMEs has an exponential increase across the last two decades thought the highest impact articles majority belong to the 1992-2012 period of the analysis. There has been an increase in the number of themes over time and, thus, an emergence of a more diverse and complex PE scientific discipline within field.

2. The thematic evolution analysis performed in this paper shows that corporate governanceentrepreneurship and innovation-management are the two big thematic areas of the recent PE research field. Therefore, it would be recommendable for family owned SMEs to promote corporate governance measures effectively.

3. The PE field presented a continuous, consistent, and cohesive growth, because there are no gaps in the main thematic areas. However, several research themes do not constitute a conceptual nexus with the classical themes and do not belong to any thematic area. Further scientific research on the field can be done for example to measure the PE's board members contribution to the acquired company's strategy: the Team Production Theory (Blair and Scout, 2001) shows that members must have knowledge of the firm to make decisions that create value (Kaufman and Englander, 2005).

The research done enables to access and assess key data of the discipline to make decisions in different frameworks:

Family owned SMEs: they might use this analysis to understand and become familiar with PE funds historical activity and the value they can add to their growth plans. In addition, the use of PE to execute internationalization plans can have additional benefits different to the mere financing and performance improvement: recent studies have demonstrated that a high degree of geographic international diversification enables multinational companies to improve its social

Arteche Bueno, L.; Prado Román, C.; Fernandez Portillo, A.(2019). Private Equity focused on Family Firms & Small and Medium Sized Companies: Review and Science Mapping Analysis of the Recent Scientific Field. *European Journal of Family Business*, 9(2), 146-158.

performance (Aguilera-Caracuel, Guerrero-Villegas and García-Sánchez, 2017).

Private Equity funds focused on the middlemarket: in the past decade there has been an increasing role of the PE industry in the financing of enterprises what can boost economic growth. PE funds can identify new market niches within the family owned SMEs segment for their acquisitions.

Academic centers: scientists could identify new and relevant challenges in their field for future research, as well as the emerging themes.

This study opens up new possibilities for discovering important research areas in the PE field within the family owned middle-market. It provides empirical analysis that can benefit from a further development of this subject as a discipline.

#### Limitations and further research

This research has several limitations, which in turn, reveal the path for future lines of research. The first limitation is related to the scope of our results and their implications. Since the study was performed on a recent sample, the results cannot be transferred to the entire scientific PE field focused on the middle-market. Future research could develop the study defined here: scientific research prior to year 1992 could be analyzed and also a new research with other keywords might result in new findings.

To choose the information sources for our analysis, we have used the WoS database: an alternative selection of databases would likely produce different results. Other limitations relate to our methodology since we use only documents published in the most those important journals indexed in WoS in the PE category. Therefore, we are missing the PE research published primarily outside of those journals that are not indexed in WoS. Other methodological bias was introduced in the cowords analysis: further research could be done by using other bibliometric techniques that complement this study and provide a systematic description of the structure of the field.

# References

- Acharya, V. V., Gottschalg, O. F., Hahn, M. and Kehoe, C. (2013). "Corporate Governance and Value Creation: Evidence from Private Equity". *Review of financial studies*, Vol. 26, No. 2, pp. 368-402. doi: 10.1093/rfs/hhs117.
- Acs, Z.J., Morck, R., Shaver, J.M., Yeung, B. (1997). "The Internationalization of Small and Mediumsized Enterprises: A Policy Perspective". Small Business Economics, Vol. 9, No. 1, pp. 7-20.

- Aguilera-Caracuel, J., Guerrero-Villegas, J. and García-Sánchez, E. (2017). "Reputation of multinational companies: Corporate social responsibility and internationalization". European Journal of Management and Business Economics, Vol. 26, No. 3, pp. 329-346. doi: 10.1108/EJMBE-10-2017-019.
- Ahlers, O. (2014). FFs and Private Equity: A Collection of Essays on Value Creation, Negotiation, and Soft Factors. Vallendar, Germany: Springer Gabler.
- Ahlers, O., Hack, A. and Kellermanns, F.W. (2014). ""Stepping into the buyers' shoes": Looking at the value of FFs through the eyes of private equity investors". *Journal of Family Business Strategy*, Vol. 5, No. 4, pp. 384-396. doi: 10.1016/j.jfbs.2014.04.002.
- Anderson, R. C. and Reeb, D. M. (2003). "Founding-Family Ownership and Firm Performance: Evidence from the S&P 500". *Journal of Finance*, Vol. 58, No. 3, pp. 1301-1328.
- Astrachan, J. H. and Shanker, M. C. (2003). "Family Businesses' Contribution to the U.S. Economy: A Closer Look", *Family Business Review*, Vol. 16, No. 3, pp. 211-219. doi: 10.1177/08944865030160030601.
- Benavides-Velasco, C.A., Quintana-García, C. and Guzmán-Parra, V.F. (2013). "Trends in family business research". Small Business Economics, Vol. 40, pp- 41-57.
- Bengtsson, P., Nagel, R. and Nguyen, A. (2008). Value Creation in Buyouts: value-enhancement practices of private equity firms with a hands-on approach. Jönköping International Business School.
- Berger, A. N. and Udell, G. F. (1998). "The economics of small business finance: The roles of private equity and debt markets in the financial growth cycle". Journal of banking & finance, Vol. 22, No. 6-8, pp. 613-673. doi: 10.1016/S0378-4266(98)00038-7.
- Bhattacharya, U. and Ravikumar, B. (2001). "Capital markets and the evolution of family businesses". *Journal of Business*, Vol. 74, pp. 187-219.
- Birley, S. and Westhead, P. (1990). "Private business sales environments in the United Kingdom". *Journal of Business Venturing*, Vol. 5, No. 2, pp. 349-373.
- Blair, M. and Stout, L.A. (2001). "Director accountability and the mediating role of the corporate board". *Washington University Law Review*, Vol. 79, pp. 403-449.
- Bonini, S. (2015). "Secondary Buyouts: Operating Performance and Investment Determinants". *Financial Management*, Vol. 44, No. 2, pp. 431-470. doi: 10.1111/fima.12086.
- Bruque, S. and Moyano, J. (2007). "Organisational determinants of information technology adoption and implementation in SMEs: The case of family and cooperative firms". *Technovation*, Vol. 27, No. 5, pp. 241-253.
- Bruton, G. D., Filatochev, I., Chahine, S. and Wright, M. (2010). "Governance, Ownership Structure, and Performance of IPO Firms: The Impact of Different Types of Private Equity Investors and Institutional Environments". Strategic Management Journal, Vol. 31, No. 5, pp. 491-509. doi: 10.1002/smj.822.

154

Arteche Bueno, L.; Prado Román, C.; Fernandez Portillo, A.(2019). Private Equity focused on Family Firms & Small and Medium Sized Companies: Review and Science Mapping Analysis of the Recent Scientific Field. *European Journal of Family Business*, 9(2), 146-158.

- Callon, M., Courtial, J. P., Turner, W. A. and Bauin, S. (1983). "From translations to problematic networks: An introduction to co-word analysis". *Social Science Information*, Vol. 22, No. 2, pp. 191-235.
- Chrisman, J., Chua, J., Pearson, A. W. and Barnett, T. (2012). "Family involvement, family influence and family-centered non-economic goals in small firms". Entrepreneurship: Theory and Practice, Vol. 36, No. 2, pp. 267-293.
- Chua, J.H., Chrisman, J.J. and Sharma, P. (2003). "Succession and non-succession concerns of family firms and agency relationship with non-family managers". *Family Business Review*, Vol. 16, pp. 89-107.
- Chua, J. H., Chrisman, J. J., Steier, L. P. and Rau, S. B. (2012). "Sources of Heterogeneity in FFs: An Introduction". *Entrepreneurship: Theory and Practice*, Vol. 36, No. 6, pp. 1103-1113.
- Cobo, M., Chiclana, F., Collop, A., de Oña, J. and Herrera-Viedma, E. (In Press, 2014). "A bibliometric analysis of the intelligent transportation systems research based on science mapping". *IEEE Transactions on Intelligent Transportation* Systems. doi:10.1109/TITS.2013.2284756.
- Cobo, M., López-Herrera, A., Herrera-Viedma, E. and Herrera, F. (2011a). "An approach for detection, quantifying, and visualizing the evolution of a research field: A practical application to the fuzzy sets theory field". Journal of Informetrics, Vol. 5, pp. 146-166. doi:10.106/j.joi2010.10.002.
- Cobo, M., López-Herrera, A., Herrera-Viedma, E. and Herrera, F. (2011b). "Science mapping software tools: Review, analysis and cooperative study among tools". Journal of the American Society for Information Science and Technology, Vol. 62, pp. 1382-1402. doi:10.1002/asi.21525.
- Cobo, M., López-Herrera, A., Herrera F. and Herrera-Viedma, E. (2012a). "A note on the ITS topic evolution in the period 2000-2009 at T-ITS". *IEEE Transactions on Intelligent Transportation Systems*, Vol. 13, pp. 413-420. doi: 10.1109/TITS.2011.2167968.
- Cobo, M., López-Herrera, A., Herrera-Viedma, E. and Herrera, F. (2012b). "SciMAT: A new science mapping analysis software tool." Journal of the American Society for Information Science and Technology, Vol. 63, No 8, pp. 1609-1630. doi:10.1002/asi.22688.
- Croce, A. and Marti J. (2016). "Productivity Growth in Private-Equity-Backed FFs". *Entrepreneurship Theory and Practice*, Vol. 40, No. 3, pp. 657-683. doi: 10.1111/etap.12138.
- Cumming, D. (2007). "Government Policy Towards Entrepreneurial Finance: Innovation Investment Funds". ournal Of Business Venturing, Vol. 22, No. 2, pp. 193-235. doi: 10.1016/j.jbusvent.2005.12.002.
- Cumming, D., Cole, R. and Li, D. (2016). "Do banks or VCs spur small firm growth?". Journal of international financial markets, institutions & money, Vol. 41, pp. 60-72. doi: 10.1016/j.intfin.2015.12.005.

- Cumming, D., Grilli, L. and Murtinu, S. (2017). "Governmental and independent venture capital investments in Europe: A firm-level performance analysis". *Journal of corporate finance*, Vol. 42, pp. 439-459. doi: 10.1016/j.jcorpfin.2014.10.016.
- Cumming, D., Siegel, D. S. and Wright, M. (2007). "Private equity, leveraged buyouts and governance". *Journal of corporate finance*, Vol. 13, No. 4, pp. 439-460. doi: 10.1016/j.jcorpfin.2007.04.008.
- Dvid, J.H., Schoorman, F.D. and Donaldson, L. (1997). "Toward a stewardship theory of management". Academy of Management Review, Vol. 22, pp. 20.47.
- Dawson, A. (2011). "Private equity investment decisions in FFs: The role of human resources and agency costs". *Journal of business venturing*, Vol. 26, No. 2, pp. 189-199. doi: 10.1016/j.jbusvent.2009.05.004.
- Degeorge, F., Martin, J., Phalippou, L. (2016). "On secondary buyouts". Journal of financial economics, Vol. 120, No. 1, pp. 124-145. doi: 10.1016/j.jfineco.2015.08.007.
- Dehlen, T., Zellwegere, T., Kammerelander, N. and Halter, F. (2014). "The Role Of Information Asymmetry In The Choice Of Entrepreneurial Exit Routes". Journal Of Business Venturing, Vol. 29, No. 2, pp. 193-209. doi: 10.1016/j.jbusvent.2012.10.001.
- Di Toma, P. and Montanari, S. (2017). "Corporate governance effectiveness along the entrepreneurial process of a FF: the role of private equity". *Journal of Management & Governance*, Vol. 21, No. 4, pp. 1023-1052. doi: 10.1007/s10997-016-9373-1.
- Dwyer, B. and Kotey, B. (2015). "Financing SME Growth: The Role of the National Stock Exchange of Australia and Business Advisors". *Australian accounting review*, Vol. 25, No. 2, pp. 114-123. doi: 10.1111/auar.12074.
- European Central Bank (November of 2017). External sources of financing and needs of SMEs in the euro area. Survey on the Access to Finance of Enterprises in the euro area - April to September 2017. pp. 12-16. Retrieved from http://www.ecb.europa.eu/stats/ecb\_surveys/saf e/html/index.en.html.
- European Commission. Entrepreneurship and Small and medium-sized enterprises (SMEs). European Commission. Retrieved from http://www.ec.europa.eu/growth/smes\_en.
- Fazekas, B. and Becksky-Nagy, P. (2015). "Private equity market in recovery". Emerging markets queries in finance and business 2014, Vol. 32, pp. 225-231. doi: 10.1016/S2212-5671(15)01386-6.
- Ferrer, M. A., Bertoni, F. and Martí Pellón, J. (2010). Effects of venture capital and private equity on investment-cash flow sensitivity of Spanish firms. Universidad Complutense de Madrid.
- Gilligan, J. and Wright, M. (2010). Private equity demystified. 2nd edition. London: ICAEW.
- Grilli, L. and Murtinu, S. (2015). "New technologybased firms in Europe: market penetration, public venture capital, and timing of investment". *Industrial and corporate change*, Vol. 24, No. 5, pp. 1109-1148. doi: 10.1093/icc/dtu025.

Arteche Bueno, L.; Prado Román, C.; Fernandez Portillo, A.(2019). Private Equity focused on Family Firms & Small and Medium Sized Companies: Review and Science Mapping Analysis of the Recent Scientific Field. *European Journal of Family Business*, 9(2), 146-158.

- Haro de Rosario, A., Caba Pérez, M. C. and Cazorla Papis, L. (2013). Eficiencia en el sector del capital riesgo y private equity: análisis del caso español (Doctoral thesis). Universidad de Almería. Almería.
- Harzing, A. and van der Wal, R. (2008). "Google scholar as a new source for citation analysis". *Ethics in Science and Environmental Politics*, Vol. 8, pp. 61-73. doi: 10.3354/esep0007.
- Herrera-Viedma, E., Martínez, M. A., Cobo, M. and Herrera, M. (2015). "Analyzing the Scientific Evolution of Social Work Using Science Mapping". *Research on Social Work Practice 2015*, Vol. 25, No. 2, pp. 257-277.
- Higashide, H. and Birley, S. (2002). "The consequences of conflict between the venture capitalist and the entrepreneurial team in the United Kingdom from the perspective of the venture capitalist". *Journal of Business Venturing*, Vol. 17, No. 1, pp. 59-81.
- Hirsch, J. (2005). "An index to quantify an individual's scientific research output". *Proceedings of the National Academy of Sciences*, Vol. 102, pp. 16569-16572.
- Howorth, C., Westhead, P. and Wright, M. (2004). "Buyouts, information asymmetry and the family management dyad". *Journal of Business Venturing*, Vol. 19, No. 4, pp. 509-534.
- Hung, Y. D. and Tsai, M. H. (2017). "Value Creation and Value Transfer of Leveraged Buyouts: A Review of Recent Developments and Challenges for Emerging Markets". Emerging markets finance and trade, Vol. 53, No. 4, pp. 877-917. doi: 10.1080/1540496X.2016.1193000.
- IFERA (2003). "Family Businesses Dominate". Family Business Review, Vol. 16, No. 4, pp. 235-240.
- Jensen, M., Kaplan, S., Ferenbach, C., Feldberg, M., Moon, I. and Davis, C. (2006). "Morgan Stanley Roundtable on Private Equity and Its import for Public Companies". Journal of Applied Corporate Finance, Vol. 18, No. 3, pp. 8-37.
- Johan, S. and Zhang, M. J. (2016). "Private equity exits in emerging markets". *Emerging markets review*, Vol. 29, pp. 133-153. doi: 10.1016/j.ememar.2016.08.016.
- Kaplan, S. N., and Schoar, A. (2005). "Private equity performance: Returns, persistence, and capital flows". Journal of finance, Vol. 60, No. 4, pp. 1791-1823. doi: 10.1111/j.1540-6261.2005.00780.x.
- Kaplan, S. N., Martel, F. and Stroberg, P. (2007). "How Do Legal Differences And Experience Affect Financial Contracts?". Journal of financial intermediation, Vol. 16, No. 3, pp. 273-311. doi: 10.1016/j.jfi.2007.03.005.
- Kaufman, A. and Englander, E. (2005). "A team production model of corporate governance". Academy of Management Executive, Vol. 19, pp. 9-22.
- Kellermanns, FW. and Eddleston, KA. (2006). "Corporate entrepreneurship in family firms: A family perspective". Entrepreneurship Theory and Practice, Vol. 30, No. 6, pp. 809-830.
- Kimhi, A. (1997). "Intergenerational succession in small family businesses: Borrowing constraints and optimal timing of succession". Small Business Economics, Vol. 9, pp. 309-318.

- L. Arteche Bueno, C. Prado Román, A. Fernandez Portillo Klein, S. B. (2000). "Family Businesses in Germany: Significance and Structure". Family Business Review, Vol. 13, No. 3, pp. 157-181.
  - Koropp, C., Kellermanns, F.W., Grichnik, D. and Stanley, L. (2014). "Financial decision in FFs: An adaptation of the theory of planned behavior". *Family Business Review*, Vol. 27, pp. 307-327.
  - KPMG and European Family Business (Sixth edition, 2017). European Family Business Barometer. Retrieved from http://www.europeanfamilybusinesses.eu.
  - Lahmann, A. D. F., Stranz, W. and Velamuri, V. K. (2017). "Value creation in SME private equity buyouts". Qualitative research in financial markets, Vol. 9, No. 1, pp. 2-33. doi: 10.1108/QRFM-01-2016-0004.
  - López-Herrera, A. G., Herrera-Viedma, E., Cobo, M.
    J., Martínez, M. A., Kou, G. and Shi, Y. (2012).
    "A conceptual snapshot of the first decade (2002-2011) of the international journal of information technology & decision making". International Journal of Information Technology & Decision Making, Vol. 11, pp. 247-270. doi:10.11142/S021962220124000020.
  - Martí J., Menéndez-Requejo S. and Rottke O.M. (2013). "The impact of venture capital on family businesses: Evidence from Spain". Journal of World Business, Vol. 48, pp. 420-430.
  - Mason, C.M. and Harrison, R.T. (2015). "Business Angel Investment Activity In The Financial Crisis: Uk Evidence And Policy Implications". *Environment And Planning C-government And Policy*, Vol. 33, No. 1, pp. 43-60. doi: 10.1068/c12324b.
  - Mason, C, and Pierrakis, Y. (2013). "Venture Capital, The Regions And Public Policy: The United Kingdom Since The Post-2000 Technology Crash". *Regional Studies*, Vol. 47, No. 7, pp. 1156-1171. doi: 10.1080/00343404.2011.588203.
  - Metrick, A. and Yasuda, A. (2005). "Venture Capital and Other Private Equity: a Survey". European Financial Management, Vol. 17, No. 4, pp. 619-654.
  - Michiels, A. and Molly, V. (2017). "Financing decision in family businesses: a review and suggestions for developing the field". *Family Business Review*, Vol. 30, No. 4, pp. 369-399. Doi: 10.1177/0894486517736958.
  - Millson, R. and Ward, M. (2005). "Corporate Governance Criteria as Applied in Private Equity Investments". South African Journal of Business Management, Vol. 36, No. 1, pp. 78-83.
  - Mohnen, P., Palm, F. C., Van Der Loeff, S. S. and Tiwari, A. (2008). "Financial Constraints And Other Obstacles: Are They A Threat To Innovation Activity?". *Economist-Netherlands*, Vol. 156, No. 2, pp. 201-214. doi: 10.1007/s10645-008-9089-y.
  - Morck, R. and Yeung, B. (2003). "Agency Problems in Large Family Business Groups". *Entrepreneurship: Theory & Practice*, Vol. 27, No. 4, pp. 367-382.
  - Neckebrouck, J., Manigart, S. and Meuleman, M. (2017). "Attitudes of FFs toward outside investors: the importance of organizational identification". *Venture Capital*, Vol. 19, No. 1-2, pp. 29-50. doi: 10.1080/13691066.2016.1255414.
  - Nisar, M. N. (2005). "Investor Influence on portfolio company growth and development strategy". *Journal of Private Equity*, Vol. 9, No. 1, 22-35.

Arteche Bueno, L.; Prado Román, C.; Fernandez Portillo, A.(2019). Private Equity focused on Family Firms & Small and Medium Sized Companies: Review and Science Mapping Analysis of the Recent Scientific Field. *European Journal of Family Business*, 9(2), 146-158.

- Noyons, E., Moed, H. and Van Rann, A. (1999b). "Integrating research performance analysis and science mapping". *Scientometrics*, Vol. 46, pp. 591-604.
- Paglia, J.K. and Harjoto, M.A. (2014). "The effects of private equity and venture capital on sales and employment growth in small and medium-sized businesses". *Journal of banking & finance*, Vol. 47, pp. 177-197. doi: 10.1016/j.jbankfin.2014.06.023.
- Palaniappan, G. (2017). "Determinants of corporate financial performance relating to board characteristics of corporate governance in Indian manufacturing industry: An empirical study". European Journal of Management and Business Economics, Vol. 26, No. 1, pp. 67-85. doi: 10.1108/EJMBE-07-2017-005.
- Peters, H. P. and Van Raan, A. F. (1993). "Co-word-based science maps of chemical engineering. Part I: Representations by direct multidimensional scaling". *Research Policy*, Vol. 22, pp. 23-45.
  Poutziouris, P.Z. (2001). "The views of family
- Poutziouris, P.Z. (2001). "The views of family companies on venture capital: Empirical evidence from the UK small to medium-size enterprising economy". Family Business Review, Vol. 14, No. 3, pp. 277-291.
- Poutziouris, P.Z. (2011). "The financial structure and performance of owner-managed FFs: Evidence from the UK economy". Universia Business Review, Vol. 32, pp. 70.
- Preqin (2010). Q3 2010 private equity data: Private equity has strongest quarter since financial crisis. Preqin. http:// privateequityblogger.com/2010/10/q3-2010private-equitydata.html.
- PWC (2012). Family Firm: A resilient model for the 21st century. From PwC Family Business Survey. Retrieved from: http://www.pwc.com/en\_GX/gx/pwc-familybusiness-survey/assets/pwc-family-businesssurvey-2012.pdf.
- **PWC (2018).** The values effect. From PwC Family Business Survey 2018. Retrieved from: https://www.pwc.com/gx/en/family-businessservices/assets/pwc-global-family-business-survey-2018.pdf.
- Ramón Llorens, M. C. and Hernández Cánovas, G. (2011). Análisis de la actuación del gestor del capital riesgo en la toma de decisiones de participación financiera. Universidad Politécnica de Cartagena.
- Renneboog, L., Simons, T. and Wright, M. (2007). "Why Do Public Firms Go Private in the UK? The Impact of Private Equity Investors, Incentive Realignment and Undervaluation". *Journal of corporate finance*, Vol. 13, No. 4, pp. 591-628. doi: 10.1016/j.jcorpfin.2007.04.005.
- Robertson, J. (2017). "Emergent new finance: hedge funds and private equity funds in East Asia". *Journal of the Asia Pacific Economy*, Vol. 22, No. 4, pp. 626-646. doi: 10.1080/13547860.2017.1349861.
- Romano, C.A., Tanewski, G.A. and Smyrnios, K.X. (2003). "Capital structure decision making: a model for family business". *Journal of Business Venturing*, Vol. 16, pp. 285-310.

- Ruiz Martín, M. (2006). Actividad de las entidades de capital riesgo en España. CNMV.
- Sánchez Hernández, M. M., Reverte Maya, C. and Rojo Ramírez, A. A. (2013). La valoración de inversiones por parte de las sociedades de capital riesgo: un estudio empírico aplicado al ámbito europeo. Universidad Politécnica de Cartagena.
- Seco Benedicto, M. T., Martí Pellón, J. (2007). La actividad de capital riesgo en el área de Asia-Pacífico: entorno y determinantes. Universidad Complutense de Madrid.
- Shanker, M.C. and Astrachan, J.H. (1996). "Myths and realities: family businesses' contributions to the US economy". *Family Business Review*, Vol. 9, No. 2, pp. 107-123.
- Siegel, D., Wright, M. and Filatotchev, I. (2011). "Private Equity, LBOs, and Corporate Governance: International Evidence". Corporate Governance: An International Review, Vol. 19, No. 3, pp. 185-194.
- Sirmon, D.G. and Hitt, M.A. (2003). "Managing resources: linking unique resources, management, and wealth creation in family firms". *Entrepreneurship Theory and Practice*, Vol. 27, No. 4, pp. 339-358.
- Small, H. and Amer, J. (1973): "Co-citation in the scientific literature: A new measure of the relationship between two documents". Soc. Inf. Sci., Vol. 24, No. 4, pp. 265-269.
- Steier, L. (2003). "Variants Of Agency Contracts In Family-financed Ventures As A Continuum Of Familial Altruistic And Market Rationalities". *Journal Of Business Venturing*, Vol. 18, No. 5, pp. 597-618. doi: 10.1016/S0883-9026(03)00012-0.
- Strömberg, P. (2008). "The new demography of private equity". In Lerner, Josh. & Gurung, A. (eds). The Global Impact of Private Equity Report 2008, Globalization of Alternative Investments, Working Papers Vol. 1, World Economic Forum, pp. 3-26.
- Tappeiner, F., Howorth, C., Achleitner, A.K. and Schraml, S. (2012). "Demand for private equity minority investments: A study of large FFs". *Journal of family business strategy*, Vol. 3, No. 1, pp. 38-51. doi: 10.1016/j.jfbs.2012.01.001.
- Tresierra Tanaka, A. E., Balboa Ramón, M. and Martí Pellón, J. (2011). Capital structure analysis in Spanish venture capital-backed firms at the expansion stage. Universidad Complutense de Madrid.
- Tsang, E. W. K. (2002). "Learning from overseas venturing experience: The case of Chinese family businesses". *Journal of Business Venturing*, Vol. 17, No. 1, pp. 21-40.
- Upton, N. and Petty, W. (2000). "Venture capital investment and US family business". Venture Capital, Vol. 2, pp. 27-39.
- Van Raan, A. F. J. (2005). "Measuring science". Handbook of Quantitative Science and Technology Research. Dordrecht, The Netherlands. pp. 19-50.
- Voordeckers, W., Le Breton-Miller, I. and Miller, D. (2014). "In search of the best of both worlds: Crafting a finance paper for the *Family Business Review*". *Family Business Review*, vol. 27, pp. 281-286.

Arteche Bueno, L.; Prado Román, C.; Fernandez Portillo, A.(2019). Private Equity focused on Family Firms & Small and Medium Sized Companies: Review and Science Mapping Analysis of the Recent Scientific Field. *European Journal of Family Business*, 9(2), 146-158.

- Wasserman, N. (2003). "Founder-Ceo Succession And The Paradox Of Entrepreneurial Success". *Organization Science*, Vol. 14, No. 2, pp. 149-172. doi: 10.1287/orsc.14.2.149.14995.
- Winton, A. and Yerramilli, V. (2008). "Entrepreneurial finance: Banks versus venture capital". Journal of financial economics, Vol. 88, No. 1, pp. 51-79. doi: 10.1016/j.jfineco.2007.05.004.
- Wright, M., Amess, K., Weir, C. and Girma, S. (2009a). "Private equity and corporate governance: Retrospect and prospect". Corporate Governance: An International Review, Vol. 17, No. 3, pp. 253-375. doi: 10.1111/j.1467-8683.2009.00744.x.
- Wright, M., Jackson, A. and Frobisher, S. (2010). "Private equity in the UK: Building a new future". *Journal of Applied Corporate Finance*, Vol. 22, pp. 86-95.
- Wu, Z., Chua, J.H. and Chrisman, J.J. (2007). "Effects of family ownership and management on small business equity financing". *Journal of Business Venturing*, Vol. 22, pp. 875-985.

Arteche Bueno, L.; Prado Román, C.; Fernandez Portillo, A.(2019). Private Equity focused on Family Firms & Small and Medium Sized Companies: Review and Science Mapping Analysis of the Recent Scientific Field. *European Journal of Family Business*, 9(2), 146-158.

158

Our Ethic statements are based on COPE's (Committee On Publication Ethics) Best Practice Guidelines for Journal Editors (http://publicationethics.org/files/u2/New\_Code.pdf) **1. Publication and authorship:** 

It is forbidden to publish the same research in more than one journal. Authors must not submit a manuscript to more than one journal simultaneously. The Authors should state that they have written an entirely original work and that the data are real and authentic.

Any sources of financial support must be mentioned or acknowledged in the paper.

It must be stated that there is not any plagiarism and fraudulent data in the work. If the Authors have used the work and/or words of others, this has been appropriately cited or quoted. Third party information and results cannot be used without explicit and authorized mention of the source.

In the paper, Authors should cite or quote previous publications, dealing with the same subject or related to the results presented. The list of references must contain all cited publications. 2. Author's responsibilities:

Authors must not submit a manuscript being under reviewed in another place for publication, nor submit the manuscript elsewhere while being under reviewed in *European Journal of Family Business*. The manuscript must not been previously published or accepted for publication in another place, either a portion of a paragraph or exhibits, or a manuscript in another language. If the manuscript contains information which overlap with the previously published works, in press, or under deliberation for publication elsewhere, the author must cite this work in the manuscript.

All Authors mentioned in the paper must have significantly contributed to the research. All co-authors of the research paper should have made significant contributions to the work and shared accountability for the results.

Authors need to ensure that all data in the submitted article are real and authentic.

The manuscript must be free from any kind of plagiarism, falsification, or fabrications. Also, Self-plagiarism is not acceptable in any case. Authors should minimize recycling their previous writings. If recycling is inevitable, the author should reference the previous writings in the manuscript. Authors must clearly cite their own works. Such self-referencing should be phrased discreetly to avoid compromising the double-blind review process.

Authors are required to provide a complete list of references cited in their paper. If authors received any financial support to conduct the research that is reported in the submitted paper, they should provide the information.

Authors should avoid any kind of conflicts of interest or the presence of conflicts of interest throughout the research process.

European Journal of Family Business imposes a double-blind review process, wherein authors do not recognize the reviewers and vice versa. Authors are obliged to participate in peer review process. Authors should respect the confidentiality of the review process and should not reveal themselves to reviewers and vice versa.

The corresponding author who submits a manuscript to *European Journal of Family Business* should keep all the co-authors of the submission process and the results of the reviewing.

*European Journal of Family Business* holds the copyright of all published articles. As such, authors should ask for consent to publish their article (or a selection from the article) elsewhere.

All authors are obliged to provide retractions or corrections of mistakes.

#### 3. Peer review / responsibility for the reviewers:

Reviewers should have no conflict of interest with respect to the research, the authors and/or the research funders. Specifically, Referees should not review manuscripts authored or co-authored by a person with whom the referee has a close personal or professional relationshipReviewers must alert the editor if a manuscript contains plagiarized material, falsified or manipulated data. They must inform the editor of any similarity between the submitted manuscript and another either published or under consideration by another journal.

Reviewed articles should be treated confidentially. Reviewers must ensure that all unpublished data, information, interpretation and discussion in a submitted article remain confidential. The manuscript should not be shown to, disclosed to, or discussed with others. Reviewers' judgments should be objective and relevant,

Reviewers' judgments should be objective and relevant, published work that is not yet cited should be point out, where appropriate.

When peer-review process is complete, reviewers must destroy/erase the manuscript and before starting the process they must inform the editor should they be unqualified to review the manuscript, or lack the time to review the manuscript, without undue delay. Reviewers must not retain or copy the submitted manuscript in any form.

#### 4. Editorial responsibilities:

The sole responsibility for acceptance or rejection of a manuscript rests with the editor. Responsible exercise of this duty requires that the editor seek advice from reviewers, chosen for their expertise and good judgment, as to the quality and reliability of manuscripts submitted for publication.

The editor and members of the editor's staff will have no conflict of interest with respect to articles they reject/accept. Editorial decisions are based on the relevance of a manuscript to the journal and on the manuscript's originality, quality, and contribution to evidence about relevant questions.

The editor and members of the editor's staff should not disclose any information about a manuscript under consideration to anyone other than those from whom professional advice is sought.

An editor should respect the intellectual independence of authors.

Unpublished information, arguments, or interpretations disclosed in a submitted manuscript should not be used in an editor's own research except with the consent of the author. When a manuscript is so closely related to the current or past research of an editor as to create a conflict of interest, the editor should arrange for some other qualified person to take editorial responsibility for that manuscript.

The editor and members of the editor's staff must preserve anonymity of reviewers.

If any error is discovered in a paper after publication, corrections will be published in the next issue or as soon as the publisher and author, both agree to the proposed changes.

5. Publishing ethics issues:

The editorial board vows to monitor/safeguard publishing ethics as stated in the Committee on Publication Ethics (COPE) guideline resources at http://publicationethics.org/resources/guidelines and the Publication Ethics Resource Kit (PERK) at http://www.elsevier.com/editors/perk/about-perk

Journal editors should consider retracting a publication if: a) They have clear evidence that the findings are unreliable, either as a result of misconduct or honest error.

b) The findings have previously been published elsewhere without proper cross-referencing, permission or justification (i.e. cases of redundant publication)

c) It constitutes plagiarism or reports unethical research.

Editors have to take responsibility for every research they publish and should have procedures to ensure the quality of the material they publish and maintain the integrity of the published record.

Research funders and sponsors should not be able to veto publication of findings that do not favour their product or position. Researchers should not enter agreements that permit the research sponsor to veto or control the publication of the findings.

Editors should always be willing to publish corrections, clarifications, retractions and apologies when needed. If an author notices a significant error or inaccuracy in his/her own published work, it is the author's responsibility to promptly notify the journal editor and cooperate with the editor to retract or correct the paper.

*European Journal of Family Business* is committed to only publishing manuscripts that are free from any plagiarism and that contain no fraudulent data.